June 7, 2005 Via Overnight Delivery



210 N. Park Ave. Winter Park, FL 32789

P.O. Drawer 200 Winter Park, FL 32790-0200

Tel: 407-740-8575 Fax: 407-740-0613 tmi@tminc.com Ms. Jaclyn A. Brilling Commission Secretary New York Department of Public Service Three Empire State Plaza Albany, NY 12223

RE: LDMI Telecommunications, Inc. d/b/a LDMI Telecommunications also d/b/a FoneTel; Petition for Approval to Transfer Control

Dear Mr. Pollock:

Enclosed please find an original and three (3) copies of a Petition for Approval to Transfer Control filed on behalf of LDMI Telecommunications, Inc. d/b/a LDMI Telecommunications also d/b/a FoneTel ("LDMI"). LDMI is certified as a resale telecommunications carrier in the state of New York. The stock of LDMI is being acquired by Talk America Inc., which also holds authority to provide telecommunications service in New York. After the transaction, LDMI will operate as a separate, wholly owned subsidiary of Talk America Inc. LDMI will retain its separate operating authority and will not change its name nor its rates as a result of this transfer; accordingly, the transaction will be transparent to its New York customers. Details of this transaction are contained in the attached Petition.

Questions regarding this filing may be directed to my attention at (407) 740-3031 or via email at sthomas@tminc.com.

Please acknowledge receipt of this filing by date stamping the extra copy of this cover letter and returning it in the self-addressed, stamped envelope enclosed for this purpose.

Sincerely,

Sharon Thomas Consultant

Enclosure

cc: B. Shires – LDMI C. Pizer - Talk File: LDMI – New York Talk – LDMI Acquisition TMS: NYx0501



BEFORE THE DEPARTMENT OF PUBLIC SERVICE STATE OF NEW YORK

Petition of LDMI Telecommunications, Inc. dba LDMI Telecommunications also dba FoneTel for Authority to Transfer Control

Docket No. _____

PETITION

)

)

)

Pursuant to NY CLS Pub. Ser. § 100, LDMI Telecommunications, Inc. d/b/a LDMI Telecommunications also d/b/a FoneTel, a Michigan corporation, ("LDMI" or "Applicant") hereby seeks approval of a transfer of control stock transaction between Applicant and Talk America Inc., a Pennsylvania corporation, ("Talk America" or "Buyer"). Through the proposed transaction ("Transaction"), Applicant will become a wholly-owned subsidiary of Buyer, but will continue to operate as a separate entity. Applicant is currently certified to resell all forms of telecommunications services in New York and will retain its status as a telecommunications also d/b/a FoneTel in New York after the Transfer.

Applicant respectfully requests that the Commission issue an order approving the Transfer. In support of this Application, Applicant provides the following summary information:

Applicant was certificated to provide resold telecommunications service in New York on April 7, 1997, in Case No. 97-C-0018. Upon consummation of the Transaction, LDMI will become a wholly owned subsidiary of Talk America Inc., whose parent company is Talk America Holdings, Inc., a Delaware corporation. Applicant will remain a separate operating company in New York, retaining its name and telecommunications operating authority.

> Petition for Authority to Transfer Control LDMI Telecommunications, Inc. Page 1

I. THE BUYER

The Buyer, Talk America Inc., is a Pennsylvania corporation headquartered in New Hope, Pennsylvania. Buyer is a direct wholly owned subsidiary of Talk America Holdings, Inc. ("Talk Holdings"), a publicly held Delaware corporation located at the same address. No entity or individual directly or indirectly owns 10% or more of the equity of Talk Holdings. Buyer was granted authority to provide telecommunications service New York on April 27, 1992, in Case No. 92-C-0222. Buyer provides both local and long distance services in New York A copy of Buyer's articles of incorporation is included as *Exhibit 1*.

Talk America is authorized to provide competitive local and long distance service virtually nationwide and offers a bundle of these phone services to residential and small business customers. Talk America has approximately 591,000 billed bundled lines in 33 states and 100,000 billed long distance customers across the United States and has annual revenues in excess of \$470 million.

II. DESCRIPTION OF THE TRANSACTION

On May 23, 2005, the parties signed an Agreement and Plan of Merger (the "Agreement") providing for Talk America's acquisition of all of the issued and outstanding capital stock of LDMI (the "Transaction"). A copy of the Agreement is attached as *Exhibit 2*. At the closing of the transaction, in exchange for all of the stock of LDMI, Talk America will pay \$24 million in cash and issue 1.8 million shares of its common stock, with an estimated value of \$16 million based on the closing price as of May 20, 2005. As of April 30, 2005, LDMI had total debt, net of cash, of \$8.1 million. The purchase price for LDMI's equity together with LDMI's current debt, net of cash, totals \$48 million. The cash portion of the purchase price, as well as the repayment of LDMI's debt at closing, will be funded by Buyer out of cash on hand. Following completion of the Transaction, LDMI will be a direct, wholly owned subsidiary of Talk America, which itself will remain a direct, wholly owned subsidiary of Talk Holdings. Closing of the Transaction is contingent upon, among other things, receipt of necessary regulatory approvals.

III. CUSTOMER IMPACT

The proposed Transaction will be generally transparent to customers of LDMI. In particular, because the Transaction is a stock transaction, the transfer of control of LDMI will not result in a change of carrier for any of LDMI's customers. Immediately after completion of the Transaction, LDMI will continue to provide the same services at the same rates, terms and conditions as at present and any future changes in rates, terms and conditions will be made consistent with applicable law and Commission rules. The Transaction is not expected to result in any discontinuance of service for LDMI customers. Moreover, in addition to having access to Buyer's experienced and high-quality management team, the operations of LDMI in numerous instances will continue to be supervised by LDMI's management, technical and customer service personnel.

IV. PUBLIC INTEREST STATEMENT

The parties respectfully submit that the Transaction serves the public interest. After consummation of the Transaction, LDMI will continue to operate under its same name and operating authorities as at present. The Transaction involves no change in the entity providing service to customers, the facilities used to provide such service or the services, rates, terms and conditions of such service. All existing tariffs will remain in place. The transfer of control will be generally transparent to customers and will not have any adverse impact on them; they will continue to use the same number to reach customer service. The only change will be in the ultimate ownership of LDMI. The contemplated transfer of control does not raise any slamming concerns or necessitate compliance with FCC or state procedures to notify customers prior to a carrier-to-carrier sale or transfer of a subscriber base as the Transaction does not involve any change in a customer's existing service provider. LDMI customers will continue to be served under LDMI's existing authorization and tariff.

The parties expect that the Transaction will increase competition in the telecommunications market by strengthening Talk's position as an effective and multifaceted telecommunications carrier.

The Transaction will allow Talk to combine its financial, technical and market resources and expertise with that of LDMI, thereby enhancing its ability to provide reliable, competitively priced services to customers. Talk will be able to expand its product offerings by adding the offerings of LDMI to its current offerings.

Both Talk and LDMI have created strong customer service organizations that provide high levels of customer satisfaction as demonstrated by their low customer churn rates. Both organizations share similar technology platforms and LDMI's focus on the small to medium sized business customers will complement Talk's focus on the consumer and small business markets. In an era of increasing telecommunications products and services, these customers require expert assistance to find the best solution for their telecommunications requirements. The similarity of business models between Talk and LDMI means that integrating their operations will not occasion significant changes in what either company does and thus the Transaction poses little risk to customers.

V. CONTACT INFORMATION

E 42

The name and address of the individual to contact regarding this filing is:

Sharon Thomas Technologies Management, Inc. 210 Park Avenue North Winter Park, Florida 32789 Telephone: (407) 740-3031 Facsimile: (407) 740-0613 email: sthomas@tminc.com

The name and address of the individual who handles regulatory affairs on behalf of LDMI is:

Jerry Finefrock, Vice President of Regulatory Affairs 27777 Franklin Road Suite 500 Southfield, MI 48034 Phone: (248) 440-4021 Fax: (866) 731-0450 e-mail: jfinefro@ldmi.com

> Petition for Authority to Transfer Control LDMI Telecommunications, Inc. Page 4

The name and address of the individual who handles regulatory affairs on behalf of Talk America is:

> Francie McComb, Vice President of Regulatory Affairs 6805 Route 202 New Hope, PA 18938 Phone: 215-862-1517 Fax: 215-862-1085 e-mail: francie@talk.com

A complete listing of officers and directors for Applicant and Buyer is attached as Exhibit 3.

WHEREFORE, for the reasons stated above, Applicant and Buyer submit that the public interest, convenience and necessity would be furthered by a grant of this Application and respectfully request that the Commission approve the Transfer.

Respectfully Submitted,

Sharon Thomas Consultant to LDMI Telecommunications, Inc. and Talk America Inc.

Petition for Authority to Transfer Control LDMI Telecommunications, Inc. Page 5

VERIFICATION

I, Patrick O'Leary, am President and CEO of LDMI Telecommunications, Inc. and am authorized to represent it and to make this verification on its behalf. The statements in the foregoing document relating to this company, except as otherwise specifically attributed, are true and correct to the best of my knowledge and belief.

I declare under penalty of perjury that the foregoing is true and correct.

Patrick O'Leary President and Chief Executive Officer

Sworn to and subscribed before me this 3 day of _____ __, 2005.

rarnowsh

Notary Public

My Commission expires:

LAURA CZARNOWSKU Noteny Petrie, Wayne County, Mi New Counterston Bornes Dec. 23, 2005

VERIFICATION

I, Aloysius T. Lawn IV, am Executive Vice President and General Counsel of Talk America Holdings, Inc. and am authorized to represent it and its subsidiaries and to make this verification on their behalf. I have read the foregoing Notice; and the statements contained therein related to this company and its subsidiaries, except as otherwise specifically attributed, are true and correct to the best of my knowledge and belief.

I declare under penalty of perjury that the foregoing is true and correct.

Aloysius A. Lawn IV Executive Vice President and General Counsel

Sworn to and subscribed before me this <u>Dul</u> day of <u>June</u>, 2005.

y McCusker

Notary Public

•

My Commission expires:

5120108

COMMONWEALTH OF PENNSYLVANIA Notarial Seal

Amy McCusker, Notary Public New Hope Boro, Bucks County My Commission Expires May 20, 2008

Member, Pennsylvania Association Of Notaries

Talk America Inc. (Buyer)

×, *,

•

Exhibit 1

ARTICLES OF INCORPORATION

.

ARTICLES OF INCORPO	RATION	DOMESTIC TU	SINELE CORPORATION	FEE
			INEST CORPORATION	
DEPARTMENT OF STATE - CORPOR			Ponation - complete b of estimate or ponational	
SOS NORTH OFFICE BUILDING, HARR	188UR0, PA 17170	ENITED WAD	D LIDENSE NO.	
TEL-SAVE, JAC.		 A second contraction 		
OIL ADDALLA OF ALGINIANS OFFICE				
Willow Grove	• · · · · · · · ·	Montgomery	PA PA	19090
DIO EXPLAIN THE PURPOSE OR PURPOS			्राव्याः ध्रीत्रद्वान् क्रु	清洁 和 [13]
The purposes for which		The second s		do any
lawful husiness for whi				
Pennsylvania Business C	orporation.L	NATION AND A REAL	lands fri gi topre anala y. Subtrite sugerier i dia matri	1 34 presto par (
	999 - 999 - 999 - 999 - 9 2019 - 975 - 975 - 976 - 976 2019 - 975 - 975 - 976 - 976 - 976	n geografie an	an an airtean an a	142397, 1998 1. 14
	an a	rentan e sen e serie e Fentan e serie	an a	n a the part of the second
	ي و المعلق العرب من المراجع (المعلي). 	etas de construcción	- 144 - 154 - 144 - 154	. A timber f
	an Bat A Marka	្នុំ ព័ត៌មកសំណាមសំណ៍សំព័ត៌សំពីសំណែង	6.313	en et per de a se ave
IATTACH EN A 11 SHEET IF NEOESSARY			and the second	
The Appropriate Horney of Barres, Direct of Sha		Bilin Pat Voice Par		
949 Number and Cleat of Shares 5000 Cc		W ANY 'N/A	STU DOO	perpicual
The Name and Address of Bach Interportion,		lei Bhirs Subscribed to b	y sich freerricewit	
COS Nume	DET OFT ABOVE	(Sweet, City, 8	uis, Zip Code)	Number & Class of Prov
COS Nume	DET OFT ABOVE	(Sweet, City, 8		
COS Nume	DET OFT ABOVE	(Sweet, City, 8	uis, Zip Code)	
COS Nume	DET OFT ABOVE	(Sweet, City, 8	uis, Zip Code)	
COS Nume	D.7 North Yo	(Svert City, S ck. Rond. Unit	us. 27.0001 1 • Willow Grov PA 19090	
Did Hume Daniel Rorislow	D17 North Yo	(Svinc Oin) (ck. Road. Unde	UN. 27 Coon N 11 ov Crov PA 19090	1000 common
COS Nume	AAAAAAAAAAAAAAAAAAAAAAAAAAAAAAAAAAAAAA	(STORE ON) S Ek. Road. Unit Antshest if Necess Sthays inghed an	UIS. 27 Cook	1000 common
DID Nore Bantel Borislow	INCORPORATOR (II H	(STORE ON) S Ek. Road. Unit Antshest if Necess Sthays inghed an	UIS. 27 Cook	1000 common
DID Nore Bantel Borislow	INCORPORATOR (II H	(STORE ON) S Ek. Road. Unit Antshest if Necess Sthays inghed an	UIS. 27 Cook	1000 common
Daniel Borislow	INCORPORATOR (II H	(STORE ON) S Ek. Road. Unit Antshest if Necess Sthays inghed an	UIS. 27 Cook	1.000 common
Bantel Rorislow	BELOGI DIJONI AAAAAA DIJONICH YO IATTACHEN NCORVONATOR (4 M DATOR MAY	(STING ONLY-	AATYI Declar THE ANTICINE DO DO DO DO DO DO DO DO DO DO	1.000 common
Bantel Borislow	COLORATOR (4 HAVE NCOLORATOR (4 HAVE DATOR MAY SODE	ISTORE ONLY	UII. 27 Cool I MITOV Crox PA 19090 AATT Desaled the Articles U U U U U U U U U U U U U	1.000 common
DIO FILED MAY 1 1 1989	Beil, Cell SIS, CHI Address D17, North Yo IATTACHEN IATTACHEN NOOPPORATOR (d) H. DAY OF MAY SODE SODE SWED BY GON B	ISTING UND ONLY-	AATT DEEALED THE ANTICLES DULL DULL TIAL HO. INCOMEST	1.000 common
DIO FILED MAY 1 1 1989	СС 1 СС 1 СС 2 СС 4 Астин ()) 7 North Yo (АТТАСН 8) КОСНИОВАТОВ (d) Нау СС 1 СС 1	ISTING UND ONLY-	HIS. 20 COM I MILON Crox PA 19090 AANI SEALED THE ANTICLES IN MALED THE ANTICLES MALED THE ANTICLES	000 common
Des FILED MAY 1 1 1989	СС / СС / Аления CO / Колоналон (с) СО / Колоналон (с) СОС / Колоналон (с) СО / Колон (с)	ISTING UND ONLY-	UII. 20 COMMINSTREES	000 common
BANTEL BOTISION IN TESTIMONY WHEREOF, THEN THIS FITAL BOOFFICED MAY 1 7 1989 Jan J. 2407 SATI	ССТ, ССТ, Алиния СО, ССТ, Кон Алиния СО, ССТ, Кон Кон Кон Саттасная Саттасная Паттасная Саттасная Патт	ISTING USE ONLY	UII. 21 Cool I . PA 19090 AATI COLED THE ANTICLIES COLED THE ANTICLES COLED THE A	1000 common от послатон 99372698 Атюн мания 07863 Соз ти и и и и и
BANTEL BOTISION IN TESTIMONY WHEREOF, THEN THIS FITAL BOOFFICED MAY 1 7 1989 Jan J. 2407 SATI	01.7 North Yo 01.7 North Yo 01.7 North Yo IATTACHE IATTACHE NCONVORATOR (d) H IAY SX01 IAY	ISTER CONTRACTOR	HIS. 29 Cool I MILLOV GEON ANYI SEALED THE ANTICLES MALED THE	000 common
BANTEL BOTISION IN TESTIMONY WHEREOF, THEN THIS FITAL BOOFFICED MAY 1 7 1989 Jan J. 2407 SATI	01.7 North Yo 01.7 North Yo 01.7 North Yo IATTACHE IATTACHE NCONVORATOR (d) H IAY SX01 IAY	ISTING USE ONLY- TOPPICS O	UII. 21 Cool I . PA 19090 AATI COLED THE ANTICLIES COLED THE ANTICLES COLED THE A	1000 common от послатон 99372698 Атюн мания 07863 Соз ти и и и и и

۰.

۰.

			0000
i. The following provisions shall regulat	is the status of the porporation	A time provider	699
(s) · (Strike out(i) or (ii) below, whi	chever is not applicable.)		** • • • • • • • • •
(1) All of the issued shares of not niors than 10		Exclusive of treasury share	as shill be hold of second by
(ii) - All of the lasued there of not mote than the smaller LRevenue Code of 1954, as		sclusive of freesury share within the meaning of t	ss, shall be held of second by- Subchapter 3 of the internal
(b) All of the lasted there of all class permitted by section 613.1 of the	ises of the corporation shall be a Business Corporation Law (1	subject to one or more of \$ P.B. § 1617.1).	The restrictions on transfer
(c) The corporation shall make no of within the meaning of the Securi	Mering of any of its shares of its shares of its Act of 1933, as smanded.	any class which would co	natitute a "public offering"
. IOptionat: BCL \$ 3728) A person (othin eliher case within the meaning Of the in holder of racord of shares of the correc corporation to be treated as an electing - has not affermatively refused to twince in - shourd of second of shares of the corpora- shourd of second of shares of the corpora-	oration. Only a person without small builters accorporation und the the election within stary d wration.	onsent li currenty in e fer Subchapler 3 of the Ca tra aller he acquires ble	englind) be enillied to be g flect to the election of the colorand weberchelder whe eleck enall be untitled to be
(Optional: BCL ()#2) The business and tather than by a board of directors.	fallatis of the corporation shall	t be managed by the shar	sholders of the corporation
(Optional: (3768) The status of the col Law shall not be jerminated without th	horstinn se a "close corporation he smrmstive vote of writer of the) theres of all cl	. PROCEERS OF INTERNATION	A The state of the
Optional BCL (JSIAL (Au starshold		of the shar	ei) of the corporation may
apply for the appointment of a provisione	director of the corporation in	the menner and upon the	sireumatances provided by -
(Optionali BCL 1 J86) (Any shereholder	1 feherebolderr boldine 10	01	
have the right at will to cause the corport			a) of the corporation shall vided by statute.
	and the second second second		
Anna All Marin ana ana	TANANON IN THE STATE	an a terministic and the state	

. <u>.</u> .

•

}. ∙ ₽

12

.+

÷ 1.001 10

5.4.9

 $\mathbf{x}_{i} \in \mathbf{X}$ uiita :ita : · . an grata And transforms A she was . arystern. T MILL MORE AND F

COMMONWEALTH OF PENNSYLVANIA



April 10, 1992 Department of State.

TO ALL WHOM THESE PRESENTS SHALL COME, GREETING:

IN RE: "TEL-SAVE, INC."

SCB-60

I, Dr. Brenda K. Mitchell, Secretary of the Commonwealth of the Commonwealth of Pennsylvania do hereby certify that the foregoing and annexed is a true and correct photocopy of Articles and Certificate of Incorporation

which appear of record in this department,



IN TESTIMONY WHEREOF, I have hereunto set my hand and caused the Seal of the Secretary's Office to be affixed, the day and year above written.

..

. • · · ·

> н. н. М. (М. (1996)) Стала (1996)

> > 1. 1.14

. ...

.

Sec. Sec. 1.

• • • • • • •

•••••

an an taon an an taon an taon an taon an tao an taona an tao an taona an tao an tao an tao Acres 6

1.88. 2.10

· ...

1

1. 11₁₀ 1

Sec. 2. Sec.

Adda St.

Secretary of the Commonwealth dp

Commonwealth of Pennsylbania



Department of State CERTIFICATE OF INCORPORATION

OFFICE OF THE SECRETARY OF THE COMONWEALTH

TO ALL TO WHON THESE PRESENTS SHALL CONE, GREETING:

WHEREAS, UNDER THE PROVISIONS OF THE LAWS OF THE CONMONNEALTH, THE SECRETARY OF THE CONMONNEALTH IS AUTHORIZED AND REQUIRED TO ISSUE & "CERTIFICATE OF INCORPORATION" EVIDENCING THE INCORPORATION OF AN ENTITY.

WHEREAS, THE STIPULATIONS AND CONDITIONS OF THE LAN HAVE BEEN FULLY CONFLIED WITH BY

TEL-BAVE, INC. A CLOSE CORPORATION

THEREFORE, KNOW YE, THAT SUBJECT TO THE CONSTITUTION OF THIS COMMONWEALTH, AND UNDER THE AUTHORITY OF THE LAWS THEREOF, I DO BY THESE PRESENTS, WHICH I HAVE CAUSED TO BE SEALED WITH THE GREAT SEAL OF THE COMMONWEALTH, DECLARE AND CERTIFY THE GREATION, ERECTION AND INCORPORATION OF THE ABOVE IN DEED AND IN LAW BY THE NAME CHOSEN HEREINBEFORE SPECIFIED.

BUCH CORPORATION SHALL HAVE AND ENJOY AND SHALL BE SUBJECT TO ALL THE POWERS, DUTIES, (REQUIREMENTS, AND RESTRICTIONS, SPECIFIED AND ENJOINED IN AND BY THE APPLICABLE LAWS OF (THIS COMMONWEALTH.



GIVEN UNDER NY HAND AND THE GREAT SEAL OF THE CONNONWEALTH, AT THE CITY OF HARAISBURG, THIS 17TH DAY OF MAY IN THE YEAR OF OUR LORD ONE THOUSAND NINE HUNDRED AND EIGHTY-NINE AND OF THE CONSOMMEALTH THE TWO HUNDRED THIRTEENTHS

ONWEALTR

AATE MURPHY D'ANIELLO & PARKELL ATTNI RICHARD B MORELL 43 EAST MARBHALL BT NORRISTONN ... PA 19401-0000

307863 08937 0698-0700

Microfilm Number	7562- 1 Siles	with the Department	of State on	SER LL 100
Enter Number 1507963		(1 10	1
		Sected	Ary of the Comm	onwealth 1911
	ENDMENT-ELECTION			
	ATUTORY CLOSE COP	2.8 Figure 11, 1997 Constraints and the second sec second second sec		
	OF TEL-SAVE, II	łC.		
A	Name of corport STATUTORY CLOSE C DSCB:15-2305 (Ref	ORPORATION		
In compliance with the requirement corporation to terminate its status as a statu	itory close corporation and to	become a business c	orporation), the i	indersigned statutory
close corporation, desiring to amend its Art become a business corporation, hereby stat		erminete its slaws es t		
1. The name of the corporation is: <u>Tel-s</u>	Seve: Inc.			
 The (a) address of this corporation's c office provider and the county of venue is conform to the records of the Department 	(the Department is hereby a	s Commonwealth or (I .thorized to correct the	 name of its co s following inform 	mmercial registered nation to
(a) 22 Milage Square	New Hope	Pennsylvania	18938	Bucks
Number and Street	City	State	Zp	County
C/o: Name of Commercial Registered Office Provid				County
For a corporation represented by a commercial corporation is located for venue and official public	registered office provider; the cou	nty in (6) shall be deeme	d the county in whic	
For a corporation represented by a commercial corporation is located for venue and official public structure and structu	registered office provider, the cou Ication purposes			h De
For a corporation represented by a commercial	registered office provider, the cou Ication purposes			h De
For a corporation represented by a commercial corporation is located for venue and official public structure and structu	registered office provider, the cou ication purposes porated is: <u>Pennsylvania</u> B			h De
For a corporation represented by a commercial corporation is located for venue and official publication of the statute by or under which it was incom	registered office provider, the coulocation purposes. porated Is: <u>Pennsylvania P</u>			h De
For a corporation represented by a commercial corporation is located for venue and official publes. The statute by or under which it was incorporation is: <u>May 17</u>	registered office provider, the could the could be a set of the following):	usiness Corporation L	aw c(1988 (15	h De
For a corporation represented by a commercial corporation is located for venue and official publes. 3. The statute by or under which it was incorporation is:	registered office provider, the could is in purposes. porated is: <u>Pennsylvania B</u> 1989 e of the following): pon filing these Articles of Ame	usiness Corporation L	ent of State.	h De Pa.C:S: 2301 et sec:)
For a corporation represented by a commercial corporation is located for venue and official publes. 3. The statute by or under which it was incorporated in the statute of its incorporation is: <u>May 17</u> 4. The date of its incorporation is: <u>May 17</u> 5. (Check, and if appropriate complete, on <u>X</u> The amendment shall be effective up The amendment shall be effective or	registered office provider, the could is <u>Pennsylvania 1</u> 1989 e of the following): fon filling these Articles of Ame 1. <u>St</u> Date	ysiness Corporation L	ert of State. Hou	h De Pa:C:S: 2301 et seq:)
For a corporation represented by a commercial corporation is located for venue and official publes. 3. The statute by or under which it was incorporation is: <u>May 17</u> 4. The date of its incorporation is: <u>May 17</u> 5. (Check, and if appropriate complete, one <u>X</u> The amendment shall be effective up The amendment shall be effective or 6. The amendment was adopted by the share 7. The corporation elects to voluntarily terminal	registered office provider, the cou- lication purposes porated is: <u>Pennsylvania E</u> 1989 e of the following): fon filing these Articles of Ame h	usiness Corporation L	ent of State. Hou	h be Pa:C:S: 2301 et seq:)
For a corporation represented by a commercial corporation is located for venue and official publes. The statute by or under which it was incorporation is: May 17 May 17 (Check, and it appropriate complete, on	registered office provider, the cou- lication purposes: porated Is: <u>Pennsylvania E</u> 1989 e of the following): non filing these Articles of Ame n. <u>st</u> Date holders (or members) pursuan ate its status as a statutory clos	usiness Corporation L	ent of State. Hou	h be Pa:C:S: 2301 et seq:)

See Exhibit "A" attached hereto.

79 . 1944 The amendment adopted by the corporation is set forth in full in Exhibit A attached hereto and made a part hereof.

SEP 22-95.

to St

No net

gatures if the anendment maining the Articles :

21. 5.

The restand Articles of incorporation supercode the original Articles and all amondments thereis.

IN TESTIMONY WHEREOF, the undersigned corporation has eaused these Articles of Amendment to be signed by a duy supported effect thereof the 22nd day of September, 1915.

9562-151

AVE INC. THE LIN) TITLE CEO Dan 1

EO a

C 0

92:20

'Z Z

56

EXHIBIT A

9542 - 152 The Corporation hereby voluntarily terminates its status as a statutory close corporation pursuant to 15 Pa.C.S. 2307. The heading of the Articles of incorporation stating that the Corporation is a statutory close corporation shall be deleted in the entirety. Articles 1, 3, 4 and 6 of the Articles of incorporation shall also be deleted in their entireties.

Microfilm Number 9941- 755	1	d with the De-	partment of State	MAY 2 4 199
Entity Number_ 1507843		u whu ulebe	W Jugge	-gulle
		Secretary	of the Commonw	éalth
ARTICLES OF AM	ENDMENT-DOMEST		CORPORATION	
	DSCB:15-1915 (H	(cv 90)		
In compliance with the requirement pusiness corporation, desiring to amend it	ts of 15 Pa.C.S. § 1915 ts Articles, hereby state	(relating to art s that:	ficles of amendm	ent), the undersigned
. The name of the corporation is: Te	≥1-Save, Inc.			
. The (a) address of this corporation	n's current registered	office in this C	Commonwealth o	(b) name of its
commercial registered office provid correct the following information to	ler and the county of	venue is (the I	Department is he	reby authorized to
(a) <u>6805 Route</u> 202,	New Hope,	PA	18938	Bucks
	City	State	Zip	County
Number and Street		自己的法律的法律的		
(b) c/o:				
(b) c/o: Name of Commercial Regist	tered Office Provider			County
(b) c/o:	tered Office Provider mmercial registered of	fice provider, ti d official publ	he county in (b) si Leation purposes.	all be deemed
(b) c/o: Name of Commercial Regist For a corporation represented by a co	tered Office Provider mmercial registered of is located for venue an	d official publ	ication purposes.	all be deemed
(b) c/o: Name of Commercial Regist For a corporation represented by a co the county in which the corporation	tered Office Provider ommercial registered of is located for venue an s incorporated is: Per	d official publ	ication purposes.	all be deemed
(b) c/o: Name of Commercial Regist For a corporation represented by a co the county in which the corporation The statute by or under which it was	tered Office Provider mmercial registered of is located for venue an s incorporated is: Per v 17, 1989	d official publ	ication purposes.	all be deemed
(b) c/o: Name of Commercial Regist For a corporation represented by a co the county in which the corporation The statute by or under which it was The date of its incorporation is: Ma (Check, and if appropriate complete	tered Office Provider mmercial registered of is located for venue an s incorporated is: <u>Per</u> y 17, 1989 , one of the following):	dofficial publ	Leation purposes.	all be deemed
(b) c/o: Name of Commercial Regist For a corporation represented by a co the county in which the corporation The statute by or under which it was The date of its incorporation is: <u>Ma</u> (Check, and if appropriate complete <u>×</u> The amendment shall be effective	tered Office Provider mmercial registered of is located for venue an s incorporated is: <u>Per</u> <u>v 17, 1989</u> , one of the following): upon filling these Arti	dofficial publ	Leation purposes.	all be deemed
(b) c/o: Name of Commercial Regist For a corporation represented by a co the county in which the corporation The statute by or under which it was The date of its incorporation is: Ma (Check, and if appropriate complete Mathematical be effective The amendment shall be effective	tered Office Provider mmercial registered of is located for venue an s incorporated is: <u>Per</u> <u>v 17, 1989</u> , one of the following): upon filling these Arti	dofficial publ	Leation purposes.	nall be deemed
(b) c/o: Name of Commercial Regist For a corporation represented by a co- the county in which the corporation The statute by or under which it was The statute by or under which it was The date of its incorporation is: Ma (Check, and if appropriate complete X The amendment shall be effective (Check one of the following):	tered Office Provider mmercial registered of is located for venue an s incorporated is: Per v 17, 1989 , one of the following); upon filing these Arti on Date	dofficial publ maylvanta B cles of Amend 81	ication purposes; usiness corpo ment in the Depa Ho	nall be deemed
(b) c/o: Name of Commercial Regist For a corporation represented by a co- the county in which the corporation The statute by or under which it was The date of its incorporation is: Ma- (Check, and if appropriate complete X The amendment shall be effective (Check one of the following): X. The amendment was adopted by the	tered Office Provider mmercial registered of is located for venue an s incorporated is: <u>Per</u> <u>y 17, 1989</u> , one of the following): upon filing these Arti on Date the shareholders (or men	dofficial publ msylvania B cles of Amenda abers) pursuan	ication purposes; usiness Corpo ment in the Depa Ho to 15 Pa.C.S. §4	nall be deemed
(b) c/o: Name of Commercial Regist For a corporation represented by a co- the county in which the corporation The statute by or under which it was The date of its incorporation is: Ma (check, and if appropriate complete A The amendment shall be effective —The amendment shall be effective (check one of the following): A The amendment was adopted by th _The amendment was adopted by th	tered Office Provider mmercial registered of is located for venue an s incorporated is: <u>Per</u> <u>v 17, 1989</u> , one of the following): upon filling these Artis on <u>Date</u> at shareholders (or men the board of directors p	dofficial publ msylvania B cles of Amenda abers) pursuan	ication purposes; usiness Corpo ment in the Depa Ho to 15 Pa.C.S. §4	nall be deemed
(b) c/o: Name of Commercial Regist For a corporation represented by a co- the county in which the corporation The statute by or under which it was The date of its incorporation is: Ma (Check, and if appropriate complete X. The amendment shall be effective (Check one of the following): X. The amendment was adopted by th The amendment was adopted by th	tered Office Provider mmercial registered of is located for venue an sincorporated is: Per y 17, 1989 , one of the following): upon filing these Artis on Date the shareholders (or men the board of directors p one of the following):	dofficial publ <u>insylvania B</u> cles of Amendi abers) pursuan wirsuant to 15	ication purposes. <u>utsiness Corpo</u> ment in the Depa Ho to 15 Pa.C.S. §1 Pa.C.S. §1914(c)	nall be deemed
(b) c/o: Name of Commercial Regist For a corporation represented by a co- the county in which the corporation The statute by or under which it was The date of its incorporation is: Ma (check, and if appropriate complete A The amendment shall be effective (check one of the following): A The amendment was adopted by th The amendment was adopted by th (check, and if appropriate complete, (check, and if appropriate complete, The amendment was adopted by th (check, and if appropriate complete, A The amendment adopted by the co-	tered Office Provider mmercial registered of is located for venue an s incorporated is: <u>Per</u> <u>v 17, 1989</u> , one of the following): upon filling these Artis on <u>Date</u> at shareholders (or men the board of directors p one of the following): opporation, set forth in	d official publ <u>insylvanta</u> B cles of Amenda abers) pursuan sursuant to 15 full, is as follo	ication purposes, uisiness Corpo ment in the Depa Ho to 15 Pa.C.S. §1914(c) Pa.C.S. §1914(c) ws:	nall be deemed
(b) c/o: Name of Commercial Regist For a corporation represented by a co- the county in which the corporation The statute by or under which it was The date of its incorporation is: Ma (Check, and if appropriate complete X. The amendment shall be effective (Check one of the following): X. The amendment was adopted by th The amendment was adopted by th	tered Office Provider mmercial registered of is located for venue an s incorporated is: <u>Per</u> <u>v 17, 1989</u> , one of the following): upon filling these Artis on <u>Date</u> at shareholders (or men the board of directors p one of the following): opporation, set forth in	d official publ <u>insylvanta</u> B cles of Amenda abers) pursuan sursuant to 15 full, is as follo	ication purposes, uisiness Corpo ment in the Depa Ho to 15 Pa.C.S. §1914(c) Pa.C.S. §1914(c) ws:	nall be deemed
(b) c/o: Name of Commercial Regist For a corporation represented by a co- the county in which the corporation The statute by or under which it was The date of its incorporation is: Ma (check, and if appropriate complete A The amendment shall be effective (check one of the following): A The amendment was adopted by th The amendment was adopted by th (check, and if appropriate complete, (check, and if appropriate complete, A The amendment was adopted by th (check, and if appropriate complete, A The amendment adopted by the co-	tered Office Provider mmercial registered of is located for venue an s incorporated is: <u>Per</u> <u>v 17, 1989</u> , one of the following): upon filling these Artis on <u>Date</u> at shareholders (or men the board of directors p one of the following): opporation, set forth in	d official publ <u>insylvanta</u> B cles of Amenda abers) pursuan sursuant to 15 full, is as follo	ication purposes, uisiness Corpo ment in the Depa Ho to 15 Pa.C.S. §1914(c) Pa.C.S. §1914(c) ws:	nall be deemed

MAY 24 State The State

-

Pt.d

. Tr. Th

19-Nay-99.04:

9941 - 756

DSC0:15-1915 (Rev 90)-2

8.

(PA. - 1427)

The restated Articles of Incorporation supersede the original Articles and all amendments therein.

IN TESTIMONY WHEREOF, the undersigned corporation has caused these Articles of Amendment to be signed by a duly surborized officer the real this 200 day of May 1999

Tel-Save. Inc. (Name of Corporation) Plapis 7 Law 1 IY: 1 (Signature)

THIR Aloysius T. Lawn IV, Secretary

Microfilm Number

APR 1 1 20 the Department of State on

of 1988

Hour

Entity Number

ARTICLES OF AMENDMENT-DOMESTIC BUSINESS CORPORAT DSCB:15-1915 (Rev 90)

Filed y

In compliance with the requirements of 15 Pa.C.S. § 1915 (relating to articles of amendment), the undersigned business corporation, desiring to amend its Articles, hereby states that:

Corp.

The (a) address of this corporation's current registered office in this Commonwealth or (b) name of its commercial registered office provider and the county of venue is (the Department is hereby authorized to correct the following information to conform to the records of the Department):

- (a) 1017 North York Road, Unit 1, Willow Grove, PA 19090
- Number and Street City State Zip County
- (b) c/o: 10.05 County

Name of Commercial Registered Office Provider

For a corporation represented by a commercial registered office provider, the county in (b) shall be deemed the county in which the corporation is located for venue and official publication purposes.

- The statute by or under which it was incorporated is: Pennsylvania Business Corporation Law
- 1 The date of its incorporation is: May 177 1989
 - (Check, and if appropriate complete, one of the following):
 - X_The amendment shall be effective upon filing these Articles of Amendment in the Department of State. The amendment shall be effective on at

Date

- (Check one of the following):
 - X. The amendment was adopted by the shareholders (or members) pursuant to 15 Pa.C.S. § 1914(a) and (b).
 - The amendment was adopted by the board of directors pursuant to 15 Pa.C.S. § 1914(c).
- (Check, and if appropriate complete, one of the following): 7.
 - X The amendment adopted by the corporation, set forth in full, is as follows:
 - "The name of the corporation is : Talk America Inc.

THIS IS A TRUE COPY OF THE ORIGINAL SIGNED DOCUMENT FILED WITH THE DEPARTMENT OF STATE

- 1427 - 10/9/92

3.

5.

6.

DSCB:15-1915 (Rev 90)-2

8.

_____The amendment adopted by the corporation as set forth in full in Exhibit A attached hereto and made a part hereof.

The restated Articles of Incorporation supersede the original Articles and all amendments thereto.

IN TESTMONY WHEREOF, the undersigned corporation has caused these Articles of Amendment to be signed by a didy a d

Talk, com Holding Corp.

(Name of Corponetion) Aloystus avu. IV (Simulare)

TILE Executive Vice President. General Counsel & Secretary Exhibit 2

· · · .

AGREEMENT AND PLAN OF MERGER

.



FORM 8-K

TALK AMERICA HOLDINGS INC - TALK

Filed: May 24, 2005 (period: May 23, 2005)

Report of unscheduled material events or corporate changes. e.g acquisition bankruptcy resignation

Table of Contents

W.

Item 1.01 Entry into a Material Definitive Agreement.

Item 7.01 Regulation FD Disclosure

Item 9.01 Financial Statements and Exhibits

SIGNATURES EXHIBIT INDEX EX-10.1 (Material contracts)

۰,

х.,

EX-99.1 (Exhibits not specifically designated by another number and by investment companies)

UNITED STATES SECURITIES AND EXCHANGE COMMISSION Washington, D.C. 20549

FORM 8-K

CURRENT REPORT

Pursuant to Section 13 or 15(d) of the Securities Exchange Act of 1934

Date of Report (Date of earliest event reported): May 23, 2005

Talk America Holdings, Inc.

(Exact name of registrant as specified in its charter)

Delaware (State of incorporation) 000 - 26728

23-2827736 (Commission File Number) (I.R.S. Employer Identification No.)

12020 Sunrise Valley Drive, Suite 250, Reston, Virginia 20191 (Address of principal executive offices) (Zip Code)

(703) 391-7500

(Registrant's telephone number, including area code)

N/A

(Former name or former address, if changed since last report)

Check the appropriate box below if the Form 8-K filing is intended to simultaneously satisfy the filing obligation of the registrant under any of the following provisions (see General Instruction A.2. below):

[] Written communications pursuant to Rule 425 under the Securities Act (17 CFR 230.425)

[] Soliciting material pursuant to Rule 14a-12 under the Exchange Act (17 CFR 240.14a-12)

[] Pre-commencement communications pursuant to Rule 14d-2(b) under the Exchange Act (17 CFR 240.14d-2(b))

[] Pre-commencement communications pursuant to Rule 13e-4(c) under the Exchange Act (17 CFR 240,13e-4(c))

Item 1.01 Entry into a Material Definitive Agreement.

On May 23, 2005, Talk America Holdings, Inc. and a subsidiary of Talk America entered into an Agreement and Plan of Merger (the "Acquisition Agreement") with LDMI Telecommunications, Inc., pursuant to which Talk America will acquire LDMI. LDMI is a privately held, facilities-based competitive local exchange carrier serving business and residential customers primarily in Michigan and Ohio. Under the terms of the Acquisition Agreement, at closing, the Talk America subsidiary will be merged into LDMI, LDMI will become an indirect wholly owned subsidiary of Talk America and, in exchange for all of the stock of LDMI, Talk America will pay \$24 million in cash and issue 1.8 million shares of its common stock. The shares of Talk America common stock will be issued only to the holders of LDMI's preferred stock and will not be registered under the Securities Act of 1933. Talk America has agreed to file a registration statement with the Securities and Exchange Commission as promptly as reasonably practicable after the closing to permit resales of common stock by such holders. The Acquisition Agreement and the LDMI acquisition transaction have been approved by both the board of directors and the stockholders of LDMI and the closing is subject to the receipt of certain regulatory approvals, which both companies expect to receive.

A copy of the Acquisition Agreement is filed as Exhibit 10.1 of this Report. The foregoing discussion of the Acquisition Agreement and the acquisition contemplated thereby is only a summary of the terms and conditions of the Acquisition Agreement and is qualified in its entirety by reference to the Acquisition Agreement included as Exhibit 10.1.

Item 7.01 Regulation FD Disclosure

On May 23, 2005, Talk America issued a release announcing that it had entered into an agreement to acquire LDMI Telecommunications, Inc., a privately held, facilities-based competitive local exchange carrier serving business and residential customers primarily in Michigan and Ohio, and also that it was amending its financial and operating guidance for the full year 2005. A copy of the release is furnished as Exhibit 99.1 of this Report.

Item 9.01 Financial Statements and Exhibits

(c) Exhibits

۰.

10.1 Agreement and Plan of Merger, dated as of May 23, 2005, among LDMI Telecommunications, Inc., Talk America Holdings, Inc. and Lion Acquisition Corp.

99.1 Release dated May 23, 2005.

SIGNATURES

Pursuant to the requirements of the Securities Exchange Act of 1934, the Registrant has duly caused this report to be signed on its behalf by the undersigned hereunto duly authorized.

TALK AMERICA HOLDINGS, INC.

Date: May 23, 2005

By: <u>/s/ Aloysius T. Lawn IV</u> Name: Aloysius T. Lawn IV Title: Executive Vice President – General Counsel and Secretary

EXHIBIT INDEX

Exhibit Number Description

10.1 Agreement and Plan of Merger dated as of May 23, 2005, among LDMI Telecommunications, Inc., Talk America
 Holdings, Inc. and Lion Acquisition Corp.
 99.1 Release Dated May 23, 2005.

Exhibit 10.1

AGREEMENT AND PLAN OF MERGER

• . • .

dated as of

May 23, 2005

Among

LDMI TELECOMMUNICATIONS, INC.

TALK AMERICA HOLDINGS, INC.

AND

LION ACQUISITION CORP.

Table of Contents

ARTICLE I DEFINITIONS

1.1. Definitions.

· . · .

ARTICLE II THE TRANSACTION

2.1. The Merger.

- 2.2. The Closing.
- 2.3. Actions at the Closing.
- 2.4. Effect of Merger.
- 2.5. Procedure for Exchange.

2.6. Escrow.

2.7. Closing of Transfer Record.

ARTICLE III REPRESENTATIONS AND WARRANTIES OF COMPANY

3.1. Organization, Qualification and Corporate Power; Transaction Authorization.

- 3.2. Capitalization.
- 3.3. Noncontravention; Consents.
- 3.4. Compliance with Laws, Licenses.
- 3.5. Customers.
- 3.6. No General Solicitation.
- 3.7. Brokers' Fees.
- 3.8. Title to Assets.
- 3.9. Subsidiaries.
- 3.10. Financial Statements.
- 3.11. Events Subsequent to Most Recent Fiscal Year End.
- 3.12. Undisclosed Liabilities.
- 3.13. Antitakeover Statutes.
- 3.14. Tax Matters.
- 3.15. Real Property.
- 3.16. Intellectual Property.
- 3.17. Tangible Assets.
- 3.18. Contracts.
- 3.19. Notes and Accounts Receivable.
- 3.20. Powers of Attorney.
- 3.21. Insurance.
- 3.22. Litigation.
- 3.23. Employees.
- 3.24. Employee Benefits.
- 3.25. Guaranties.
- 3.26. Environmental, Health and Safety Matters.
- 3.27. Certain Business Relationships with Company and its Subsidiaries.
- 3.28. Accounts; Lockboxes; Safe Deposit Boxes.
- 3.29. Securities.
- 3.30. Accounting Matters.
- 3.31. Disclosure.

ARTICLE IV REPRESENTATIONS AND WARRANTIES OF PARENT AND MERGER SUB

4.1. Organization.

· . · .

- 4.2. Capitalization.
- 4.3. Authorization of Transaction.
- 4.4. Noncontravention.
- 4.5. Brokers' Fees.
- 4.6. Company Reports.
- 4.7. Authorization for Parent Shares; Capital Resources.
- 4.8. NASDAQ Compliance.
- 4.9. No Material Adverse Changes.
- 4.10. Undisclosed Liabilities.
- 4.11. Compliance with Laws.
- 4.12. Litigation.
- 4.13. No General Solicitation.

ARTICLE V **COVENANTS**

- 5.1. Interim Operations.
- 5.2. Listing.
- 5.3. Filings; Other Actions; Notification.5.4. Company Financial Statements.
- 5.5. Access.
- 5.6. Director and Officer Liability. 5.7. Employee Benefits After the Merger.
- 5.8. Lockup Agreement; Transfer Restrictions.
- 5.9. Notices and Filing by Company.

CONDITIONS TO OBLIGATIONS TO CLOSE ARTICLE VI

- 6.1. Conditions to Each Party's Obligation.
- 6.2. Conditions to Obligation of Parent and Merger Sub.
- 6.3. Conditions to Obligation of Company.

ARTICLE VII SURVIVAL OF REPRESENTATIONS AND WARRANTIES; INDEMNIFICATION

- 7.1. Survival of Representations and Warranties.
- 7.2. Indemnification.

ARTICLE VIII TERMINATION

- 8.1. Termination of Agreement.
- 8.2. Effect of Termination

ii

ARTICLE IX **REGISTRATION RIGHTS**

9.1. Definitions.

۰.

۰.

- 9.2. Shelf Registration; Suspension of Use of Prospectus.
 9.3. Registration Procedures.
 9.4. Registration Expenses.

- 9.5. Indemnification and Contribution.
- 9.6. Successors and Assigns; Survival.

ARTICLE X **MISCELLANEOUS**

10.1. Press Releases and Public Announcements.

- 10.2. No Third-Party Beneficiaries.
- 10.3. Entire Agreement.
- 10.4. Binding Effect; Assignment. 10.5. Counterparts.
- 10.6. Headings.
- 10.7. Notices.
- 10.8. Governing Law.
- 10.9. Amendments and Waivers.
- 10.10. Severability.
- 10.11. Expenses.
- 10.12. Survival.
- 10.13. Construction.
- 10.14. Incorporation of Exhibits and Schedules. 10.15. Specific Performance.
- 10.16. Submission to Jurisdiction.
- 10.17. Waiver of Jury Trial.

Exhibits:

A. Form of Escrow Agreement

iii

AGREEMENT AND PLAN OF MERGER

THIS AGREEMENT AND PLAN OF MERGER (the "Agreement") is dated effective May 23, 2005, by and among TALK AMERICA HOLDINGS, INC., a Delaware corporation ("Parent"), LION ACQUISITION CORP., a Michigan corporation and indirectly wholly owned subsidiary of Parent ("Merger Sub"), and LDMI TELECOMMUNICATIONS, INC., a Michigan corporation ("Company"). Parent, Merger Sub and Company are referred to collectively herein as the "Parties," and Company and Merger Sub are sometimes collectively referred to as the "Constituent Corporations."

WITNESSETH:

WHEREAS, this Agreement contemplates a transaction whereby Parent will acquire all of the outstanding capital stock of Company through a merger of Merger Sub with and into Company;

WHEREAS, the Board of Directors of each of Parent, Merger Sub and Company has approved the acquisition of Company by Parent, including the merger of Merger Sub with and into Company (the "Merger"), upon the terms and subject to the conditions set forth herein, and the Board of Directors of each of Company and Merger Sub has adopted this Agreement;

WHEREAS, the Board of Directors of Company has determined that the Merger is advisable and is fair to and in the best interests of the holders of the Company Shares and recommended the approval of the Merger and this Agreement by the stockholders of Company and this Agreement and the Merger have been approved by the requisite vote of the stockholders of Company; and

NOW, THEREFORE, in consideration of the premises and the mutual promises set forth herein, and in consideration of the representations, warranties and covenants set forth herein, the Parties agree as follows:

ARTICLE I DEFINITIONS

1.1. Definitions.

• • • •

(a) The following terms, as used herein, have the following meanings:

"A/D Aggregate Preferred Preference Amount" means the amount equal to the sum of the Series A Preference Amount, the Series B Preference Amount, the Series C Preference Amount and the Series D Preference Amount.

"*A/D Consideration*" means (x) the Preferred Consideration Amount, minus (y) the sum of (i) the Series E Preference Amount plus (ii) the A Participation.

"Affiliate" has the meaning set forth in Rule 12b-2 of the regulations promulgated under the Securities Exchange Act.

"*Affiliated* Group" means any affiliated group within the meaning of Code Section 1504(a) or any similar group defined under a similar provision of federal, state, local or foreign law.

"A Participation" means the product of \$0.20 times the number of shares of Common Stock into which a share of Series A Preferred is convertible as of the Effective Time.

"Cash Cap" means \$24,000,000, minus the sum of the amount by which Transaction Costs and Severance Amounts, as set forth in the Closing Costs Certificate, exceeds the Deduction Amount.

"*Cash Component*" means (a) the portion of the aggregate Merger Consideration that is paid in cash and (b) the cash payable for Employee Options and Warrants pursuant to Section 2.4(g).

"Claim Date" means the earlier of (a) the 60th day after Parent files its first Annual Report on Form 10-K with the SEC after the Closing Date and May 31, 2006.

"COBRA" means the requirements of Part 6 of Subtitle B of Title I of ERISA and Code Section 4980B and of any similar state law.

"Code" means the Internal Revenue Code of 1986, as amended.

"Combined Material Adverse Effect" means a Material Adverse Effect on Company or a Material Adverse Effect on Parent and its Subsidiaries (including the Surviving Corporation), taken as a whole, after the Effective Time.

"Company Board" means the board of directors of Company.

"Confidentiality Agreement" means the letter agreement dated February 22, 2005 between Parent and Company, as it may be amended, providing that, among other things, each Party would maintain confidential certain information of the other Party.

"Consenting Stockholders" means the Preferred Stockholders that approve this Agreement, the Merger and the Articles Amendment by written consent pursuant to Stockholder Consents.

"Consideration" of a Series means, as to such Series of Preferred Stock, the amount set forth below opposite such Series reference:

Series A – amount equal to the sum of (x) the product of (i) the result of (A) the A Preference Amount <u>divided</u> by (B) the A/D Aggregate Preference Amount, <u>multiplied</u> by (ii) the A/D Consideration, <u>plus</u> (y) the A Participation.

Series B – amount equal to the product of (x) the result of (i) the B Preference Amount <u>divided</u> by (ii) the A/D Aggregate Preferred Preference Amount, <u>multiplied</u> by (y) the A/D Consideration.

Series C – amount equal to the product of (x) the result of (i) the C Preference Amount <u>divided</u> by (ii) the A/D Aggregate Preference Amount, <u>multiplied</u> by (y) the A/D Consideration.

Series D – amount equal to the product of (x) the result of (i) the D Preference Amount <u>divided</u> by (ii) the A/D Aggregate Preferred Preference Amount, multiplied by (y) the A/D Consideration.

Series E – the Series E Preference Amount.

"Deduction Amount" means \$4,500,000.

•

"Deferred Intercompany Transaction" has the meaning set forth in Treas. Reg. Section 1.1502-13.

"Dissenting Stockholder" means a Preferred Stockholder that is not a Consenting Stockholder and that demands cash payment for its Preferred Stock under Section 450.1767 of the Michigan Law.

"Employee Benefit Plan" means any "employee benefit plan" (as such term is defined in ERISA Section 3(3)) and any other employee benefit plan, program or arrangement of any kind.

"Employee Options" means any stock options to purchase shares of Common Stock granted under any employee stock option or compensation plan or arrangement of Company.

"Employee Pension Benefit Plan" has the meaning set forth in ERISA Section 3(2).

"Employee Welfare Benefit Plan" has the meaning set forth in ERISA Section 3(1).

"Environmental, Health and Safety Requirements" means all federal, state, local and foreign statutes, regulations, ordinances and other provisions having the force or effect of law, all judicial and administrative orders and determinations, all contractual obligations and all common law concerning public health and safety, worker health and safety, and pollution or protection of the environment, including all those relating to the presence, use, production, generation, handling, transportation, treatment, storage, disposal, distribution, labeling, testing, processing, discharge, release, threatened release, control, or cleanup of any hazardous materials, substances or wastes, chemical substances or mixtures, pesticides, pollutants, contaminants, toxic chemicals, petroleum products or byproducts, asbestos, polychlorinated biphenyl, noise or radiation, each as amended and as now or hereafter in effect.

"ERISA" means the Employee Retirement Income Security Act of 1974, as amended.

"ERISA Affiliate" means each entity that is treated as a single employer with Company for purposes of Code Section 414.

"Excess Loss Account" has the meaning set forth in Treas. Reg. Section 1.1502-19.

"FCC" means the Federal Communications Commission.

"FCC Consent" means the grant by the FCC of its consent to the transfer of the FCC Licenses in connection with the consummation of the transactions contemplated hereby.

"Fiduciary" has the meaning set forth in ERISA Section 3(21).

÷ .

"GAAP" means United States generally accepted accounting principles as in effect from time to time.

"Governmental Entity" means any United States federal, state or local or any foreign government, governmental regulatory or administrative authority, agency, commission (including any department or political subdivision of any of the foregoing), court, tribunal or judicial or arbitral body.

"Governmental Order" means any order, ruling, writ, judgment, injunction, decree, charge, stipulation, determination or award entered by or with any Governmental Entity.

"Intellectual Property" means (a) all inventions (whether patentable or unpatentable and whether or not reduced to practice), all improvements thereto, and all patents, patent applications and patent disclosures, together with all reissuances, continuations, continuations-in-part, revisions, extensions and reexaminations thereof, (b) all trademarks, service marks, trade dress, logos, trade names and corporate names, together with all translations, adaptations, derivations and combinations thereof and including all goodwill associated therewith, and all applications, registrations and renewals in connection therewith, (c) all copyrightable works, all copyrights and all applications, registrations and renewals in connection therewith, (d) all mask works and all applications, registrations and renewals in connection therewith, (e) all trade secrets and confidential business information (including ideas, research and development, know-how, formulae, compositions, manufacturing and production processes and techniques, technical data, designs, drawings, specifications, customer and supplier lists, pricing and cost information, and business and marketing plans and proposals), (f) all Computer Software (including data and related documentation), (g) all other proprietary rights and (h) all copies and tangible embodiments thereof (in whatever form or medium).

"Knowledge" of any Person that is not an individual means the actual knowledge of (i) such Person's executive officers, in the case of Parent and Merger Subsidiary, and (ii) Patrick O'Leary, Michael Mahoney, Mark Wayne, Kim McDonald or Linda Hanson, in the case of the Company, in each case after reasonable investigation by such individual.

"Laws" mean any laws, statutes, rules, ordinances, regulations, codes, plans, injunctions, judgments, orders, writs, decrees, rulings and charges thereunder of any Governmental Entity.

"Liability" means any liability (whether known or unknown, whether asserted or unasserted, whether absolute or contingent, whether accrued or unaccrued, whether liquidated or unliquidated, and whether due or to become due), including any liability for Taxes.

"*Main Leases*" means collectively all leases by Company or any of its Subsidiaries of (i) Suite 100 in the building known as 300 Galleria Officentre, Southfield, Michigan 48034 and (ii) Suites #400, #500 and #1600 and Storage Spaces #7 and #8 in the building known as the American Center, located at 27777 Franklin Road, Southfield, Michigan 48034.

"*Material Adverse Effect*" means, with respect to or of or on any Person, (a) a material adverse change in, or materially adverse effect on, the business, assets, revenues, financial condition or results of operations of such Person and its Subsidiaries taken as a whole, excluding any such effect resulting from (i) changes in political or regulatory conditions generally, (ii) changes or conditions generally affecting the U.S. economy or financial markets or generally affecting the segments of the telecommunications industry in which such Person or any of its Subsidiaries operates, (iii) changes in GAAP, or (iv) the announcement or consummation of this Agreement, or (b) an effect that would prevent, materially delay or materially impair the ability of such Person to consummate the Merger and the other transactions contemplated by this Agreement.

"Michigan Law" means the State of Michigan Business Corporations Act, 1972, PA 284, sections 450.1101 to 450.2098 of the Michigan Compiled Laws, as amended.

"Most Recent Balance Sheet" means the balance sheet contained within the Most Recent Financial Statements.

"Multiemployer Plan" has the meaning set forth in ERISA Section 3(37).

"NASD" means the National Association of Securities Dealers, Inc.

"Nasdaq" means the Nasdaq National Market.

"Ordinary Course of Business" means the ordinary course of business consistent with past practice.

"Parent Board" means the board of directors of Parent.

"Parent Share Valuation" means \$8.42 per share of Parent Stock.

"Parent Shares" means any shares of Parent Stock.

"Parent Stock" means the common stock, par value \$.01 per share, of Parent.

"PBGC" means the Pension Benefit Guaranty Corporation.

"Per Share Preference Amount" of a Series means, as to such Series of Preferred Stock, the sum of the respective amounts set forth below opposite such Series reference below, plus the amount equal to the accrued and unpaid dividends on a share of such Series as of the Effective Time:

Series A - \$1.87 Series B - \$2.04 Series C - \$4.24 Series D - \$2.15 Series E - \$1.55

"Person" means an individual, a partnership, a corporation, a limited liability company, an association, a joint stock company, a trust, a joint venture, an unincorporated organization or a Governmental Entity.

"Preference Amount" of a Series means, as to such Series of Preferred Stock, the amount that equals the product of such Series' Per Share Preference Amount times the number of shares of such Series issued and outstanding as of the Effective Time.

"Preferred Consideration Amount" means the amount equal to the sum of the Remaining Cash, plus the Stock Value.

"Preferred Stockholder" means a holder of record of shares of Preferred Stock.

"Prohibited Transaction" has the meaning set forth in ERISA Section 406 and Code Section 4975.

"Ratio" of a Series means, as to such Series, the fraction, the numerator of which is the Consideration Amount of such Series and the denominator of which is the Preferred Consideration Amount.

"Remaining Cash" means an amount of cash equal to the Cash Cap, minus the sum of (i) the total amount of cash payable as Merger Consideration pursuant to Section 2.4(f)(1) and (ii) the cash payable in respect of the Employee Options and the Warrants pursuant to Section 2.4(g).

"*Remaining Parent Shares*" means the number of Parent Shares equal to the Stock Cap minus the total number of Parent Shares payable as Merger Consideration (without giving effect to the withdrawal of the Escrowed Consideration pursuant to Section 2.5) pursuant to Section 2.4(f)(2)(E).

"Reportable Event" has the meaning set forth in ERISA Section 4043.

"Representatives" has the meaning set forth in the Escrow Agreement.

"Requisite Stockholder Approval" means approval by the affirmative vote, at a meeting or by written consent, of (i) the holders of such number of shares of Common Stock or of Preferred Stock convertible into such number of shares of Common Stock, voting as a single class, as would in the aggregate equal at least a majority of the sum of the outstanding shares of Common Stock and the shares of Common Stock issuable upon conversion of the outstanding shares of Preferred Stock, (ii) the holders of a majority of each of the outstanding Series A Preferred, Series B Preferred, Series C Preferred and Series D Preferred, (iii) the holders of at least 60% of the outstanding Series E Preferred, voting on an as-converted basis, and (iv) the holders of at least 60% of the outstanding Series E Preferred, voting on an as-converted basis.

"Resolution Date" means the date as of which any potential dissenting Preferred Stockholder either demands payment or loses its right to demand payment under Section 450.1767 of the Michigan Law as to its shares of Preferred Stock by reason of the Articles Amendment.

"SEC" means the Securities and Exchange Commission.

"Securities Act" means the Securities Act of 1933, as amended, and the rules and regulations promulgated thereunder.

"Securities Exchange Act" means the Securities Exchange Act of 1934, as amended, and the rules and regulations promulgated thereunder.

"Security Interest" means any mortgage, pledge, lien, encumbrance, charge or other security interest, other than (a) mechanic's, materialman's and similar liens; (b) liens for taxes not yet due and payable; (c) purchase money liens and liens securing rental payments under capital lease arrangements; and (d) other liens arising in the Ordinary Course of Business and not incurred in connection with the borrowing of money.

"Severance Amounts" means all out-of-pocket compensation, bonuses, fees and expenses payable by Company or any of its Subsidiaries to any of its employees or directors in connection with or by reason of the transactions contemplated hereby, and any other amount that is stated elsewhere in this Agreement to be, or be deemed to be, a Severance Amount.

"State PUC Consent" means the grant by any state Governmental Entity that granted or issued any of the State Licenses or otherwise has authority in the matter of its consent to the transfer of the State Licenses in connection with the consummation of the transactions contemplated hereby.

"Stock Cap" means 1,800,000 Parent Shares, as such number may be adjusted as the Stock Component as provided in Section 2.4(e).

"Stock Component" means that portion of the Merger Consideration that is paid in Parent Shares.

"Stockholder Consent" means a Consent, executed by a Preferred Stockholder, approving this Agreement, the Merger and the Articles Amendment.

"Stock Value" means the amount equal to the product of the Stock Cap multiplied by the Parent Share Valuation.

"Subsidiary" of a specified Person means any corporation, limited liability company, partnership, joint venture or other legal entity of which the specified Person (either alone or together with any other Subsidiary of the specified Person) owns, directly or indirectly, more than 50% of the stock or other equity, partnership, limited liability company or equivalent interests, the holders of which are generally entitled to vote for the election of the board of directors or other governing body of such corporation or other legal entity, or otherwise has the power to vote or direct the voting of sufficient securities to elect a majority of such board of directors or other governing body.

"Switch Lease" means the Sublease, dated as of the date of this Agreement, between Parent and Company, providing for the sublease by Company to a Subsidiary of Parent of certain space leased by Company at Suite 100 in the building known as 300 Galleria Officentre, Southfield, Michigan 48034.

"Tax" means any federal, state, local or foreign income, gross receipts, license, payroll, employment, excise, severance, stamp, occupation, premium, windfall profits, environmental (including taxes under Code Section 59A), customs duties, capital stock, franchise, profits, withholding, social security (or similar), unemployment, disability, real property, personal property, sales, use, transfer, registration, value added, alternative or add-on minimum, estimated or other tax of any kind whatsoever, including any interest, penalty, or addition thereto, whether disputed or not.

"Tax Return" means any report, return, declaration or other information required to be supplied to a taxing authority in connection with Taxes.

"Total Merger Consideration" means cash in the aggregate amount of the Cash Cap and Parent Shares in the aggregate number of the Stock Cap.

"Transaction Costs" means all fees and expenses of legal counsel, investment bankers, brokers or other representatives and consultants, including Brown Brothers Harriman, Dykema Gossett PLLC (for services subsequent to February 1, 2005), Chadbourne & Parke LLP and Altman Vilandrie, incurred or accrued by Company, or for which Company or any of its Subsidiaries is liable, in connection with the negotiation, execution and delivery of this Agreement, the performance of the obligations of Company and its Subsidiaries hereunder and the consummation of the transactions contemplated hereby, and any other amount that is stated elsewhere in this Agreement to be, or be deemed to be, a Transaction Cost.

(b) Each of the following terms is defined in the Section set forth opposite such term:

AgreementPreambleAllocated Portion7.2(e)(2)Articles Amendment3.1(b)Basket Amount7.2(f)Certificate of Amendment2.3Closing2.2Closing Costs Certificate6.2(i)Closing Date2.2Common Stock3.2CompanyPreambleCompany Disclosure LetterARTICLEIII100Company Indemnified Party7.2(a)(2)Company Shares3.2Computer Software3.16Compensation Plan5.1(a)(18)Constituent CorporationsPreambleDissenter Cash2.4(f)(4)Dissenter Shares2.4(f)(4)Escrowed Consideration3.1(c)(iii)Escrow Agent2.6(a)Escrow Agent2.6(a)Escrow Agent2.5(a)Exchange Agent2.5(a)FCC Licenses3.4Financial Statements3.10Holder9.1Indemnified Party7.2(b)(1)IT Assets3.16Licenses3.4Statements3.10Holder9.1Indemnifying Party7.2(b)(1)IT Assets3.16Licenses3.4Local Licenses3.4Majority Holder9.1Material Contract3.18Merger Consideration2.4(c)Merger Shares5.8(a)	TERM	SECTION
Allocated Portion7.2(e)(2)Articles Amendment3.1(b)Basket Amount7.2(f)Certificate of Amendment2.3Closing2.2Closing Costs Certificate6.2(i)Closing Date2.2Common Stock3.2Communications Licenses3.4CompanyPreambleCompany Disclosure LetterARTICLEIIIIIICompany Indemnified Party7.2(a)(2)Company Shares3.2Computer Software3.16Compensation Plan5.1(a)(18)Constituent CorporationsPreambleDissenter Cash2.4(f)(4)Effective Time2.4(a)Escrowed Consideration3.1(c)(iii)Escrow Agent2.6(a)Escrow Agent2.6(a)Escrow Agent2.5(a)Financial Statements3.10Holder9.1Indemnified Party7.2(b)(1)IT Assets3.16Licenses3.4Local Licenses3.4Majority Holder9.1Material Contract3.18MergerRecitalsMarger Consideration2.4(e)	Agreement	Preamble
Articles Amendment $3.1(b)$ Basket Amount $7.2(f)$ Certificate of Amendment 2.3 Closing 2.2 Closing Costs Certificate $6.2(i)$ Closing Date 2.2 Common Stock 3.2 Communications Licenses 3.4 CompanyPreambleCompany Disclosure LetterARTICLEIIICompany Indemnified Party $7.2(a)(2)$ Company Shares 3.2 Computer Software 3.16 Compensation Plan $5.1(a)(18)$ Consituent CorporationsPreambleDissenter Cash $2.4(f)(4)$ Effective Time $2.4(a)$ Escrowed Consideration $2.5(a)$ Directors' Recommendation $3.1(c)(iii)$ Escrow Agent $2.6(a)$ Escrow Agent $2.6(a)$ Escrow Fund $7.2(e)(1)$ Exchange Fund $2.5(a)$ FCC Licenses 3.4 Financial Statements 3.10 Holder 9.1 Indemnified Person $5.6(a)$ Indemnifying Party $7.2(b)(1)$ IT Assets 3.16 Licenses 3.4 Local Licenses 3.4 Majority Holder 9.1 Material Contract 3.18 MergerRecitalsMerger Consideration $2.4(e)$	-	
Basket Amount7.2(f)Certificate of Amendment2.3Closing2.2Closing Costs Certificate6.2(i)Closing Date2.2Common Stock3.2Communications Licenses3.4CompanyPreambleCompany Disclosure LetterARTICLEIII10Company Shares3.2Computer Software3.16Compensation Plan5.1(a)(18)Constituent CorporationsPreambleDissenter Cash2.4(f)(4)Dissenter Shares2.4(f)(4)Effective Time2.4(a)Escrowed Consideration3.1(c)(iii)Escrow Agent2.6(a)Escrow Agent2.6(a)Escrow Agent2.5(a)Exchange Agent2.5(a)FC Licenses3.4Financial Statements3.10Holder9.1Indemnified Party7.2(b)(1)IT Assets3.16Licenses3.4Licenses3.4Local Licenses3.4Majority Holder9.1Material Contract3.18MergerRecitalsMerger Consideration2.4(c)Merger Consideration2.4(c)		
Certificate of Amendment2.3Certificate of Merger2.3Closing2.2Closing Costs Certificate6.2(i)Closing Date2.2Common Stock3.2Communications Licenses3.4CompanyPreambleCompany Disclosure LetterARTICLEIIICompany SharesComputer Software3.16Compensation Plan5.1(a)(18)Constituent CorporationsPreambleDissenter Cash2.4(f)(4)Effective Time2.6(a)Escrowed Consideration2.5(a)Directors' Recommendation3.1(c)(iii)Escrow Agent2.6(a)Escrow Agent2.6(a)Escrow Agent2.5(a)Exchange Agent2.5(a)Financial Statements3.10Holder9.1Indemnified Party7.2(b)(1)IT Assets3.16Licenses3.4Licenses3.4Local Licenses3.4Majority Holder9.1Material Contract3.18Merger Consideration2.4(e)Merger Shares5.8(a)		7.2(f)
Certificate of Merger2.3Closing2.2Closing Costs Certificate6.2(i)Closing Date2.2Common Stock3.2Communications Licenses3.4CompanyPreambleCompany Disclosure LetterARTICLEIIIIIICompany Indemnified Party7.2(a)(2)Company Shares3.2Computer Software3.16Compensation Plan5.1(a)(18)Constituent CorporationsPreambleDissenter Cash2.4(f)(4)Effective Time2.4(a)Escrowed Consideration2.5(a)Directors' Recommendation3.1(c)(iii)Escrow Agent2.6(a)Escrow Agent2.6(a)Escrow Agent2.5(a)Exchange Agent2.5(a)FCC Licenses3.4Financial Statements3.10Holder9.1Indemnified Person5.6(a)Indemnified Person5.6(a)Indemnified Person5.6(a)Indemnified Person5.6(a)Indemnified Person5.6(a)Indemnified Person5.6(a)Indemnified Person5.6(a)Indemnified Person5.6(a)Indemnifying Party7.2(b)(1)IT Assets3.16Licenses3.4Local Licenses3.4Material Contract3.18MergerRecitalsMerger Consideration2.4(e)Merger Shares5.8(a)		
Closing2.2Closing Costs Certificate6.2(i)Closing Date2.2Common Stock3.2Communications Licenses3.4CompanyPreambleCompany Disclosure LetterARTICLEIIICompany Indemnified Party7.2(a)(2)Company Shares3.2Computer Software3.16Compensation Plan5.1(a)(18)Constituent CorporationsPreambleDissenter Cash2.4(f)(4)Dissenter Shares2.4(f)(4)Escrowed Consideration2.5(a)Directors' Recommendation3.1(c)(iii)Escrow Agent2.6(a)Escrow Agent2.6(a)Escrow Agent2.5(a)Exchange Agent2.5(a)FCC Licenses3.4Financial Statements3.10Holder9.1Indemnified Party7.2(b)(1)IT Assets3.16Licenses3.4Local Licenses3.4Majority Holder9.1Material Contract3.18MergerRecitalsMerger Shares5.8(a)		
Closing Costs Certificate6.2(i)Closing Date2.2Common Stock3.2Communications Licenses3.4CompanyPreambleCompany Disclosure LetterARTICLEIIIIIICompany Indemnified Party7.2(a)(2)Company Shares3.2Computer Software3.16Compensation Plan5.1(a)(18)Constituent CorporationsPreambleDissenter Cash2.4(f)(4)Effective Time2.4(a)Escrowed Consideration2.5(a)Directors' Recommendation3.1(c)(iii)Escrow Agent2.6(a)Escrow Agent2.6(a)Escrow Fund7.2(e)(1)Exchange Agent2.5(a)FCC Licenses3.4Financial Statements3.10Holder9.1Indemnified Party7.2(b)(1)IT Assets3.16Licenses3.4Local Licenses3.4Majority Holder9.1Material Contract3.18MergerRecitalsMerger Consideration2.4(e)Merger Shares5.8(a)	0	
Closing Date2.2Common Stock3.2Communications Licenses3.4CompanyPreambleCompany Disclosure LetterARTICLEIIIIIICompany Indemnified Party7.2(a)(2)Company Shares3.2Computer Software3.16Compensation Plan5.1(a)(18)Constituent CorporationsPreambleDissenter Cash2.4(f)(4)Effective Time2.4(a)Escrowed Consideration2.5(a)Directors' Recommendation3.1(c)(iii)Escrow Agent2.6(a)Escrow Agent2.6(a)Escrow Agent2.5(a)Exchange Agent2.5(a)Exchange Fund2.5(a)FCC Licenses3.4Financial Statements3.10Holder9.1Indemnified Party7.2(b)(1)IT Assets3.16Licenses3.4Local Licenses3.4Majority Holder9.1Material Contract3.18MergerRecitalsMerger Shares5.8(a)		
Common Stock3.2Communications Licenses3.4CompanyPreambleCompany Disclosure LetterARTICLEIIIIIICompany Indemnified Party7.2(a)(2)Company Shares3.2Computer Software3.16Compensation Plan5.1(a)(18)Constituent CorporationsPreambleDissenter Cash2.4(f)(4)Effective Time2.4(a)Escrowed Consideration2.5(a)Directors' Recommendation3.1(c)(iii)Escrow Agent2.6(a)Escrow Agent2.6(a)Escrow Agent2.5(a)Exchange Agent2.5(a)Exchange Fund2.5(a)FCC Licenses3.4Financial Statements3.10Holder9.1Indemnified Party7.2(b)(1)IT Assets3.16Licenses3.4Local Licenses3.4Majority Holder9.1Material Contract3.18MergerRecitalsMerger Consideration2.4(e)		
Communications Licenses3.4CompanyPreambleCompany Disclosure LetterARTICLEIIIIIICompany Indemnified Party7.2(a)(2)Company Shares3.2Computer Software3.16Compensation Plan5.1(a)(18)Constituent CorporationsPreambleDissenter Cash2.4(f)(4)Dissenter Shares2.4(f)(4)Effective Time2.4(a)Escrowed Consideration2.5(a)Directors' Recommendation3.1(c)(iii)Escrow Agent2.6(a)Escrow Agent2.6(a)Escrow Agent2.5(a)Exchange Agent2.5(a)Exchange Fund2.5(a)FCC Licenses3.4Financial Statements3.10Holder9.1Indemnified Party7.2(b)(1)IT Assets3.16Licenses3.4Local Licenses3.4Majority Holder9.1Material Contract3.18MergerRecitalsMerger Consideration2.4(e)Merger Shares5.8(a)		
CompanyPreambleCompany Disclosure LetterARTICLEIIIIIICompany Indemnified Party7.2(a)(2)Company Shares3.2Computer Software3.16Compensation Plan5.1(a)(18)Constituent CorporationsPreambleDissenter Cash2.4(f)(4)Effective Time2.4(a)Escrowed Consideration2.5(a)Directors' Recommendation3.1(c)(iii)Escrow Agent2.6(a)Escrow Agent2.6(a)Escrow Agent2.5(a)Exchange Agent2.5(a)Exchange Fund2.5(a)FCC Licenses3.4Financial Statements3.10Holder9.1Indemnified Party7.2(b)(1)IT Assets3.16Licenses3.4Local Licenses3.4Majority Holder9.1Material Contract3.18MergerRecitalsMerger Consideration2.4(e)Merger Shares5.8(a)		• ·
Company Disclosure LetterARTICLEIIIIIICompany Indemnified Party7.2(a)(2)Company Shares3.2Computer Software3.16Compensation Plan5.1(a)(18)Constituent CorporationsPreambleDissenter Cash2.4(f)(4)Dissenter Shares2.4(f)(4)Effective Time2.4(a)Escrowed Consideration2.5(a)Directors' Recommendation3.1(c)(iii)Escrow Agent2.6(a)Escrow Agent2.6(a)Escrow Agent2.5(a)Escrow Fund7.2(e)(1)Exchange Agent2.5(a)FCC Licenses3.4Financial Statements3.10Holder9.1Indemnified Party7.2(b)(1)IT Assets3.16Licenses3.4Local Licenses3.4Majority Holder9.1Material Contract3.18MergerRecitalsMerger Consideration2.4(e)Merger Shares5.8(a)		
IIICompany Indemnified Party7.2(a)(2)Company Shares3.2Computer Software3.16Compensation Plan5.1(a)(18)Constituent CorporationsPreambleDissenter Cash2.4(f)(4)Dissenter Shares2.4(f)(4)Effective Time2.4(a)Escrowed Consideration2.5(a)Directors' Recommendation3.1(c)(iii)Escrow Agent2.6(a)Escrow Agent2.6(a)Escrow Agent2.6(a)Escrow Agent2.5(a)Exchange Agent2.5(a)Exchange Agent2.5(a)Exchange Fund2.5(a)FCC Licenses3.4Financial Statements3.10Holder9.1Indemnified Party7.2(b)(1)IT Assets3.16Licenses3.4Local Licenses3.4Majority Holder9.1Material Contract3.18MergerRecitalsMerger Consideration2.4(c)Merger Shares5.8(a)		
Company Indemnified Party7.2(a)(2)Company Shares3.2Computer Software3.16Compensation Plan5.1(a)(18)Constituent CorporationsPreambleDissenter Cash2.4(f)(4)Dissenter Shares2.4(f)(4)Effective Time2.4(a)Escrowed Consideration2.5(a)Directors' Recommendation3.1(c)(iii)Escrow Agent2.6(a)Escrow Agent2.6(a)Escrow Agent2.5(a)Escrow Fund7.2(e)(1)Exchange Agent2.5(a)Exchange Fund2.5(a)FCC Licenses3.4Financial Statements3.10Holder9.1Indemnified Party7.2(b)(1)IT Assets3.16Licenses3.4Local Licenses3.4Majority Holder9.1Material Contract3.18MergerRecitalsMerger Consideration2.4(e)Merger Shares5.8(a)	Company Disclosure Letter	
Company Shares3.2Computer Software3.16Compensation Plan5.1(a)(18)Constituent CorporationsPreambleDissenter Cash2.4(f)(4)Dissenter Shares2.4(f)(4)Effective Time2.4(a)Escrowed Consideration2.5(a)Directors' Recommendation3.1(c)(iii)Escrow Agent2.6(a)Escrow Agreement2.6(a)Escrow Agreement2.6(a)Escrow Fund7.2(e)(1)Exchange Agent2.5(a)Exchange Fund2.5(a)FCC Licenses3.4Financial Statements3.10Holder9.1Indemnified Party7.2(b)(1)IT Assets3.16Licenses3.4Local Licenses3.4Majority Holder9.1Material Contract3.18MergerRecitalsMerger Consideration2.4(e)Merger Shares5.8(a)	Company Indemnified Party	
Computer Software3.16Compensation Plan5.1(a)(18)Constituent CorporationsPreambleDissenter Cash2.4(f)(4)Dissenter Shares2.4(f)(4)Effective Time2.4(a)Escrowed Consideration2.5(a)Directors' Recommendation3.1(c)(iii)Escrow Agent2.6(a)Escrow Agreement2.6(a)Escrow Agreement2.6(a)Escrow Agent2.5(a)Exchange Agent2.5(a)Exchange Fund7.2(e)(1)Exchange Fund2.5(a)FCC Licenses3.4Financial Statements3.10Holder9.1Indemnified Party7.2(b)(1)IT Assets3.16Licenses3.4Local Licenses3.4Majority Holder9.1Material Contract3.18MergerRecitalsMerger Consideration2.4(e)Merger Shares5.8(a)		
Compensation Plan5.1(a)(18)Constituent CorporationsPreambleDissenter Cash2.4(f)(4)Dissenter Shares2.4(f)(4)Effective Time2.4(a)Escrowed Consideration2.5(a)Directors' Recommendation3.1(c)(iii)Escrow Agent2.6(a)Escrow Agenement2.6(a)Escrow Agreement2.6(a)Escrow Amount7.2(e)(1)Exchange Agent2.5(a)Exchange Fund2.5(a)FCC Licenses3.4Financial Statements3.10Holder9.1Indemnified Party7.2(b)(1)IT Assets3.16Licenses3.4Local Licenses3.4Majority Holder9.1Material Contract3.18MergerRecitalsMerger Consideration2.4(e)Merger Shares5.8(a)		
Constituent CorporationsPreambleDissenter Cash2.4(f)(4)Dissenter Shares2.4(f)(4)Effective Time2.4(a)Escrowed Consideration2.5(a)Directors' Recommendation3.1(c)(iii)Escrow Agent2.6(a)Escrow Agent2.6(a)Escrow Agreement2.6(a)Escrow Agreement2.6(a)Escrow Agreement2.6(a)Escrow Fund7.2(e)(1)Exchange Agent2.5(a)Exchange Fund2.5(a)FCC Licenses3.4Financial Statements3.10Holder9.1Indemnified Party7.2(b)(1)IT Assets3.16Licenses3.4Local Licenses3.4Majority Holder9.1Material Contract3.18MergerRecitalsMerger Consideration2.4(e)Merger Shares5.8(a)		
Dissenter Cash2.4(f)(4)Dissenter Shares2.4(f)(4)Effective Time2.4(a)Escrowed Consideration2.5(a)Directors' Recommendation3.1(c)(iii)Escrow Agent2.6(a)Escrow Agreement2.6(a)Escrow Agreement2.6(a)Escrow Agreement2.6(a)Escrow Fund7.2(e)(1)Exchange Agent2.5(a)Exchange Fund2.5(a)FCC Licenses3.4Financial Statements3.10Holder9.1Indemnified Party7.2(b)Indemnified Person5.6(a)Indemnifying Party7.2(b)(1)IT Assets3.16Licenses3.4Local Licenses3.4Majority Holder9.1Material Contract3.18MergerRecitalsMerger Consideration2.4(e)Merger Shares5.8(a)		
Dissenter Shares $2.4(f)(4)$ Effective Time $2.4(a)$ Escrowed Consideration $2.5(a)$ Directors' Recommendation $3.1(c)(iii)$ Escrow Agent $2.6(a)$ Escrow Agreement $2.6(a)$ Escrow Agreement $2.6(a)$ Escrow Fund $7.2(e)(1)$ Exchange Agent $2.5(a)$ Exchange Fund $2.5(a)$ FCC Licenses 3.4 Financial Statements 3.10 Holder 9.1 Indemnified Party $7.2(b)(1)$ IT Assets 3.16 Licenses 3.4 Local Licenses 3.4 Majority Holder 9.1 Material Contract 3.18 MergerRecitalsMerger Consideration $2.4(e)$ Merger Shares $5.8(a)$		
Effective Time2.4(a)Escrowed Consideration2.5(a)Directors' Recommendation3.1(c)(iii)Escrow Agent2.6(a)Escrow Agreement2.6(a)Escrow Agreement7.2(e)(1)Escrow Fund7.2(e)(1)Exchange Agent2.5(a)Exchange Fund2.5(a)FCC Licenses3.4Financial Statements3.10Holder9.1Indemnified Party7.2(b)Indemnifying Party7.2(b)(1)IT Assets3.16Licenses3.4Local Licenses3.4Majority Holder9.1Material Contract3.18MergerRecitalsMerger Consideration2.4(e)Merger Shares5.8(a)		
Escrowed Consideration2.5(a)Directors' Recommendation3.1(c)(iii)Escrow Agent2.6(a)Escrow Agreement2.6(a)Escrow Amount7.2(e)(1)Escrow Fund7.2(e)(1)Exchange Agent2.5(a)Exchange Fund2.5(a)FCC Licenses3.4Financial Statements3.10Holder9.1Indemnified Party7.2(b)Indemnified Person5.6(a)Indemnifying Party7.2(b)(1)IT Assets3.16Licenses3.4Local Licenses3.4Majority Holder9.1Material Contract3.18MergerRecitalsMerger Consideration2.4(e)Merger Shares5.8(a)		2.4(1)(4)
Directors' Recommendation3.1(c)(iii)Escrow Agent2.6(a)Escrow Agreement2.6(a)Escrow Amount7.2(e)(1)Escrow Fund7.2(e)(1)Exchange Agent2.5(a)Exchange Fund2.5(a)FCC Licenses3.4Financial Statements3.10Holder9.1Indemnified Party7.2(b)Indemnified Person5.6(a)Indemnifying Party7.2(b)(1)IT Assets3.16Licenses3.4Local Licenses3.4Majority Holder9.1Material Contract3.18MergerRecitalsMerger Consideration2.4(e)Merger Shares5.8(a)		2.4(a)
Escrow Agent2.6(a)Escrow Agreement2.6(a)Escrow Amount7.2(e)(1)Escrow Fund7.2(e)(1)Exchange Agent2.5(a)Exchange Fund2.5(a)FCC Licenses3.4Financial Statements3.10Holder9.1Indemnified Party7.2(b)Indemnified Person5.6(a)Indemnifying Party7.2(b)(1)IT Assets3.16Licenses3.4Local Licenses3.4Majority Holder9.1Material Contract3.18MergerRecitalsMerger Consideration2.4(e)Merger Shares5.8(a)		
Escrow Agreement2.6(a)Escrow Amount7.2(e)(1)Escrow Fund7.2(e)(1)Exchange Agent2.5(a)Exchange Fund2.5(a)FCC Licenses3.4Financial Statements3.10Holder9.1Indemnified Party7.2(b)Indemnified Person5.6(a)Indemnifying Party7.2(b)(1)IT Assets3.16Licenses3.4Local Licenses3.4Majority Holder9.1Material Contract3.18MergerRecitalsMerger Consideration2.4(e)Merger Shares5.8(a)		
Escrow Amount7.2(e)(1)Escrow Fund7.2(e)(1)Escrow Fund7.2(e)(1)Exchange Agent2.5(a)Exchange Fund2.5(a)FCC Licenses3.4Financial Statements3.10Holder9.1Indemnified Party7.2(b)Indemnified Person5.6(a)Indemnifying Party7.2(b)(1)IT Assets3.16Licenses3.4Local Licenses3.4Majority Holder9.1Material Contract3.18MergerRecitalsMerger Consideration2.4(e)Merger Shares5.8(a)	9	
Escrow Fund7.2(e)(1)Exchange Agent2.5(a)Exchange Fund2.5(a)FCC Licenses3.4Financial Statements3.10Holder9.1Indemnified Party7.2(b)Indemnified Person5.6(a)Indemnifying Party7.2(b)(1)IT Assets3.16Licenses3.4Local Licenses3.4Majority Holder9.1Material Contract3.18MergerRecitalsMerger Consideration2.4(e)Merger Shares5.8(a)	5	
Exchange Agent2.5(a)Exchange Fund2.5(a)FCC Licenses3.4Financial Statements3.10Holder9.1Indemnified Party7.2(b)Indemnified Person5.6(a)Indemnifying Party7.2(b)(1)IT Assets3.16Licenses3.4Local Licenses3.4Majority Holder9.1Material Contract3.18MergerRecitalsMerger Consideration2.4(e)Merger Shares5.8(a)		7.2(e)(1)
Exchange Fund2.5(a)FCC Licenses3.4Financial Statements3.10Holder9.1Indemnified Party7.2(b)Indemnified Person5.6(a)Indemnifying Party7.2(b)(1)IT Assets3.16Licenses3.4Local Licenses3.4Majority Holder9.1Material Contract3.18MergerRecitalsMerger Consideration2.4(e)Merger Shares5.8(a)		7.2(e)(1)
FCC Licenses3.4Financial Statements3.10Holder9.1Indemnified Party7.2(b)Indemnified Person5.6(a)Indemnifying Party7.2(b)(1)IT Assets3.16Licenses3.4Local Licenses3.4Majority Holder9.1Material Contract3.18MergerRecitalsMerger Consideration2.4(e)Merger Shares5.8(a)		
Financial Statements3.10Holder9.1Indemnified Party7.2(b)Indemnified Person5.6(a)Indemnifying Party7.2(b)(1)IT Assets3.16Licenses3.4Local Licenses3.4Majority Holder9.1Material Contract3.18MergerRecitalsMerger Consideration2.4(e)Merger Shares5.8(a)	-	
Holder9.1Indemnified Party7.2(b)Indemnified Person5.6(a)Indemnifying Party7.2(b)(1)IT Assets3.16Licenses3.4Local Licenses3.4Majority Holder9.1Material Contract3.18MergerRecitalsMerger Consideration2.4(e)Merger Shares5.8(a)		÷ · ·
Indemnified Party7.2(b)Indemnified Person5.6(a)Indemnifying Party7.2(b)(1)IT Assets3.16Licenses3.4Local Licenses3.4Majority Holder9.1Material Contract3.18MergerRecitalsMerger Consideration2.4(e)Merger Shares5.8(a)		
Indemnified Person5.6(a)Indemnifying Party7.2(b)(1)IT Assets3.16Licenses3.4Local Licenses3.4Majority Holder9.1Material Contract3.18MergerRecitalsMerger Consideration2.4(e)Merger Shares5.8(a)		
Indemnifying Party7.2(b)(1)IT Assets3.16Licenses3.4Local Licenses3.4Majority Holder9.1Material Contract3.18MergerRecitalsMerger Consideration2.4(e)Merger Shares5.8(a)	•	
IT Assets3.16Licenses3.4Local Licenses3.4Majority Holder9.1Material Contract3.18MergerRecitalsMerger Consideration2.4(e)Merger Shares5.8(a)		
Licenses3.4Local Licenses3.4Majority Holder9.1Material Contract3.18MergerRecitalsMerger Consideration2.4(e)Merger Shares5.8(a)		
Local Licenses3.4Majority Holder9.1Material Contract3.18MergerRecitalsMerger Consideration2.4(e)Merger Shares5.8(a)		
Majority Holder9.1Material Contract3.18MergerRecitalsMerger Consideration2.4(e)Merger Shares5.8(a)		* • •
Material Contract3.18MergerRecitalsMerger Consideration2.4(e)Merger Shares5.8(a)		
MergerRecitalsMerger Consideration2.4(e)Merger Shares5.8(a)		9.1
Merger Consideration2.4(e)Merger Shares5.8(a)		3.18
Merger Shares 5.8(a)		
Merger Shares 5.8(a)		2.4(e)
		5.8(a)
Merger Sub Preamble	Merger Sub	Preamble

Most Recent Financial Statements 3.10 Most Recent Fiscal Month End 3.10 Most Recent Fiscal Year End 3.10 Parent Preamble Parent Indemnified Party 7.2(a)(1) Parent SEC Documents 4.6 Preamble Parties Per Share Merger Consideration 2.4(e)Preferred Stock 3.2 Prospectus 9.1 **Reserved Insurance Proceeds** 7.2(a)(1) Securities. 9.1 2.4(f)(2)(A)Series A Cash Amount Series A Preferred 3.2 Series B Preferred 3.2 Series C Preferred 3.2 Series D Preferred 3.2 Series E Preferred 3.2 Shelf Registration 9.2 Shelf Registration Period 9.2(b) Shelf Registration Statement 9.1 SOX Act 4.6 State Licenses 3.4 Stockholder 3.2 Surviving Corporation 2.1 Surviving Corporation New Plans 5.7(b) Termination Date 8.1(D) Third Party Claim 7.2(b)(1) Termination Fee 8.2(d) Warrants 3.2(b) Withheld Cash 2.4(f)(4)

(c) Any reference to any federal, state, local or foreign statute or law shall be deemed also to refer to all rules and regulations promulgated thereunder, unless the context requires otherwise. The words "include," "includes" and "including" shall mean include, includes or including without limitation. The phrase "business day" shall mean any day other than a day on which banks in the State of New York or the State of Michigan are required or authorized to be closed. The phrases "herein," "hereof," "hereunder" and words of similar import shall be deemed to refer to this Agreement as a whole, including the Exhibits and Schedules hereto, and not to any particular provision of this Agreement. The word "or" shall be inclusive and not exclusive. Any pronoun shall include the corresponding masculine, feminine and neuter forms.

ARTICLE II THE TRANSACTION

2.1. The Merger. On and subject to the terms and conditions of this Agreement, Merger Sub will merge with and into Company at the Effective Time and the separate corporate existence of Merger Sub will thereupon cease. Company shall be the surviving corporation in the Merger (the "Surviving Corporation"), and the separate corporate existence of Company, with all of its rights, privileges, summary powers and franchises, shall continue unaffected by the Merger, except as provided for in Section 2.4.

2.2. The Closing. The closing of the transactions contemplated by this Agreement (the "Closing") shall take place (i) at the offices of Dykema Gossett PLLC, 400 Renaissance Center, Detroit, Michigan, commencing at 9:00 A.M. local time on the first day that is at least three business day after the satisfaction or waiver of all conditions set forth in ARTICLE VI (other than those conditions that by their terms are to be satisfied at the Closing, but subject to the satisfaction or waiver of those conditions) or (ii) at such other place and time or on such other date as Parent and Company may agree in writing (the "Closing Date").

2.3. Actions at the Closing. At the Closing, (i) Company will deliver to Parent and Merger Sub the various certificates, instruments and documents referred to in Section 6.2; (ii) Parent and Merger Sub will deliver to Company the various certificates, instruments and documents referred to in Section 6.3; (iii) Company and Merger Sub will execute and file with the Michigan Department of Labor & Economic Growth, Bureau of Commercial Services, a Certificate of Merger (the "Certificate of Merger") as provided in Section 450.1707 of the Michigan Law and (iv) Parent will deliver or cause to be delivered the Exchange Fund to the Exchange Agent in the manner provided below in this ARTICLE II.

2.4. Effect of Merger.

(a) General. The Merger shall become effective at the time the Certificate of Merger is filed with the Michigan

Department of Labor & Economic Growth, Bureau of Commercial Services, or at such later time as the Parties may agree in writing and specify in the Certificate of Merger (the

"Effective Time"). The Merger shall have the effects set forth in the Michigan Law, and from and after the Effective Time, the Surviving Corporation shall possess all

the rights, powers, privileges and franchises and be subject to all the obligations, liabilities, restrictions and disabilities of the Constituent Corporations, all as

provided under the Michigan Law. The Surviving Corporation may, at any time after the Effective Time, take any action (including executing and delivering any

document) in the name and on behalf of either Company or Merger Sub in order to carry out and effectuate the transactions contemplated by this Agreement.

(b) Articles of Incorporation. At the Effective Time, the articles of incorporation of the Surviving Corporation shall be amended to read in their entirety as did the articles of incorporation of Merger Sub in effect immediately prior to the Effective Time, except that the name of the Surviving Corporation shall be LDMI Telecommunications, Inc., and, as so amended, shall be the articles of incorporation of the Surviving Corporation until thereafter amended in accordance with their terms and as provided by law.

(c) By-laws. The By-laws of the Surviving Corporation shall be amended and restated at and as of the Effective Time to read in their entirety as did the By-laws of Merger Sub in effect immediately prior to the Effective Time and shall be the By-laws of the Surviving Corporation until amended in accordance with their terms and as provided by law.

(d) Directors and Officers. The directors and officers of Merger Sub immediately prior to the Effective Time shall be the directors and officers of the Surviving Corporation at and as of the Effective Time (retaining their respective positions and terms of office), until the earlier of their respective resignation, removal or otherwise ceasing to be a director or officer, respectively, or until their respective successors are duly elected and qualified, as the case may be.

(e) Conversion of Company Shares. At and as of the Effective Time, (A) each issued and outstanding Company Share will be converted into the right to receive the Merger Consideration set forth below in Section 2.4(f) (as to each Company Share, its "Per Share Merger Consideration"), and all such Company Shares will no longer be outstanding, will be canceled and retired and will cease to exist, and each holder of a certificate representing any such Company Shares will thereafter cease to have any rights with respect to such Company Shares, except the right to receive the respective Per Share Merger Consideration for each such Company Share to which the holder of such Company Shares is entitled pursuant to Section 2.4(f) upon the surrender of such certificate in accordance with Section 2.4(f) (collectively, the "Merger Consideration") except that the Stock Component of the Per Share Merger Consideration shall be subject to equitable and proportionate adjustment in the event of any Parent stock split, stock dividend, reverse stock split, subdivision, combination, recapitalization or similar event or action respecting the Parent Shares between the date of this Agreement and the Closing Date, and (B) each Company Share owned by Company or any Subsidiary of Company shall be canceled and retired without payment of any consideration therefor and shall cease to exist. No Company Share shall be deemed to be outstanding or to have any rights other than those set forth above in this Section 2.4(e) after the Effective Time. Notwithstanding anything to the contrary in this Section 2.4(e), no fractional Parent Shares shall be issued to then former holders of Company Shares. In lieu thereof, each then former holder of a Company Share who would otherwise have been entitled to receive a fraction of a Parent Share (after taking into account all certificates delivered by such then former holder at any one time) shall receive an amount in cash equal to such fraction of a Parent Share multiplied by the Parent Share Valuation.

(f) Per Share Merger Consideration. The Company Shares will be converted into the respective rights to receive the following Per Share Merger Consideration, which shall be payable in cash, Parent Shares or a combination thereof, as provided, and subject to the limitations set forth, below:

(1) each share of Common Stock will be converted into the right to receive \$0.20 in cash;

۶.,

(2) subject to adjustment as provided in Section 2.4(f)(4), each share of Preferred Stock, except the shares held by a Dissenting Stockholder, will be converted into the right to receive the consideration set opposite the Series reference below:

A. Series A Preferred – Merger Consideration of (i) cash in the amount equal to the result of (x) the product of (I) the Series A Ratio times (II) the Remaining Cash, divided by (y) the number of shares of Series A Preferred issued and outstanding as of the Effective Time and (iii) Parent Shares in the number equal to the result of (x) (i) the product of (I) the Series A Ratio times (II) the Stock Value, (ii) divided by the Parent Share Valuation, divided by (y) the number of shares of Series A Preferred issued and outstanding as of the Effective Time;

B. Series B Preferred – Merger Consideration of (i) cash in the amount equal to the result of (x) the product of (I) the Series B Ratio times (II) the Remaining Cash, divided by (y) the number of shares of Series B Preferred issued and outstanding as of the Effective Time and (iii) Parent Shares in the number equal to the result of (x) (i) the product of (I) the Series B Ratio times (II) the Stock Value, (ii) divided by the Parent Share Valuation, divided by (y) the number of shares of Series B Preferred issued and outstanding as of the Effective Time;

8 J 🛓

C. Series C Preferred – Merger Consideration of (i) cash in the amount equal to the result of (x) the product of (I) the Series C Ratio <u>times</u> (II) the Remaining Cash, <u>divided</u> by (y) the number of shares of Series C Preferred issued and outstanding as of the Effective Time and (iii) Parent Shares in the number equal to the result of (x) (i) the product of (I) the Series C Ratio <u>times</u> (II) the Stock Value, (ii) <u>divided</u> by the Parent Share Valuation, <u>divided</u> by (y) the number of shares of Series C Preferred issued and outstanding as of the Effective Time;

D. Series D Preferred – Merger Consideration of (i) cash in the amount equal to the result of (x) the product of (I) the Series D Ratio times (II) the Remaining Cash, divided by (y) the number of shares of Series D Preferred issued and outstanding as of the Effective Time and (iii) Parent Shares in the number equal to the result of (x) (i) the product of (I) the Series D Ratio times (II) the Stock Value, (ii) divided by the Parent Share Valuation, divided by (y) the number of shares of Series D Preferred issued and outstanding as of the Effective Time; and

E. Series E Preferred – Merger Consideration of (i) cash in the amount equal to the result of (x) the product of (l) the Series E Ratio $\underline{\text{times}}$ (l1) the Remaining Cash, $\underline{\text{divided}}$ by (y) the number of shares of Series E Preferred issued and outstanding as of the Effective Time and (iii) Parent Shares in the number equal to the result of (x) (i) the product of (l) the Series E Ratio $\underline{\text{times}}$ (II) the Stock Value, (ii) $\underline{\text{divided}}$ by the Parent Share Valuation, $\underline{\text{divided}}$ by (y) the number of shares of Series E Preferred issued and outstanding as of the Effective Time.

(3) Notwithstanding the provisions of Section 2.4(f)(1) or 2.4(f)(2), the aggregate of the Merger Consideration into which all of the issued and outstanding Company Shares shall be converted plus the cash payable in respect of the Employee Options and the Warrants pursuant to Section 2.4(g) shall not exceed the Total Merger Consideration, the Cash Component shall not exceed the Cash Cap and the Stock Component shall not exceed the Stock Cap.

(4) Notwithstanding the provisions of Section 2.4(f)(2), the amounts of cash and numbers of Parent Shares that are otherwise deliverable pursuant to the terms of Section 2.4(f)(2) to the Preferred Stockholders are subject to adjustment as follows: (i) at the Effective Time, the aggregate amount of cash (the "Dissenter Cash") and number of Parent Shares (such Parent Shares, the "Dissenter Shares") that are otherwise deliverable pursuant to Section 2.4(f)(2) to any Preferred Stockholder that is not a Consenting Stockholder, together with the amount of cash (the "Withheld Cash") equal to the value of such number of Dissenter Shares, valued at the Parent Share Valuation, shall be set aside and held by the Exchange Agent until the Resolution Date, and any distribution to any Consenting Stockholder of the Cash Component of such Preferred Stockholder prior to the Resolution Date shall be reduced by such Consenting Stockholder's pro rata share of the Withheld Cash, less, in each case, the 5% thereof that is required to be withheld and deposited in escrow as Escrowed Consideration pursuant to Section 2.5; and (ii) if, as of the Resolution Date, any Preferred Stockholder that is not a Consenting Stockholder has taken such action as leaves it only with the right to receive a cash dissenter payment under Section 450.1769 of the Michigan Law, the number of Parent Shares that are deliverable as Merger Consideration pursuant to Section 2.4(f)(2)to each Consenting Stockholder will be deemed in all respects to have been adjusted to increase such amount by each such Consenting Stockholder's pro rata share of the Dissenter Shares, other than the portion constituting Escrowed Consideration, and the amount of cash that is deliverable as Merger Consideration pursuant to Section 2.4(f)(2) to each Consenting Stockholder will be deemed in all respects to have been adjusted to decrease such amount by each such Consenting Stockholder's pro rata share of the Withheld Cash and (iii) if, as of the Resolution Date, any Preferred Stockholder that is not a Consenting Stockholder has not taken the necessary action to be entitled to payment for its shares of Preferred Stock under Section 450.1767 of the Michigan Law, any cash withheld from distributions pursuant to clause (i) of this Section 2.4(f)(4), other than the portion constituting Escrowed Consideration, shall be distributed pro rata to the Consenting Stockholders from whom so withheld and the amounts of cash and numbers of Parent Shares that are deliverable pursuant to Section 2.4(f)(2) to the Preferred Stockholders will not be adjusted. In the event of adjustments as provided in clause (ii) of this Section 2.4(f)(4), the Exchange Agent shall deliver to Parent from the Exchange Fund an amount of cash equal to sum of the Withheld Cash and the Dissenter Cash, other than the portion constituting Escrowed Consideration.

2.

(g) Conversion of Employee Stock Options; Warrants. At or immediately prior to the Effective Time, (i) each Employee Option that is outstanding, unexercised and unexpired, as of the Effective Time, whether or not vested or exercisable and without regard to any agreements qualifying the right to retain or exercise any such Employee Option, and (ii) each Warrant that is outstanding, unexercised and unexpired as of the Effective Time shall be canceled, and each holder of any such Employee Option or Warrant shall have the right to receive for each such Employee Option or Warrant cash in the amount that equals the amount by which the Common Stock Per Share Merger Consideration exceeds the per share exercise price or base price, as the case may be, of such Option or Warrant as of the Effective Time.

(h) Conversion of Capital Stock of Merger Sub. At and as of the Effective Time, each share of common stock, \$.01 par value per share, of Merger Sub shall be converted into one share of common stock, \$.01 par value per share, of the Surviving Corporation.

2.5. Procedure for Exchange.

(a) Immediately after the Effective Time, (A) Parent shall furnish to StockTrans, Inc., its transfer agent, or such other bank or trust company reasonably acceptable to Company to act as exchange agent (the "Exchange Agent"), a corpus (the "Exchange Fund") consisting of Parent Shares and cash sufficient to permit the Exchange Agent to make full payment of the Merger Consideration to the holders of all of the issued and outstanding Company Shares (other than any Company Shares owned by Company), less such number and amount of the Parent Shares and cash that equals 5% of the aggregate number of Parent Shares and amount of cash that would have been deliverable to the Preferred Stockholders as Merger Consideration pursuant to Section 2.4(f)(2) were all outstanding shares of Preferred Stock as of the Effective Time to be converted and without giving effect to the withholding therefrom pursuant to this Section 2.5(a) or to any adjustment pursuant to Section 2.4(f)(4), which number of Parent Shares and amount of cash (the "Escrowed Consideration") will be withheld from the amounts otherwise to be delivered to the holders of Preferred Stock as Merger Consideration to which each such holder of Preferred Stock would, but for this Section 2.5(a), be entitled to be delivered pursuant to this Agreement, pro rata in proportion to the respective amounts of cash and Parent Shares otherwise so deliverable and (B) Parent will cause the Exchange Agent to mail a letter of transmittal (with instructions for its use) in a form to be mutually agreed upon by Company and Parent prior to Closing to each holder of issued and outstanding Company Shares (other than any Company Shares owned by Company) for the holder to use in surrendering the certificates that, immediately prior to the Effective Time, represented his or its Company Shares against payment of the Merger Consideration to which the holder is entitled pursuant to Section 2.4(f), subject to the escrow of the Escrowed Consideration pursuant to the Escrow Agreement. Notwithstanding the withholding of the Escrowed Consideration and deposit thereof with the Escrow Agent pursuant to Section 2.6(a), each Preferred Stockholder shall, for all purposes of Section 2.4, be deemed to have received any such Escrowed Consideration so withheld and deposited. Upon surrender to the Exchange Agent of these certificates, together with the letter of transmittal, duly executed and completed in accordance with the letter of transmittal instructions, subject to the escrow of the Escrow Amount pursuant to the Escrow Agreement, Parent shall promptly cause to be issued a certificate representing that number of whole Parent Shares, if any, to which the Persons are entitled (in each case bearing the legends set forth in Sections 5.8(b) and 5.8(d) and after giving effect to any required tax withholdings and the withholding of the Escrowed Consideration) and a check representing the sum of (x) the Cash Component (after giving effect to any required tax withholdings, the withholding of the Escrowed Consideration and the provisions of Section 2.4(f)(4) and (y) the amount of cash in lieu of any fractional shares to which the Persons are entitled, after giving effect to any required tax withholdings. No interest will be paid or accrued on the Cash Component or the cash in lieu of fractional shares payable to recipients of Parent Shares. If payment is to be made to a Person other than the registered holder of the certificate surrendered, it shall be a condition of payment that the surrendered certificate must be properly endorsed or otherwise in proper form for transfer and that the Person requesting such payment shall pay any transfer or other taxes required by reason of the payment to a Person other than the registered holder of the certificate surrendered or establish to the reasonable satisfaction of the Surviving Corporation or the Exchange Agent that this tax has been paid or is not applicable. If any certificate representing Company Shares is lost, stolen or destroyed, upon the making of an affidavit of that fact by the Person claiming a certificate to be lost, stolen or destroyed, the Exchange Agent will issue in exchange for this lost, stolen or destroyed certificate the Merger Consideration deliverable in respect thereof except that the Person to whom this Merger Consideration is paid shall, as a condition precedent to the payment thereof, indemnify the Surviving Corporation in a manner reasonably satisfactory to it against any claim that may be made against the Surviving Corporation with respect to the certificate alleged to have been lost, stolen or destroyed. No dividends or other distributions declared after the Effective Time with respect to Parent Shares and payable to the holders of record thereof will be paid to the holder of any unsurrendered certificate until the holder thereof shall surrender this certificate in accordance with this Section 2.5. After the surrender of a certificate in accordance with this Section 2.5, the record holder thereof is entitled to receive any such dividends or other distributions, without any interest thereon, which previously had become payable with respect to the Parent Shares represented by such certificate. No holder of an unsurrendered certificate is entitled, until the surrender of such certificate, to vote the Parent Shares into which his or its Company Shares shall have been converted.

(b) Parent shall pay, or shall cause the Surviving Corporation to pay, all charges and expenses of the Exchange Agent.

2.6. Escrow.

(a) At the Effective Time, Parent, Merger Sub, Company, the Representative and the Escrow Agent shall execute and deliver an escrow agreement substantially in the form of the attached Exhibit A (the "*Escrow Agreement*") under which U.S. Bank Corporate Trust Services or other Person mutually satisfactory to Parent and Company shall act as escrow agent (the "*Escrow Agent*") with respect to the Escrowed Consideration deposited with the Escrow Agent. Parent shall deposit the Escrowed Consideration with the Escrow Agent, which shall be withheld from the Merger Consideration as provided in Section 2.5 in connection with the indemnification obligations set forth in Section 7.2.

(b) Subject to the provisions of this Section 2.6 and the Escrow Agreement, the Escrowed Consideration shall be paid to the Preferred Stockholders on the Claim Date, as reduced by the amount of any indemnification payments based on, arising from or in connection with all claims for indemnification asserted in writing prior to the Claim Date pursuant to Section 7.2 that have not been fully resolved.

(c) For all purposes of this Agreement and the Escrow Agreement, whenever Parent Shares shall be required to be delivered to satisfy an indemnity or contribution obligation of any Party hereto, each Parent Share shall be valued at the Parent Share Valuation. In the event of any stock split, reverse stock split, stock combination or reclassification of the Parent Shares or any merger, consolidation or combination of Parent with any other entity or entities, the deemed value specified above for the Parent Shares shall be proportionally adjusted so that the deemed value of the Parent Shares after such event shall be the same as the deemed value of the Parent Shares prior to such event. All such adjustments shall be made successively.

2.7. Closing of Transfer Record. After the Effective Time, no transfer of Company Shares outstanding prior to the Effective Time may be made on the stock transfer books of the Surviving Corporation. If, after the Effective Time, certificates representing such shares are presented for transfer to the Exchange Agent, they shall be canceled and exchanged for Merger Consideration as provided in Section 2.5.

ARTICLE III REPRESENTATIONS AND WARRANTIES OF COMPANY

Except as set forth in the disclosure letter (subject to Section 10.13(c)) delivered to Parent by Company at or prior to entering into this Agreement (the "Company Disclosure Letter"), Company hereby represents and warrants to Parent and Merger Sub that:

3.1. Organization, Qualification and Corporate Power; Transaction Authorization.

۰.

(a) Each of Company and its Subsidiaries is a corporation duly organized, validly existing, and in good standing under the Laws of the jurisdiction of its incorporation and has all requisite corporate or similar power and authority to own and operate its properties and assets and to carry on its business as presently conducted and is qualified to do business and is in good standing as a foreign corporation in each jurisdiction where the ownership or operation of its assets or properties or conduct of its business requires such qualification, except where the failure to be so organized, validly existing, qualified or in good standing, or to have such power or authority, would not be reasonably expected to have a Material Adverse Effect on Company. Section 3.1 of the Company Disclosure Letter lists for each of Company and its Subsidiaries (i) the directors and officers, (ii) the state of incorporation and (iii) the jurisdictions in which the corporation is qualified to do business. Company has delivered to Parent correct and complete copies of the charter and bylaws of each of Company and its Subsidiaries (as amended to date). The minute books (containing the records of meetings of the stockholders, the board of directors and any committees of the board of directors), the stock certificate books, and the stock record books of each of Company and its Subsidiaries is in default under or in violation of any provision of its charter or bylaws.

(b) The Company Board has (i) declared that the Merger and the other transactions contemplated hereby are advisable and has adopted this Agreement, (ii) determined on a reasonable basis (and as contemplated by Section 450.1703a(2)(d) of the Michigan Law) that the holders of the Common Stock, as a class, are to receive consideration under this Agreement upon effectiveness of the Merger that has a fair value in respect of such Common Stock not less than the fair value thereof as of the date of the Company Board's adoption of this Agreement, (iii) approved the amendment (the "Articles Amendment") of Company's Articles of Incorporation to permit the payment of Merger Consideration to the holders of Common Stock as provided herein, (iv) recommended adoption and approval of this Agreement and the transactions contemplated hereby, including the Merger, and of the Articles Amendment by the Stockholders, (v) directed that this Agreement and the transactions contemplated hereby, including the Merger and the Articles Amendment, be submitted to the Preferred Stockholders for their approval and adoption on or before the date of this Agreement, and (vi) taken all necessary action to provide that Employee Options and Warrants outstanding as of the Effective Time will either be converted into cash and cancelled pursuant to Section 2.4(a) or otherwise in all respects cancelled and of no further force and effect from and after the Effective Time. The approval and adoption of this Agreement and the transactions contemplated hereby, including the Merger and the Articles Amendment, require only the Requisite Stockholder Approval. Company has received the consents of the Consenting Stockholders in the Stockholder Consents, a true and correct copy of each of which has been provided to Parent, and such consents so received constitute the Requisite Stockholder Approval. This Agreement and the transactions contemplated hereby. including the Merger and the Articles Amendment, have been duly authorized by all necessary corporate action.

(c) Company has all requisite corporate power and authority and has taken all corporate action necessary in order to execute, deliver and perform its obligations under this Agreement and to consummate the Merger in accordance with its terms. This Agreement is a valid and binding agreement of Company enforceable against Company in accordance with its terms, subject to bankruptcy, insolvency, fraudulent transfer, reorganization, moratorium and similar laws of general applicability relating to or affecting creditors' rights and to general equity principles and except as indemnification obligations hereunder may be limited by applicable securities laws. No Stockholder is entitled to any dissenter's rights with respect to, or other rights of appraisal of, its Company Shares in respect or by reason of the Merger, the Articles Amendment or any of the transactions contemplated hereby, except only that any Preferred Stockholder that is not a Consenting Stockholder shall have the rights under Section 450.1621 of the Michigan Law to dissent from the Articles Amendment.

(d) Company has all requisite corporate power and authority and has taken all corporate action necessary in order to execute, deliver and perform its obligations under the Switch Lease and the Switch Lease is a valid and binding agreement of Company enforceable against Company in accordance with its terms, subject to bankruptcy, insolvency, fraudulent transfer, reorganization, moratorium and similar laws of general applicability relating to or affecting creditors' rights and to general equity principles.

3.2. Capitalization.

(a) As of the date of this Agreement, Company's authorized capitalization consisted of:

(1) 45,000,000 shares of Common Stock, par value \$0.01 per share (the "Common Stock"), of which 7,990,113 shares were issued and outstanding, an aggregate of 7,680,408 shares were issuable upon exercise of outstanding Employee Options and an aggregate of 20,680,628 shares were issuable upon conversion of shares of the Preferred Stock, as set forth below; and

(2) 23,050,000 shares of Preferred Stock, no par value (the "Preferred Stock" and, collectively with the Common Stock, the "Company Shares"), designated as:

- (i) 1,950,000 shares of Series A Senior Convertible Participating Preferred ("Series A Preferred"), of which 1,938,320 shares were issued and outstanding and were convertible into 1,980,687 shares of Common Stock;
- (ii) 5,100,000 shares of Series B Senior Convertible Preferred ("Series B Preferred"), of which 4,902,000 shares were issued and outstanding and were convertible into 5,076,182 shares of Common Stock;
- (iii) 4,500,000 shares of Series C Senior Convertible Preferred ("Series C Preferred"), of which 4,086,085 shares were issued and outstanding and were convertible into 4,921,874 shares of Common Stock;
- (iv) 7,500,000 shares of Series D Senior Convertible Preferred ("Series D Preferred"), of which 4,651,162 shares were issued and outstanding and were convertible into 4,830,917 shares of Common Stock; and
- (v) 4,000,000 shares of Series E Senior Convertible Preferred ("Series E Preferred"), of which 3,870,968 shares were issued and outstanding and were convertible into 3,870,968 shares of Common Stock.

(b) All of the issued and outstanding Company Shares have been duly authorized, are validly issued, fully paid and nonassessable, and, as of the date of this Agreement, are held of record by the respective stockholders as set forth in Section 3.2 of the Company Disclosure Letter (each a "Stockholder" and collectively, the "Stockholders"). The Company has no treasury stock. Other than the Employee Options that are exercisable for 7,680,408 shares of Common Stock as of the date of this Agreement and warrants ("Warrants") to purchase 311,581 shares of Common Stock as of the date of this Agreement (in each case of the Employee Options and Warrants, as identified in Section 3.2 of the Company Disclosure Letter with all relevant material information including but not limited to exercise price, exercise term, transferability restrictions, employment related conditions (if any) and vesting rights), there are no outstanding or authorized options, warrants, purchase rights, subscription rights, conversion rights, exchange rights, or other contracts or commitments that could require Company to issue, sell or otherwise cause to become outstanding any of its capital stock. There are no outstanding or authorized stock appreciation, phantom stock, profit participation or similar rights with respect to Company. There are no voting trusts, proxies or other agreements or understandings with respect to the voting of the capital stock of Company, provided that the foregoing representation and warranty is to the Knowledge of Company as respects voting trusts, proxies or other agreements or understandings to which none of Company and its Subsidiaries is a party or has acknowledged in writing. Each Stockholder holds of record and, to Company's Knowledge, owns beneficially, as of the date of this Agreement, the number of Company Shares set forth next to his or its name and record address in Section 3.2 of the Company Disclosure Letter, free and clear of any restrictions on transfer (other than any generally applicable restrictions on transfer under the Securities Act and state securities laws), Taxes, Security Interests, options, warrants, purchase rights, contracts, commitments, equities, claims and demands. To Company's Knowledge, no Stockholder is a party to any option, warrant, purchase right or other contract or commitment that could require the Stockholder to sell, transfer or otherwise dispose of any capital stock of Company (other than this Agreement). From and after the Effective Time, all Employee Options and Warrants will be cancelled and shall represent no rights of any holder thereof except only the right of the holders of certain of the Employee Options and Warrants to receive cash as and the extent provided in Section 2.4(a).

3.3. Noncontravention; Consents. Neither the execution and the delivery of this Agreement, nor

the consummation of the transactions contemplated hereby, will (i), assuming the filing of the Certificate of Amendment as contemplated by Section 2.3, violate any provision of the charter or bylaws of any of Company and its Subsidiaries or (ii), assuming compliance with the matters referred to in the next sentence of this Section 3.3, (A) violate any Laws or Governmental Order to which any of Company and its Subsidiaries is subject or (B) with or without notice, lapse of time or both, conflict with, result in a breach of, constitute a default under, result in the acceleration of, create in any party the right to accelerate, terminate, modify or cancel, or require any notice under any agreement, contract, lease, license, instrument or other arrangement to which any of Company and its Subsidiaries is a party or by which it is bound or to which any of its assets is subject (or result in the imposition of any Security Interest upon any of its assets), except, in the case of clause (ii), such violation, breach, default, acceleration or other change that would not, individually or in the aggregate, reasonably be expected to have a Material Adverse Effect on Company. Except for (a) any FCC Consent, State PUC Consent or consent or approval of any other Governmental Entity identified in Section 3.3 of the Company Disclosure Letter, in each case as required by applicable Laws, (b) the filing of the Certificate of Amendment with the Michigan Department of Labor & Economic Growth, Bureau of Commercial Services pursuant to the Michigan Law, (c) the filing of the Certificate of Merger with the Michigan Department of Labor & Economic Growth, Bureau of Commercial Services pursuant to the Michigan Law and of appropriate documents with relevant authorities of other states in which Company is qualified to do business to reflect such Certificate of Merger filing, and (d) any other third party approvals as are reflected in Section 3.3 of the Company Disclosure Letter, including with respect to any Computer Software program and databases (other than commercial, non-exclusive end-user licenses having a total consideration, with respect to each license, of less than \$50,000), the execution, delivery and performance by Company of this Agreement and the transactions contemplated hereby do not require any consents, waivers, authorizations or approvals of, or filings with, any Governmental Entity or any other third Person except for those that the failure to make or obtain would not be reasonably expected to have a Material Adverse Effect on Company.

3.4. Compliance with Laws, Licenses.

• .,

(a) Company and its Subsidiaries are not in violation of any Laws, License or Governmental Order applicable to any of the businesses in which any of Company and its Subsidiaries is engaged except to the extent that noncompliance would not reasonably be expected to have a Material Adverse Effect on Company and no action, suit, proceeding, hearing, investigation, charge, complaint, claim, demand or notice has been filed or, to Company's Knowledge, commenced, and currently pending, against any of them alleging any failure so to comply, except to the extent such failure would not reasonably be expected to have a Material Adverse Effect on Company.

(b) Company and its Subsidiaries hold all permits, licenses, certificates, variances, exemptions, orders, approvals, tariffs, rate schedules and similar documents from Governmental Entities (collectively, "*Licenses*") that are necessary to own, lease and operate the assets and properties they currently own, lease and operate and to conduct their respective businesses and operations in the manner currently conducted, except where the failure to hold such Licenses would not reasonably be expected to have a Material Adverse Effect on Company. Section 3.4 of the Company Disclosure Letter sets forth all Licenses issued or granted to Company or any of its Subsidiaries by the FCC ("*FCC Licenses*"), all Licenses issued or granted to Company or any of its Subsidiaries by any state public utility commission or other state commission or authority regulating telecommunications businesses or services ("*State Licenses*") and all Licenses issued or granted to Company or any of its Subsidiaries by any local government regulating telecommunications businesses or services ("*Company or any of its Subsidiaries by any local government ("Local Licenses"*") and all other material Licenses held by Company or its Subsidiaries, together with any pending applications filed by Company or its Subsidiaries, together with any pending applications filed by Company or its Subsidiaries for Communications Licenses or other material Licenses that would be Licenses if issued or granted or for modification, extension or renewal of any License. Company has delivered to Parent correct and complete copies of all Licenses (including the applications related thereto) and all pending applications listed on Section 3.4 of the Company Disclosure Letter.

(c) Each of Company and its Subsidiaries is in compliance in all material respects with each Communications License. Each of Company and its Subsidiaries is in compliance with (A) its obligations under each of the Licenses and (B) the rules and regulations of the Governmental Entity issuing such Licenses, except for any failures to be in compliance that would not, individually or in the aggregate, reasonably be expected to result in a Material Adverse Effect on Company. There is not pending or, to the Knowledge of Company, threatened in writing before the FCC or any other Governmental Entity any material proceeding, notice of violation, order of forfeiture or complaint or investigation against Company or any of its Subsidiaries relating to any of the Licenses, except, in the case of Licenses other than Communications Licenses, for any of the foregoing that would not, individually or in the aggregate, reasonably be expected to result in a Material Adverse Effect on Company. As of the date of this Agreement, Company has received no written notice, and, as to any Communications License, has no Knowledge, that any event has occurred with respect to any such License or application that would permit the revocation, termination, suspension or denial thereof or would result in any impairment of the rights of the holder thereof. No written notice has been received and to Company's Knowledge no investigation or review is pending or threatened in writing by any Governmental Entity with regard to any alleged violation by Company or any of its Subsidiaries of any License or any alleged failure by Company or any of its Subsidiaries to have any Licenses. The actions of the applicable Governmental Entities granting all Licenses have not been reversed, stayed, enjoined, annulled or suspended, and there is not pending or, to the Knowledge of Company, threatened in writing, any material application, petition, objection or other pleading with the FCC or any other Governmental Entity that challenges or questions the validity of or any rights of the holder under any License, except, in the case of Licenses other than Communications Licenses, for any of the foregoing that would not, individually or in the aggregate, reasonably be expected to result in a Material Adverse Effect on Company.

3.5. Customers. Listed in Section 3.5 of the Company Disclosure Letter are the names and addresses of the 50 most significant separate customer billing accounts (by revenue) of Company and its Subsidiaries for the twelve-month period ended December 31, 2004 and the amount for which each such customer was invoiced during such period. As of the date of this Agreement, Company has not received any written notice nor does it have any Knowledge that any of the above-listed significant customers of Company or any of its Subsidiaries has ceased, or will cease, to use the products, equipment, goods or services of Company or any of its Subsidiaries, or has substantially reduced or will substantially reduce, the use of such products, equipment, goods or services at any time.

3.6. No General Solicitation. Neither Company nor, to the Knowledge of Company, any Affiliate of Company or any person acting on its or their behalf has engaged in any form of general solicitation or general advertising (within the meaning of Regulation D under the Securities Act) in connection with any offer or sale of the Parent Shares to be delivered pursuant to this Agreement.

3.7. Brokers' Fees. Except for fees payable to Brown Brothers Harriman & Co. as set forth in the Company Disclosure Letter, neither Company nor its Subsidiaries has any Liability or obligation to pay any fees or commissions to any broker, finder, or agent with respect to the transactions contemplated by this Agreement.

3.8. Title to Assets. Company and its Subsidiaries have good and marketable title to, or a valid leasehold interest in, the properties and assets used by them, located on their premises or shown on the Most Recent Balance Sheet or acquired after the date thereof, free and clear of all Security Interests, except for (i) properties and assets disposed of in the Ordinary Course of Business since the date of the Most Recent Balance Sheet, (ii) Security Interests disclosed in Section 3.8 of the Company Disclosure Letter, (iii) Security Interests or imperfections of title that are not, individually, material in character, amount or extent and that do not, individually or in the aggregate, materially detract from the value or materially interfere with the present or presently contemplated use by Company of the assets subject thereto or affected thereby, (iv) Security Interests arising under conditional sale or title retention agreements, real property leases, equipment leases or lease purchase agreements that are disclosed in Section 3.8 of the Company Disclosure Letter, (v) Security Interests arising in the Ordinary Course of Business (including, but not limited to, liens for Taxes or governmental charges or levies, Security Interests of mechanics, carriers, workmen and repairmen, Security Interests incurred in connection with workmen's compensation, unemployment insurance, social security and other like laws) for amounts that are not delinquent, except such Security Interests as are being contested in good faith, and (vi) Security Interests or imperfections of title that would not, individually or in the aggregate, reasonably be expected to have a Material Adverse Effect on Company.

3.9. Subsidiaries. Section 3.9 of the Company Disclosure Letter sets forth for each Subsidiary of Company (i) its name and jurisdiction of incorporation, (ii) the number of shares of authorized capital stock of each class of its capital stock, (iii) the number of issued and outstanding shares of each class of its capital stock, the names of the holders thereof, and the number of shares held by each such holder, and (iv) the number of shares of its capital stock held in treasury. All of the issued and outstanding shares of capital stock of each Subsidiary of Company have been duly authorized and are validly issued, fully paid and nonassessable. All of the outstanding shares of each Subsidiary of Company is free and clear of any restrictions on transfer (other than generally applicable restrictions under the Securities Act and state securities laws), Taxes, Security Interests, options, warrants, purchase rights, contracts, commitments, equities, claims and demands. There are no outstanding or authorized options, warrants, purchase rights, subscription rights, conversion rights, exchange rights or other contracts or commitments that could require any of Company and its Subsidiaries to sell, transfer or otherwise dispose of any capital stock, of any of its Subsidiary of Company. There are no voting trusts, proxies or other agreements or understandings with respect to the voting of any capital stock of any Subsidiary of Company. None of Company and its Subsidiaries or other agreements or understandings with respect to the voting of any capital stock of any Subsidiary of Company. None of Company and its Subsidiaries controls directly or indirectly or has any direct or indirect equity participation in any corporation, partnership, trust or other business association, which is not a Subsidiary of Company.

3.10. Financial Statements. Company has furnished to Parent prior to the date of this Agreement audited consolidated balance sheets and statements of income, changes in stockholders' equity, and cash flow as of, and for the fiscal years ended, December 31, 2002, December 31, 2003 and December 31, 2004 (the "Most Recent Fiscal Year End") for Company and its Subsidiaries, together with the reports of Company's independent auditors thereon, and the unaudited consolidated balance sheet and statements of income, changes in stockholders' equity, and cash flow as of, and for the fiscal quarter ended, March 31, 2005. Such financial statements so furnished and the subsequent unaudited quarterly financial statements that may be delivered to Parent as provided in this Agreement are collectively the "Financial Statements." The Financial Statements (including the notes thereto) present fairly (or will present fairly, in the case of those furnished after the date hereof) the financial condition of Company and its Subsidiaries as of their respective dates and the results of operations of Company and its Subsidiaries for their respective periods (subject, in the case of the unaudited statements, to notes and normal year-end audit adjustments that will not be material in amount or effect), in each case in accordance with GAAP applied on a consistent basis throughout the periods covered thereby.

3.11. Events Subsequent to Most Recent Fiscal Year End. Since the Most Recent Fiscal Year End, Company and its Subsidiaries have conducted their businesses only in, and have not engaged in any material transaction other than in accordance with, the Ordinary Course of Business of Company and its Subsidiaries. Since the Most Recent Fiscal Year End and prior to the date hereof, there has not been any Material Adverse Effect on Company. Since the Most Recent Fiscal Year End and prior to the date hereof:

(a) none of Company and its Subsidiaries has sold, leased, transferred, or assigned any of its assets, tangible or intangible, other than in the Ordinary Course of Business;

(b) none of Company and its Subsidiaries has entered into any agreement, contract, lease or license (or series of related agreements, contracts, leases and licenses with the same other party or Affiliates of such other party) either involving more than \$250,000 or other than in the Ordinary Course of Business;

(c) no party (including any of Company and its Subsidiaries) has accelerated, terminated, modified or cancelled any agreement, contract, lease or license (or series of related agreements, contracts, leases and licenses with the same other party or Affiliates of such other party) involving more than \$250,000 to which any of Company and its Subsidiaries is a party or by which any of them is bound;

(d) none of Company and its Subsidiaries has made any capital expenditure (or series of related capital expenditures) either involving more than \$250,000 or other than in the Ordinary Course of Business;

(e) none of Company and its Subsidiaries has made any capital investment in, any loan to, or any acquisition of the securities or assets of, any other Person (or series of related capital investments, loans or acquisitions with the same other party or Affiliates of such other party) either involving more than \$50,000 or other than in the Ordinary Course of Business;

(f) none of Company and its Subsidiaries has issued any note, bond or other debt security or created, incurred, assumed or guaranteed any indebtedness for borrowed money or capitalized lease obligation either involving more than \$50,000 singly or \$250,000 in the aggregate;

(g) none of Company and its Subsidiaries has delayed or postponed the payment of accounts payable or other Liabilities other than in the Ordinary Course of Business;

(h) none of Company and its Subsidiaries has cancelled, compromised, waived or released any right or claim (or series of related rights and claims) either involving more than \$50,000 or other than in the Ordinary Course of Business;

(i) none of Company and its Subsidiaries has granted any license or sublicense of any rights under or with respect to any Intellectual Property;

(j) other than as contemplated by this Agreement, there has been no change made or authorized in the charter or bylaws of any of Company and its Subsidiaries;

(k) none of Company and its Subsidiaries has issued, sold or otherwise disposed of any of its capital stock, or granted any options, warrants or other rights to purchase or obtain (including upon conversion, exchange or exercise) any of its capital stock except for the issuance of shares of Common Stock upon exercise of Employee Stock Options or Warrants outstanding as of the Most Recent Fiscal Year End in accordance with their terms;

(1) none of Company and its Subsidiaries has declared, set aside or paid any dividend or made any distribution with respect to its capital stock (whether in cash, property, stock or any combination thereof) or redeemed, purchased or otherwise acquired any of its capital stock;

(m) none of Company and its Subsidiaries has experienced any damage, destruction or loss (whether or not covered by insurance) to its property that could reasonably be expected to have a Material Adverse Effect on Company;

(n) none of Company and its Subsidiaries (x) has made any loan or advance to, or entered into any other transaction with, any of its directors, officers or stockholders, or (y) made any loans or advances to, or entered into any other transactions with, any of its employees that were in the aggregate as to an employee in excess of \$10,000 at any one time outstanding, other than, in the case of transactions with officers or employees (including stockholders in their capacity as officers or employees) referenced in either of clause (x) or (y), employment arrangements in the Ordinary Course of Business;

(o) none of Company and its Subsidiaries has entered into any employment contract or collective bargaining agreement, written or oral, or modified the terms of any such existing contract or agreement;

(p) none of Company and its Subsidiaries has granted any increase in the base compensation of any of its directors or officers or, other than in the Ordinary Course of Business, any of its employees;

(q) none of Company and its Subsidiaries has adopted, amended, modified or terminated any bonus, profit-sharing, incentive, severance or other plan, contract or commitment for the benefit of any of its directors, officers or employees (or taken any such action with respect to any other Employee Benefit Plan);

(r) none of Company and its Subsidiaries has made any other change in employment terms for any of its directors or officers;

(s) none of Company and its Subsidiaries has made or pledged to make any charitable or other capital contribution other than in the Ordinary Course of Business; and

(t) none of Company and its Subsidiaries has committed to any of the foregoing.

3.12. Undisclosed Liabilities. There are no liabilities or obligations of Company or any Subsidiary of Company, whether or not accrued, contingent or otherwise and whether or not required to be disclosed, nor any other facts or circumstances that would reasonably be expected to result in any liabilities or obligations of Company or any of its Subsidiaries, other than:

(a) liabilities or obligations to the extent (i) reflected on the Most Recent Balance Sheet or (ii) readily apparent in the notes thereto;

(b) liabilities or obligations incurred in the Ordinary Course of Business since the Most Recent Fiscal Year End (none of which results from, arises out of, relates to, is in the nature of or was caused by any breach of contract, breach of warranty, tort, infringement or violation of law);

(c) liabilities or obligations under this Agreement;

(d) performance obligations under contracts required in accordance with their terms, or performance obligations to the extent required under applicable Laws, in each case to the extent arising after the date hereof; and

(e) liabilities or obligations that would not, individually or in the aggregate, reasonably be expected to result in a Material Adverse Effect on Company.

3.13. Antitakeover Statutes. No anti-takeover or similar statute or regulation under Michigan Law applies to any of the transactions contemplated by this Agreement. No other "control share acquisition," "fair price," "moratorium" or other similar anti-takeover laws or regulations enacted under Michigan Law or under any other laws of the State of Michigan apply to this Agreement or any of the transactions contemplated hereby.

3.14. Tax Matters.

(a) Each of Company and its Subsidiaries has filed all Tax Returns that it was required to file, except, in the case of Tax Returns other than federal or state income, sales and use Tax Returns, where failure to file such Return would not reasonably be expected to have a Material Adverse Effect on Company. All such Tax Returns were correct and complete in all material respects. All Taxes owed by any of Company and its Subsidiaries (whether or not shown on any Tax Return) have been paid, except where failure to pay such Taxes would not reasonably be expected to have a Material Adverse Effect on Company. None of Company and its Subsidiaries currently is the beneficiary of any extension of time within which to file any Tax Return. No claim has ever been made by a Governmental Entity in a jurisdiction where any of Company and its Subsidiaries does not file Tax Returns that it is or may be subject to taxation by that jurisdiction. There are no Security Interests on any of the assets of any of Company and its Subsidiaries that arose in connection with any failure (or alleged failure) to pay any Tax.

(b) Each of Company and its Subsidiaries has, in all material respects, withheld and paid all Taxes required to have been withheld and paid in connection with amounts paid, or owing to any employee, independent contractor, creditor, stockholder or other third party.

(c) There is no dispute or claim concerning any Liability for any Tax of any of Company and its Subsidiaries either (A) claimed or raised by any Governmental Entity in writing received by Company or any of its Subsidiaries or (B) as to which any of the directors and officers (and employees responsible for Tax matters) of Company and its Subsidiaries has Knowledge based on personal contact with any agent of such Governmental Entity. Section 3.14 of the Company Disclosure Letter lists all federal, state and foreign income Tax Returns filed with respect to any of Company and its Subsidiaries for taxable periods ended on or after December 31, 2002, indicates those Tax Returns that have been audited and indicates those Tax Returns that currently are being audited. Company has delivered to Parent correct and complete copies of all federal income Tax Returns, examination reports and statements of deficiencies assessed against or agreed to by any of Company and its Subsidiaries since December 31, 2002.

(d) None of Company and its Subsidiaries has waived any statute of limitations in respect of Taxes or agreed to any extension of time with respect to a Tax assessment or deficiency.

(e) None of Company and its Subsidiaries has made any payments, is obligated to make any payments or is a party to any agreement that under certain circumstances could obligate it to make any payments that will not be deductible under Code Section 280G. None of Company and its Subsidiaries has been a United States real property holding corporation within the meaning of Code Section 897(c)(2) during the applicable period specified in Code Section 897(c)(1)(A)(ii). Each of Company and its Subsidiaries has disclosed on its federal income Tax Returns all positions taken therein that could give rise to a substantial understatement of federal income Tax within the meaning of Code Section 6662. None of Company and its Subsidiaries is a party to any Tax allocation or sharing agreement. None of Company and its Subsidiaries (A) has been a member of an Affiliated Group filing a consolidated federal income Tax Return (other than a group the common parent of which was Company) or (B) has any Liability for the Taxes of any Person (other than any of Company and its Subsidiaries) under Treas. Reg. Section 1.1502-6 (or any similar provision of state, local or foreign law), as a transferee or successor, by contract or otherwise.

(f) Section 3.14 of the Company Disclosure Letter sets forth the following information with respect to each of Company and its Subsidiaries (or, in the case of clause (B) below, with respect to each of the Subsidiaries) as of the most recent practicable date: (A) the amount of any net operating loss, net capital loss, unused investment or other credit, unused foreign tax or excess charitable contribution allocable to Company or Subsidiary; and (B) the amount of any deferred gain or loss allocable to Company or Subsidiary arising out of any Deferred Intercompany Transaction. Promptly following the execution of this Agreement, Company shall update Section 3.14 of the Company Disclosure Letter to add (C) the basis of Company or Subsidiary in its assets and (D) the basis of the stockholder(s) of the Subsidiary in its stock (or the amount of any Excess Loss Account).

3.15. Real Property.

(a) Neither Company nor any of its Subsidiaries owns any real property.

(b) Section 3.15 of the Company Disclosure Letter lists and describes briefly all real property leased or subleased to any of Company and its Subsidiaries. Company has delivered to Parent correct and complete copies of the leases and subleases with respect to the real property listed in Section 3.15 of the Company Disclosure Letter. With respect to each lease and sublease listed in Section 3.15 of the Company Disclosure Letter. With respect to each lease and sublease listed in Section 3.15 of the Company Disclosure Letter and, other than with respect to the Main Leases, except as would not reasonably be expected to have a Material Adverse Effect on Company:

(1) the lease or sublease is legal, valid, binding, enforceable and in full force and effect;

(2) the lease or sublease will continue to be legal, valid, binding, enforceable and in full force and effect on identical terms following the consummation of the transactions contemplated hereby;

(3) no party to the lease or sublease is in breach or default, and no event has occurred that, with notice or lapse of time, would constitute a breach or default or permit termination, modification or acceleration thereunder, except for any such breach or default as would not reasonably be expected to have a Material Adverse Effect on Company;

(4) no party to the lease or sublease has repudiated any provision thereof;

(5) there are no disputes, oral agreements or forbearance programs in effect as to the lease or sublease;

(6) with respect to each sublease, to the Knowledge of Company, the representations and warranties set forth in subsections (1) through (5) above are true and correct with respect to the underlying lease;

(7) none of Company and its Subsidiaries has assigned, transferred, conveyed, mortgaged, deeded in trust or encumbered any interest in the leasehold or subleasehold;

(8) all facilities leased or subleased thereunder have received all approvals of Governmental Entities (including Licenses) required in connection with the operation thereof required to be obtained by Company or any of its Subsidiaries and have been operated and maintained by Company and its Subsidiaries in accordance with all Laws, except in each case (including the Main Leases) where failure to receive such approval or so operate and maintain would not reasonably be expected to have a Material Adverse Effect on Company;

(9) all facilities leased or subleased thereunder are supplied with utilities and other services necessary for the operation of said facilities; and

(10) to the Knowledge of Company, there are no restrictions that impair the current use or occupancy of the property that is subject to the lease.

3.16. Intellectual Property.

(a) Set forth in Section 3.16 of the Company Disclosure Letter is a complete and correct list of all material patents, patent applications, and all registrations or applications for registration of trademarks, servicemarks, copyrights and mask works owned or used by Company or its Subsidiaries. With respect to all Intellectual Property owned or used by Company or its Subsidiaries, except as would not reasonably be expected to have a Material Adverse Effect on Company, (A) Company and/or its Subsidiaries own or have the right to use all of such Intellectual Property free and clear of any Security Interest, license or other restriction, other than commercial, non-exclusive end-user licenses having a total consideration, with respect to each license, of less than \$50,000; (B) no proceedings have been instituted, are pending or, to the Knowledge of Company, are threatened in writing that challenge the rights of Company and/or its Subsidiaries, in respect of such Intellectual Property or the validity thereof and, to the Knowledge of Company, there is no basis for any such proceedings; (C) none of such Intellectual Property violates any Laws, or has at any time infringed on or violated any rights of others, or is being infringed by others; and (D) none of such Intellectual Property is subject to any outstanding Governmental Order except for rulings generated in the ordinary course of ex parte prosecution of applications for patents or for registration of trademarks, servicemarks, copyrights or mask works.

(b) Company and its Subsidiaries own or have the right to use pursuant to license, sublicense, agreement or permission all Intellectual Property necessary for the operation of the businesses of Company and its Subsidiaries as presently conducted. Each item of Intellectual Property owned or used by any of Company and its Subsidiaries immediately prior to the Closing hereunder will be owned or available for use by the Surviving Corporation or Company's Subsidiary, on identical terms and conditions immediately subsequent to the Closing hereunder, except as such non-ownership or unavailability or change in terms would not, individually or in the aggregate, reasonably be expected to have a Material Adverse Effect on Company. Each of Company and its Subsidiaries has taken all necessary action to maintain and protect each item of Intellectual Property that it owns or uses, except where failure to take such action would not, individually or in the aggregate, reasonably be expected to have a Material Adverse Effect on Company and its Subsidiaries has taken of Company and its Subsidiaries has any obligation to indemnify any Person for or against any interference, infringement, misappropriation or other conflict with respect to any item included in such Intellectual Property owned by Company or any of its Subsidiaries.

(c) Except where failure so to operate and perform would not, individually or in the aggregate, reasonably be expected to have a Material Adverse Effect on Company, the IT Assets of Company and its Subsidiaries operate and perform in accordance with their documentation and functional specifications and otherwise as required by Company and its Subsidiaries for the operation of their respective businesses, and, except as the result of such malfunction or failure did not have a Material Adverse Effect on Company at the time, have not malfunctioned or failed within the three (3) year period immediately preceding the date of this Agreement. To the Knowledge of Company, no Person has gained unauthorized access to such IT Assets. Company and its Subsidiaries have implemented and maintained for the three (3) year period immediately preceding the date of this Agreement reasonable and sufficient backup and disaster recovery technology consistent with industry practices.

26

**

As used in this Agreement,

۰.

(1) "<u>Computer Software</u>" means all computer software and databases (including source code, object code, and all related documentation).

(2) "*IT Assets*" means computers, Computer Software, firmware, middleware, servers, workstations, routers, hubs, switches, data communications lines, and all other information technology equipment and elements, and all associated documentation.

3.17. Tangible Assets. Company and its Subsidiaries own or lease all buildings, machinery, equipment and other tangible assets necessary for the conduct of their businesses as presently conducted. Except as would not, individually or in the aggregate, reasonably be expected to have a Material Adverse Effect on Company, each such tangible asset is free from material defects (patent and latent), has been maintained in accordance with normal industry practice, is in good operating condition and repair (subject to normal wear and tear), and is suitable for the purposes for which it presently is used.

3.18. Contracts. Section 3.18 of the Company Disclosure Letter lists the following contracts and other agreements to which any of Company and its Subsidiaries is a party or by which its assets are bound, in each case as of the date of this Agreement (each contract or other agreement required to be so listed, a "Material Contract"):

(a) any agreement (or group of related agreements with the same other party or Affiliates of such other party) for the lease of personal property to or from any Person providing for annual lease payments in excess of \$50,000 per annum;

(b) any agreement (or group of related agreements with the same other party or Affiliates of such other party) for the purchase of raw materials, commodities, supplies, products or other personal property, or for the receipt of services, the performance of which will extend over a period of more than one year or involve annual consideration in excess of \$250,000;

(c) any agreement (or group of related agreements with the same other party or Affiliates of such other party) for the sale of raw materials, commodities, supplies, products or other personal property, or for the furnishing of services, that involves annual consideration in excess of \$250,000

(d) any agreement concerning a partnership or joint venture;

(e) any agreement (or group of related agreements) under which it has created, incurred, assumed or guaranteed any indebtedness for borrowed money, or any capitalized lease obligation, in excess of \$50,000 or under which it has imposed a Security Interest on any of its assets, tangible or intangible;

(f) any agreement limiting the solicitation or hiring by Company or its Subsidiaries of employees or agents of, or the pursuit or consummation by Company or its Subsidiaries of investments in, or transactions with, any other Person or otherwise concerning competition or noncompetition by Company or its Subsidiaries;

(g) any agreement with any of the Stockholders and their Affiliates (other than Company and its Subsidiaries);

(h) any profit sharing, stock option, stock purchase, stock appreciation, deferred compensation, severance or other plan or arrangement for the benefit of its current or former directors, officers or employees;

(i) any collective bargaining agreement;

۰.

(j) any agreement for the employment of any individual on a full-time, part-time, consulting or other basis providing annual compensation in excess of \$250,000 or providing severance benefits;

(k) any agreement under which Company or its Subsidiaries (x) has made any loan or advance to any of its directors, officers or stockholders, or (y) has made any loans or advances to any of its employees that were in the aggregate as to an employee in excess of \$10,000 at any one time outstanding;

(1) any agreement under which Company or its Subsidiaries would be required, by its terms, to pay in excess of \$250,000 to terminate the agreement; and

(m) any other agreement (or group of related agreements with the same other party or Affiliates of such other party) the performance of which involves annual consideration in excess of \$250,000.

Company has delivered to Parent a correct and complete copy of each written agreement listed in Section 3.18 of the Company Disclosure Letter (as amended to date) and a written summary setting forth the terms and conditions of each oral agreement referred to in Section 3.18 of the Company Disclosure Letter. Except as the failure of this representation and warranty to be true and correct, individually or in the aggregate as to all such agreements, would not reasonably be expected to have a Material Adverse Effect, with respect to each such agreement, to Company's Knowledge: (A) the agreement is legal, valid, binding, enforceable and in full force and effect; (B) the agreement will continue to be legal, valid, binding, enforceable and in full force and effect on identical terms following the consummation of the transactions contemplated hereby; (C) no party is in material breach or material default, and no event has occurred that with notice or lapse of time would constitute a material breach or material default, or permit termination, modification or acceleration under the agreement; and (D) no party has repudiated any material provision of the agreement.

3.19. Note's and Accounts Receivable. Listed in Section 3.19 of the Company Disclosure Letter are notes or accounts receivable of Company or any of its Subsidiaries in excess of \$50,000.

3.20. Powers of Attorney. There are no outstanding powers of attorney executed on behalf of any of Company and its Subsidiaries.

3.21. Insurance. Section 3.21 of the Company Disclosure Letter sets forth the following information with respect to each current insurance policy (including policies providing property, casualty, liability and workers' compensation coverage and bond and surety arrangements) to which any of Company and its Subsidiaries is a party, a named insured or otherwise the beneficiary of coverage:

(a) the name, address and telephone number of the agent;

(b) the name of the insurer, the name of the policyholder, and the name of each covered insured;

(c) the policy number and the period of coverage;

(d) the scope (including an indication of whether the coverage was on a claims made, occurrence or other basis) and amount (including a description of how deductibles and ceilings are calculated and operate) of coverage; and

(e) a description of any retroactive premium adjustments or other loss-sharing arrangements.

With respect to each such insurance policy, to Company's Knowledge: (A) the policy is legal, valid, binding, enforceable and in full force and effect; (B) the Policy will continue to be legal, valid, binding, enforceable and in full force and effect on identical terms following the consummation of the transactions contemplated hereby; (C) neither any of Company and its Subsidiaries nor any other party to the policy is in breach or default (including with respect to the payment of premiums or the giving of notices), and no event has occurred that, with notice or the lapse of time, would constitute such a breach or default, or permit termination, modification or acceleration under the policy; and (D) no party to the policy has repudiated any provision thereof. Each of Company and its Subsidiaries has been covered during the past 5 years by insurance in scope and amount customary and reasonable for the businesses in which it has engaged during the aforementioned period. Neither Company nor any of its Subsidiaries has maintained any self-insurance arrangements during the past 5 years.

3.22. Litigation. Section 3.22 of the Company Disclosure Letter sets forth each instance in which any of Company and its Subsidiaries is, as of the date of this Agreement, (i) subject to any outstanding Governmental Order or (ii) a party or, to the Knowledge of Company, threatened in writing to be made a party to any action, suit, proceeding, hearing or investigation of, in or before, any Governmental Entity or quasi-judicial or administrative agency of any federal, state, local or foreign jurisdiction or before any arbitrator or mediator. There are no (i) civil, criminal or administrative actions, suits, claims, hearings, arbitrations, investigations or proceedings pending or, to Company's Knowledge, threatened in writing, or, to the Knowledge of Jerry Finefrock, threatened, against Company or any of its Subsidiaries or (ii) litigations, arbitrations, investigations or other proceedings, or Governmental Orders relating thereto, pending or, to Company's Knowledge, threatened in writing, or, to the Knowledge of Jerry Finefrock, threatened, against Company or any of its Subsidiaries before any Governmental Entity, including the FCC, except in the case of either clause (i) or (ii), for those that would not, individually or in the aggregate, reasonably be expected to result in a Material Adverse Effect on Company.

3.23. Employees. To the Knowledge of Company, there is no organizational effort presently being made or threatened by or on behalf of any labor union with respect to employees of any of Company and its Subsidiaries. None of Company and its Subsidiaries is a party to or bound by any collective bargaining agreement, nor has any of them experienced any strikes, grievances, claims of unfair labor practices or other collective bargaining disputes. None of Company and its Subsidiaries has committed any unfair labor practice, except any such as would not, individually or in the aggregate, reasonably be expected to result in a Material Adverse Effect on Company.

3.24. Employee Benefits.

(a) Section 3.24 of the Company Disclosure Letter lists each Employee Benefit Plan that any of Company and its Subsidiaries maintains or to which any of Company and its Subsidiaries contributes or has any obligation to contribute.

(1) To the Knowledge of Company, each such Employee Benefit Plan (and each related trust; insurance contract or fund) has been maintained, funded and administered in accordance with the terms of such Employee Benefit Plan and complies in form and in operation in all material respects with the applicable requirements of ERISA, the Code and other applicable laws.

(2) To the Knowledge of Company, all material required reports and descriptions (including annual reports (IRS Form 5500), summary annual reports, and summary plan descriptions) have been timely filed and/or distributed in accordance with the applicable requirements of ERISA and the Code with respect to each such Employee Benefit Plan. To the Knowledge of Company, the requirements of COBRA have been met in all material respects with respect to each such Employee Benefit Plan that is an Employee Welfare Benefit Plan subject to COBRA.

(3) All material contributions (including all employer contributions and employee salary reduction contributions) that are due have been made within the time period prescribed by ERISA to each such Employee Benefit Plan that is an Employee Pension Benefit Plan and all material contributions for any period ending on or before the Closing Date that are not yet due have been made to each such Employee Pension Benefit Plan or accrued in accordance with the past custom and practice of Company and its Subsidiaries. All premiums or other payments for all periods ending on or before the Closing Date that are due on or before the Closing Date have been paid with respect to each such Employee Benefit Plan that is an Employee Welfare Benefit Plan.

(4) Each such Employee Benefit Plan that is intended to meet the requirements of a "qualified plan" under Code Section 401(a) has received a determination from the Internal Revenue Service that such Employee Benefit Plan is so qualified, and to the Knowledge of Company nothing has occurred since the date of such determination that could adversely affect the qualified status of any such Employee Benefit Plan.

(5) There is no Employee Benefit Plan that is an Employee Pension Benefit Plan.

(6) Company has delivered to Parent correct and complete copies of the plan documents and summary plan descriptions, the most recent determination letter received from the Internal Revenue Service, the most recent annual report (IRS Form 5500, with all applicable attachments), and all related trust agreements, insurance contracts, and other funding arrangements that implement each such Employee Benefit Plan.

(b) With respect to each Employee Benefit Plan that any of Company, its Subsidiaries, and any ERISA Affiliate maintains or to which any of them contributes or has any obligation to contribute:

(1) To the Knowledge of Company, there have been no Prohibited Transactions with respect to any such Employee Benefit Plan. No Fiduciary has any material Liability for breach of fiduciary duty or any other failure to act or comply in connection with the administration or investment of the assets of any such Employee Benefit Plan. No action, suit, proceeding, hearing, or investigation with respect to the administration or the investment of the assets of any such Employee Benefit Plan (other than routine claims for benefits) is pending or to the Knowledge of Company is threatened. None of the directors and officers (and employees with responsibility for employee benefits matters) of Company and its Subsidiaries has any Knowledge of any basis for any such action, suit, proceeding, hearing or investigation.

(2) None of Company and its Subsidiaries has incurred any material Liability under COBRA with respect to any such Employee Benefit Plan that is an Employee Welfare Benefit Plan.

(c) None of Company, its Subsidiaries and any ERISA Affiliate contributes to, has any obligation to contribute to, or has any Liability (including withdrawal liability as defined in ERISA Section 4201) under or with respect to any Multiemployer Plan.

(d) Section 3.24(d) of the Company Disclosure Letter lists each Employee Welfare Benefit Plan that any of Company and its Subsidiaries maintains, to which any of Company and its Subsidiaries contributes or has any obligation to contribute, and describes any Liability or potential Liability that may be incurred by or imposed on Company or any of its Subsidiaries with respect to medical, health or life insurance or other welfare-type benefits for current or future retired or terminated employees, their spouses or their dependents (other than in accordance with COBRA).

3.25. Guaranties. None of Company and its Subsidiaries is a guarantor or otherwise is liable for any Liability or obligation (including indebtedness) of any other Person.

3.26. Environmental, Health and Safety Matters.

(a) Except as would not reasonably be expected to have a Material Adverse Effect on Company, the properties and facilities currently occupied by Company and its Subsidiaries are not being used by Company or its Subsidiaries to make, store, handle, treat, dispose, generate, or transport hazardous substances in violation of any Environmental, Health and Safety Requirement.

(b) To the Knowledge of Company, hazardous substances have never been made, stored, handled, treated, disposed of, generated, or transported on or from the properties and facilities occupied by Company and its Subsidiaries during the term of such occupancy, except in accordance with Environmental, Health and Safety Requirements and except as would not reasonably be expected to have a Material Adverse Effect on Company.

(c) The properties, facilities and operations of Company and its Subsidiaries and their respective predecessors and Affiliates have complied and are in compliance in all material respects with all applicable Environmental, Health and Safety Requirements. Without limiting the generality of the foregoing, each of Company, its Subsidiaries and their respective Affiliates has obtained and complied with, and is in compliance with, all permits, licenses and other authorizations that are required pursuant to Environmental, Health and Safety Requirements for the occupation of its facilities and the operation of its business, except for those the failure of which to obtain or comply or be in compliance with would not reasonably be expected to result in a Material Adverse Effect on Company, a list of all such permits, licenses and other authorizations 3.26 of the Company Disclosure Letter.

(d) To the Knowledge of Company, none of the properties, facilities or operations of Company and its Subsidiaries is subject to any judicial or administrative proceedings alleging the violation of any applicable Environmental, Health and Safety Requirements.

(e) To the Knowledge of Company, none of the properties, facilities or operations of Company and its Subsidiaries is the subject of federal, state or local investigation evaluating whether any remedial action is needed to respond to a release of any hazardous or toxic waste, substance or constituent, any petroleum or petroleum product or any other hazardous, illegal or unlawful substance into the environment.

ARTICLE IV REPRESENTATIONS AND WARRANTIES OF PARENT AND MERGER SUB

Except as set forth in the disclosure letter (subject to Section 10.13(c)) delivered to Company by Parent at or prior to entering into this Agreement (the "*Parent Disclosure Letter*"), or, to the extent that the qualifying nature of such disclosure with respect to a specific representation and warranty is readily apparent therefrom, as set forth in the Parent SEC Documents filed prior to the date hereof, Parent hereby represents and warrants to Company that:

4.1. Organization. Each of Parent and Merger Sub is a corporation duly organized, validly existing and in good standing under the Laws of the jurisdiction of its incorporation and has all requisite corporate or similar power and authority to own and operate its properties and assets and to carry on its business as presently conducted and is qualified to do business and is in good standing as a foreign corporation in each jurisdiction where the ownership or operation of its assets or properties or conduct of its business requires such qualification, except where the failure to be so organized, validly existing, qualified or in good standing, or to have such power or authority, would not be reasonably expected to have a Material Adverse Effect on Parent. Neither of Parent or Merger Sub is in default under or in violation of any provision of its charter or bylaws. The respective forms of Parent's Amended and Restated Certificate of Incorporation and Bylaws, as incorporated by reference in Parent's Annual Report on Form 10–K for the year ended December 31, 2004 filed with the SEC are correct and complete copies of Parent's charter and bylaws as amended to the date of this Agreement.

4.2. Capitalization. The entire authorized capital stock of Parent consists of 105,000,000 Parent Shares, of which 100,000,000 shares are designated as Parent Stock and 5,000,000 shares are designated as preferred stock. As of March 31, 2005, 27,212,160 shares of Parent Stock were issued and outstanding, 1,315,789 shares of Parent Stock were held in treasury and no shares of the authorized Parent preferred stock were issued or outstanding. Other than options for 4,686,628 shares of Parent Stock, warrants for 490,472 shares of Parent Stock, the rights under Parent's Rights Agreement, dated as of August 19, 1999, as amended, and a maximum of 655,135 additional shares of Parent Stock (subject to increase upon the expiration, other termination or surrender, unexercised, of any options referenced above issued pursuant to such plans) authorized and available for grant and issuance pursuant to employee or director benefit plans, there were, as of March 31, 2005, no outstanding or authorized options, warrants, purchase rights, subscription rights, conversion rights, exchange rights, or other contracts or commitments as of such date that could require Parent to issue, sell or otherwise cause to become outstanding any of its capital stock. The entire authorized capital stock of Merger Sub consists of 60,000 shares, no par value, all of one class designated as common, of which 100 shares are issued and outstanding.

4.3. Authorization of Transaction. Each of Parent and Merger Sub has all requisite corporate power and authority to, and has taken all corporate action necessary in order to, execute, deliver and perform its obligations under this Agreement. This Agreement is a valid and binding agreement of each of Parent and Merger Sub, enforceable against it in accordance with its terms, subject to bankruptcy, insolvency, fraudulent transfer, reorganization, moratorium and similar laws of general applicability relating to or affecting creditors' rights and to general equity principles. The Subsidiary of Parent executing the Switch Lease has all requisite corporate power and authority to, and has taken all corporate action necessary in order to, execute, deliver and perform its obligations under the Switch Lease, and the Switch Lease, when executed by such Subsidiary, will be a valid and binding agreement of such Subsidiary, enforceable against it in accordance with its terms, subject to bankruptcy, insolvency, fraudulent transfer, reorganization, moratorium and similar laws of general applicability relating to or affecting creditors' not execute, deliver and perform its obligations under the Switch Lease, and the Switch Lease, when executed by such Subsidiary, will be a valid and binding agreement of such Subsidiary, enforceable against it in accordance with its terms, subject to bankruptcy, insolvency, fraudulent transfer, reorganization, moratorium and similar laws of general applicability relating to or affecting creditors' rights and to general equity principles and except as indemnification obligations hereunder may be limited by applicable securities laws.

(f) Neither Company nor its Subsidiaries has filed any notice under any Environmental, Health and Safety Requirements indicating past or present treatment or disposal of a hazardous waste, hazardous substance or any petroleum or petroleum product, or reporting a spill or release of a hazardous or toxic waste, substance or constituent, any petroleum or petroleum product or any other substance into the environment.

۰.

(g) None of Company and its Subsidiaries have within the past year received written notice nor are they aware of any Liability of any of Company and its Subsidiaries in connection with any release of any hazardous or toxic waste, substance or constituent, any petroleum or petroleum product or any other substance into the environment.

3.27. Certain Business Relationships with Company and its Subsidiaries. None of the Stockholders and their Affiliates has been involved in any business arrangement or relationship with any of Company and its Subsidiaries within the past 12 months, and none of the Company's employees, the Stockholders and their respective Affiliates owns any asset, tangible or intangible, that is used in the business of any of Company and its Subsidiaries.

3.28. Accounts; Lockboxes; Safe Deposit Boxes. Section 3.28 of the Company Disclosure Letter contains a true and complete list of (i) the names of each bank, savings and loan association, securities or commodities broker or other financial institution in which any of Company and its Subsidiaries has an account, including cash contribution accounts, and the names of all persons authorized to draw thereon or have access thereto and (ii) the location of all lockboxes and safe deposit boxes of Company or its Subsidiaries and the names of all persons authorized to draw thereon or have access thereto. The Stockholders and their Affiliates have not commingled monies or accounts of Company or its Subsidiaries with other monies or accounts of the Stockholders and their Affiliates or relating to their other businesses nor have the Stockholders or their Affiliates transferred monies or accounts of Company or its Subsidiaries. At the Effective Time, all monies and accounts of Company and its Subsidiaries shall be held by, and be accessible only to, Company or its Subsidiaries.

3.29. Securities. To the Knowledge of Company, the outstanding shares of Company were issued in accordance with the registration or qualification provisions of the Securities Act and any relevant state securities laws or pursuant to valid exemptions therefrom.

3.30. Accounting Matters. Listed in Section 3.30 of the Company Disclosure Letter are all predecessor companies of Company, the names of any Persons from which, since January 1, 1999, Company previously acquired material properties or assets in a single transaction or series of related transactions with a single party in excess of \$2,000,000, and the changes in Company's capital structure and capital stock ownership since October 1, 2003.

3.31. Disclosure. The representations and warranties contained in this ARTICLE III do not contain any untrue statement of a material fact or omit to state any material fact necessary to make the statements contained in this ARTICLE III, in the light of the circumstances under which they are made, not misleading.

4.4. Noncontravention. Neither the execution and the delivery of this Agreement, nor the consummation of the transactions contemplated hereby, will (i) violate any provision of the charter or bylaws of any of Parent and its Subsidiaries or (ii), assuming compliance with the matters referred to in the next sentence of this Section 4.4, (A) violate any Laws or Governmental Order to which any of Parent and its Subsidiaries is subject or (B) with or without notice, lapse of time or both, conflict with, result in a breach of, constitute a default under, result in the acceleration of, create in any party the right to accelerate, terminate, modify or cancel, or require any notice under any agreement, contract, lease, license, instrument or other arrangement to which any of Parent and its Subsidiaries is a party or by which it is bound or to which any of its assets is subject (or result in the imposition of any Security Interest upon any of its assets), except, in the case of clause (ii), such violation, breach, default, acceleration or other change that would not, individually or in the aggregate, reasonably be expected to have a Material Adverse Effect on Parent. Except for (a) any FCC Consent, State PUC Consent or consent or approval of any other Governmental Entity identified in Section 3.3 of the Company Disclosure Letter or in Section 4.4 of the Parent Disclosure Letter, in each case as required by applicable Laws, (b) the filing of the Certificate of Merger with the Michigan Department of Labor & Economic Growth, Bureau of Commercial Services pursuant to the Michigan Law and of appropriate documents with relevant authorities of other states in which Company is qualified to do business to reflect such Certificate of Merger filing, (c) filings by Parent under, and compliance by Parent with the requirements under, the Securities Exchange Act, the Securities Act, applicable state securities laws and the applicable requirements of Nasdaq and (d) any other third party approvals as are reflected in Section 4.4 of the Parent Disclosure Letter, the execution, delivery and performance by Parent and Merger Sub of this Agreement and the transactions contemplated hereby do not require any consents, waivers, authorizations or approvals of, or filings with, any Governmental Entity or any other third Person except for those that the failure to make or obtain would not be reasonably expected to have a Material Adverse Effect on Parent.

4.5. Brokers' Fees. Except for fees payable to Green Mountain Finance LLC, neither Parent nor Merger Sub has any Liability or obligation to pay any fees or commissions to any broker, finder, or agent with respect to the transactions contemplated by this Agreement for which any of Company and its Subsidiaries could become liable or obligated.

4.6. Company Reports.

(a) A true and complete copy of each annual, quarterly and other report, registration statement and definitive proxy statement filed by Parent with the SEC since January 1, 2004 (the forms, statements, reports and documents filed by Parent with the SEC since January 1, 2004 and those filed with the SEC subsequent to the date of this Agreement, if any, including any amendments thereto, the "*Parent SEC Documents*") is available on the web site maintained by the SEC at http://www.sec.gov. As of their respective filing dates, the Parent SEC Documents complied or will comply in all material respects with the applicable requirements of the Securities Exchange Act and the rules and regulations of the SEC promulgated thereunder in effect on such dates and applicable to such Parent SEC Documents, except in each case to the extent superseded or amended by a Parent SEC Document filed prior to the date hereof. As of their respective dates, none of the Parent SEC Documents contained, and none of the Parent SEC Documents filed with the SEC subsequent to the date hereof will contain, any untrue statement of a material fact or omitted to state a material fact required to be stated therein or necessary to make the statements therein, in light of the circumstances under which they were made, not misleading, except to the extent superseded or amended by Parent SEC Documents filed subsequently and prior to the date hereof. The Parent SEC Documents included or will include all certificates required to be included therein pursuant to Sections 302 and 906 of the Sarbanes–Oxley Act of 2002, as amended (the "SOX Act"), and the internal control report and attestation of Parent's outside auditors required by Section 404 of the SOX Act.

(b) Management of Parent has established and maintains disclosure controls and procedures (as such term is defined in Rule 13a-15(e) under the Securities Exchange Act). Such disclosure controls and procedures: (i) are designed to ensure that material information relating to Parent and its Subsidiaries is made known to Parent's chief executive officer and its chief financial officer by others within those entities, particularly during the periods in which Parent's reports and filings under the Securities Exchange Act are being prepared, (ii) have been evaluated for effectiveness as of the end of the most recent annual period reported to the SEC, and (iii) except as may be indicated in the Parent SEC Documents, are effective to perform the functions for which they were established. Neither Parent's auditors nor the Audit Committee of the Parent Board have been advised of: (x) except as may be indicated in the Parent SEC Documents, are effective to securities Exchange Act) of the internal controls over financial reporting (as such term is defined in Rule 13a-15(f) under the Securities Exchange Act) of Parent that have materially affected Parent's internal control over financial reporting; or (y) any fraud, whether or not material, that involves management or other employees who have a role in the internal controls over financial reporting of Parent.

(c) Since December 31, 2003, neither Parent, any of its Subsidiaries, nor any director, officer, employee, auditor, accountant or representative of Parent or its Subsidiaries, has received or otherwise had or obtained knowledge of any complaint, allegation, assertion or claim, whether written or oral, regarding the accounting or auditing practices, procedures, methodologies or methods of either of Parent, its Subsidiaries or its internal accounting controls, including any material complaint, allegation, assertion or claim that Parent or its Subsidiaries has engaged in questionable accounting or auditing practices, and (ii) no attorney representing Parent or its Subsidiaries, whether or not employed by Parent or its Subsidiaries, has reported evidence of a violation of securities laws, breach of fiduciary duty or similar violation by Parent or its Subsidiaries or any of their officers, directors, employees or agents to the board of directors of Parent or its Subsidiaries or any committee thereof or to any director or officer of Parent or its Subsidiaries.

(d) The consolidated financial statements of Parent included in the Parent SEC Documents, in each case as they may have been amended or superseded by Parent SEC Documents filed subsequently and prior to the date hereof, complied or, in the case of Parent SEC Documents filed with the SEC subsequent to the date hereof, will comply as to form in all material respects with the published rules and regulations of the SEC with respect thereto, have been prepared or, in the case of Parent SEC Documents filed with the SEC subsequent to the date hereof, will be prepared in accordance with GAAP applied on a consistent basis throughout the periods indicated and consistent with each other (except that unaudited financial statements do not contain footnotes and other presentation items that may be required by GAAP) and fairly present or, in the case of Parent SEC Documents filed with the SEC subsequent to the date hereof, will fairly present Parent's consolidated financial condition, operating results and cash flows as of the dates and during the periods indicated therein, subject in the case of the unaudited financial statements to normal year end adjustments that will not be material in amount or effect.

4.7. Authorization for Parent Shares; Capital Resources.

(a) Parent will take all necessary action prior to the Closing Date to permit it to issue the number of Parent Shares required to be issued in the Merger pursuant to this Agreement.

(b) All of the Parent Shares to be issued in the Merger have been duly authorized and, upon consummation of the Merger and issuance thereof in exchange for Company Shares in accordance with the terms of this Agreement, will be validly issued, fully paid and nonassessable, and no Person will have any preemptive right of subscription or purchase in respect thereof.

(c) All Parent Shares issued pursuant to this Agreement will, when issued, be registered or exempt from registration under the Securities Act and the Securities Exchange Act and registered or exempt from registration under any applicable state securities laws.

(d) Parent has sufficient capital resources to pay the Cash Component of the Merger Consideration.

4.8. NASDAQ Compliance. Parent is in compliance with all applicable maintenance criteria and other requirements necessary to permit continued listing of the Parent Shares on the NASDAQ, and Parent has not received evidence to the contrary from the NASD.

4.9. No Material Adverse Changes. Since December 31, 2004, Parent and its Subsidiaries have conducted their businesses only in, and have not engaged in any material transaction other than in accordance with, the Ordinary Course of Business of Parent and its Subsidiaries. Since December 31, 2004 to the date hereof, there has not been any Material Adverse Effect on Parent.

4.10. Undisclosed Liabilities. There are no liabilities or obligations of Parent or any Subsidiary of Parent, whether or not accrued, contingent or otherwise and whether or not required to be disclosed, nor any other facts or circumstances that would reasonably be expected to result in any liabilities or obligations of Parent or any of its Subsidiaries, other than:

(a) liabilities or obligations to the extent (i) reflected on the consolidated balance sheet of Parent or (ii) readily apparent in the notes thereto, in each case included in Parent's annual report on Form 10-K for the year ended December 31, 2004;

(b) liabilities or obligations incurred in the Ordinary Course of Business since December 31, 2004 (none of which results from, arises out of, relates to, is in the nature of or was caused by any breach of contract, breach of warranty, tort, infringement or violation of law);

(c) liabilities or obligations under this Agreement;

(d) performance obligations under contracts required in accordance with their terms, or performance obligations to the extent required under applicable Laws, in each case to the extent arising after the date hereof; and

(e) liabilities or obligations that would not, individually or in the aggregate, reasonably be expected to result in a Material Adverse Effect on Parent.

4.11. Compliance with Laws. The businesses of Parent and each of its Subsidiaries have not been conducted in violation of any Laws, except for violations that would not, individually or in the aggregate, reasonably be expected to result in a Material Adverse Effect on Parent. No investigation or review by any Governmental Entity with respect to Parent or any of its Subsidiaries is pending or, to the Knowledge of Parent, threatened in writing, except for any such investigations or reviews that would not, individually or in the aggregate, reasonably be expected to result in a Material Adverse Effect on Parent. Each of Parent and its Subsidiaries has obtained and is in substantial compliance with all Licenses necessary to conduct its business as presently conducted, except those the absence of which or the noncompliance with which would not, individually or in the aggregate, reasonably be expected to result in a Material Adverse Effect on Parent. Each of Parent and its Subsidiaries has obtained and is in substantial compliance with all Licenses necessary to conduct its business as presently conducted, except those the absence of which or the noncompliance with which would not, individually or in the aggregate, reasonably be expected to result in a Material Adverse Effect on Parent.

4.12. Litigation. There is no action, suit, investigation or proceeding pending, or, to the Knowledge of Parent, threatened in writing, against Parent, any of its Subsidiaries, any present or former officer, director or employee of Parent or any of its Subsidiaries or any other Person for whom Parent or any such Subsidiary may be liable before any court or arbitrator or before or by any governmental body, agency or official, domestic or foreign, as to which there is a reasonable likelihood that such action, suit, investigation or proceeding will be resolved in a manner adverse to Parent or any of its Subsidiaries, except where such adverse resolution would not reasonably be expected to have a Material Adverse Effect on Parent.

4.13. No General Solicitation. Neither Parent nor, to the Knowledge of Parent, any Affiliate of Parent or any person acting on its or their behalf has engaged in any form of general solicitation or general advertising (within the meaning of Regulation D under the Securities Act) in connection with any offer or sale of the Parent Shares to be delivered pursuant to this Agreement.

ARTICLE V COVENANTS

5.1. Interim Operations. The Parties agree as follows with respect to the period from and after the execution of this Agreement.

(a) Company shall not knowingly take or permit any of its Subsidiaries to take any action or refrain from taking any action the result of which would be reasonably and foreseeably likely to prevent the consummation of the Merger by the Termination Date. Company covenants and agrees as to itself and its Subsidiaries that, after the date hereof and prior to the Effective Time, unless Parent shall otherwise approve in writing, and except as otherwise expressly contemplated by this Agreement or as required by applicable Laws, the business of it and its Subsidiaries shall be conducted in the ordinary and usual course and, to the extent consistent therewith, it and its Subsidiaries shall use their respective reasonable best efforts to preserve their business organizations intact and maintain existing relations and goodwill with Governmental Entities, customers, suppliers, distributors, creditors, lessors, employees and business associates and keep available the services of the present employees and agents of Company and its Subsidiaries, maintain the validity of the Communications Licenses and, except as disclosed in Section 5.1 of the Company Disclosure Letter, comply in all material respects with all requirements of the Communications Licenses and the rules and regulations of the FCC and State PUCs. Without limiting the generality of the foregoing and in furtherance thereof, from the date of this Agreement until the Effective Time, except (A) as otherwise expressly required by this Agreement or as permitted in Section 5.6(d), (B) as Parent may approve in writing, (C) as set forth in Section 5.1(a) of the Company Disclosure Letter and (D), in the case of any of the following clauses in this Section 5.1(a), as may be expressly permitted by another of the following clauses in this Section 5.1(a), the Company will not and will not permit its Subsidiaries to:

(1) except for the Articles Amendment, adopt or propose any change in its articles of incorporation or by-laws or other applicable governing instruments or amend any term of the Company Shares;

(2) merge or consolidate Company or any of its Subsidiaries with any other Person, except for any such transactions among wholly owned Subsidiaries of Company that are not obligors or guarantors of third-party indebtedness, or adopt a plan of liquidation;

(3) acquire assets outside of the Ordinary Course of Business from any other Person with a value or purchase price in excess of 50,000 in the aggregate, other than acquisitions pursuant to Contracts to the extent in effect immediately prior to the execution of this Agreement and as otherwise set forth in Section 5.1(a)(3) of the Company Disclosure Letter, and other than capital expenditures as permitted by Section 5.1(a)(12);

(4) (x) enter into any material line of business in any geographic area other than the current lines of business of Company or any of its Subsidiaries, and in the geographic areas where they are currently conducted, as of the date hereof or (y) engage in the conduct of any business in any state that would require the receipt or transfer of a Communications License;

(5) file for any License outside of the Ordinary Course of Business;

۰.

(6) other than as set forth in Section 5.1(a)(6) of the Company Disclosure Letter and other than the issuance of shares of Common Stock upon exercise of Employee Stock Options or Warrants or conversion of shares of Preferred Stock outstanding as of the date of this Agreement, issue, sell, pledge, dispose of, grant, transfer, lease, license, guarantee, encumber, or authorize the issuance, sale, pledge, disposition, grant, transfer, lease, license, guarantee or encumbrance of, any shares of capital stock of Company or any its Subsidiaries (other than the issuance of shares by a wholly owned Subsidiary of Company to Company or another wholly owned Subsidiary), or securities convertible or exchangeable into or exercisable for any shares of such capital stock, or any options, warrants, conversion rights, stock appreciation rights, redemption rights, repurchase rights, agreements, arrangements, calls, commitments or other rights of any kind to acquire any shares of such capital stock or such convertible or exchangeable securities;

(7) other than (i) in connection with receivables facilities and securitizations as in effect on the date hereof and disclosed in the Company Disclosure Letter and renewals thereof in the Ordinary Course of Business, (ii) in connection with the refinancing of Company's indebtedness under its credit facility as in effect on the date hereof and disclosed in the Company Disclosure Letter, (iii) Liens created or incurred to secure the purchase price of assets acquired as permitted by Section 5.1(a)12) and (iv) Liens described in clause (ii), (iii), (iv), (v) or (vi) of Section 3.8, create or incur any Lien on any assets of the Company or any of its Subsidiaries;

۰.

(8) other than loans and advances to employees of Company or its Subsidiaries in the Ordinary Course of Business and not in excess of \$10,000 at any time outstanding to any employee, make any loans, advances or capital contributions to or investments in any Person (other than Company or any direct or indirect wholly owned Subsidiary of Company);

(9) declare, set aside or pay any dividend or distribution with respect to Company's capital stock (whether in cash, stock or property or any combination thereof) or redeem, purchase or acquire any of its capital stock;

(10) reclassify, split, combine, subdivide or repurchase, redeem or otherwise acquire, directly or indirectly, any of its capital stock or securities convertible or exchangeable into or exercisable for any shares of its capital stock;

(11) other than (i) in connection with receivables facilities and securitizations as in effect on the date hereof and disclosed in the Company Disclosure Letter and renewals thereof in the Ordinary Course of Business, (ii) in connection with the refinancing of Company's indebtedness under its credit facility as in effect on the date hereof and disclosed in the Company Disclosure Letter, and (iii) indebtedness incurred to finance the capital expenditures permitted by Section 5.1(a)(12) and guarantees thereof, incur any indebtedness for borrowed money or guarantee such indebtedness of another Person, or issue or sell any debt securities or warrants or other rights to acquire any debt security of the Company or any of its Subsidiaries;

(12) except for the capital expenditures set forth in Section 5.1(a)(12) of the Company Disclosure Letter and asset acquisitions otherwise permitted by Section 5.1(a)(3) (without giving effect to the exception therein for capital expenditures as permitted by this clause (12)), make or authorize any capital expenditure;

(13) enter into any contract or other agreement (x) that would have been a Material Contract as described in Section 3.18 (d), (f) or (g) had it been entered into prior to the date of this Agreement, (y) other than in the Ordinary Course of Business, that involves annual consideration in excess of \$50,000 or (z) that involves annual consideration in excess of \$250,000 and is not terminable by Company and its Subsidiaries without additional payment or penalty (including by any acceleration of remaining amounts), upon not more than 90 days' notice;

(14) make any changes with respect to accounting policies or procedures, except as required by changes in GAAP or by applicable Laws or except as Company, based upon the advice of its independent auditors after consultation with Parent, determines in good faith is advisable to conform to best accounting practices;

(15) settle any litigation or other proceedings before or threatened to be brought before a Governmental Entity for an amount to be paid by Company or any of its Subsidiaries in excess of \$25,000 or that would be reasonably likely to have any adverse impact on the operations of Company or any of its Subsidiaries;

(16) other than in the Ordinary Course of Business, (i) amend or modify in any material respect adverse to Company or its Subsidiaries, or terminate or waive any material right or benefit of Company or its Subsidiaries under, any Material Contract, or (ii) cancel, modify or waive any debts or claims held by it or waive any rights;

(17) sell, lease, license or otherwise dispose of any assets of Company or its Subsidiaries except (i) in the Ordinary Course of Business or obsolete assets or (ii) as set forth in Section 5.1(a)(17) of the Company Disclosure Letter;

(18) except as (x) required pursuant to existing written, binding agreements in effect prior to the date of this Agreement or as otherwise required by applicable Laws, (y) set forth in Section 5.1(a)(18) of the Company Disclosure Letter or (z) the costs and expenses of which will be a Transaction Expense or Severance Amount, (i) enter into any commitment to provide any severance or termination benefits to (or amend any existing arrangement with) any director, officer or employee of Company or any of its Subsidiaries, other than for severance or termination benefits to employees (other than officers) in the Ordinary Course of Business consistent with past practice and pursuant to the terms of plans, programs or arrangements in effect prior to the date of this Agreement and disclosed on Section 3.18 or 3.25 of the Company Disclosure Letter, (ii) increase the benefits payable under any existing severance or termination benefit policy or employment agreement (other than as required to be increased pursuant to the existing terms of any such policy or agreement or as a result of ordinary pay raises or promotions), (iii) enter into any employment, severance, change in control, termination, deferred compensation or other similar agreement (or amend any such existing agreement) with any director, officer or employee of the Company or any of its Subsidiaries other than pursuant to the terms of any plan or agreement in effect on the date hereof and disclosed on Section 3.18 or 3.25 of the Company Disclosure Letter, (iv) establish, adopt, amend or terminate any employee or director compensation or other benefit, employment or severance plan, program or agreement (including Employee Benefit Plans, each, a "Compensation Plan"), except for technical amendments in the Ordinary Course of Business, provided that such amendments do not materially increase the cost of such arrangements to Company, (v) increase the compensation, bonus or other benefits of, make any new awards under any Compensation Plan to, or pay any bonus to any director, officer, employee, consultant or independent contractor of the Company or any of its Subsidiaries, except for increases, new awards or payments in the Ordinary Course of Business for employees who are not officers of Company, (vi) take any action to fund or in any other way secure the payment of compensation or benefits under any Compensation Plan, except as required pursuant to the terms thereof as in effect as of the date of this Agreement, (vii) take any action to accelerate the vesting or payment of any compensation or benefits under any Compensation Plan, to the extent not already required in any such Compensation Plan, or (viii) enter into any collective bargaining agreements; provided, however, that the prohibitions contained in the foregoing clauses (i) and (v) shall not apply in connection with newly hired or newly promoted employees, in each case to the extent consistent with past practice;

(19) take any action that may reasonably be expected to jeopardize the validity of any of the Communications Licenses or result in the revocation, surrender or any adverse modification of, forfeiture of, or fail to renew under regular terms, any of the Communications Licenses, (b) fail to prosecute with due diligence any pending applications with respect to the Communications Licenses, including any renewals thereof, and (c) with respect to Communications Licenses, fail to make all material filings and reports and pay all material fees necessary or reasonably appropriate for the continued operation of the Business, as and when such approvals, consents, permits, licenses, filings, or reports or other authorizations are necessary or appropriate or (d) fail to initiate appropriate steps to renew any material Licenses held by Company or any of its Subsidiaries that are scheduled to terminate prior to or within 60 days after the Effective Time or to prosecute any pending applications for any material License; or

(20) agree or commit to do any of the foregoing.

۰.

(b) Parent shall not knowingly take or permit any of its Subsidiaries to take any action or refrain from taking any action the result of which would be reasonably and foreseeably likely to prevent the consummation of the Merger by the Termination Date. Without limiting the generality of the foregoing, from the date of this Agreement until the Effective Time, except (A) as otherwise expressly required by this Agreement, (B) as Company may approve in writing or (C) as set forth in Section 5.1(b) of the Parent Disclosure Letter, Parent will not and will not permit its Subsidiaries to:

(1) adopt or propose any material change in Parent's certificate of incorporation or by laws or other applicable governing instruments or amend any term of the shares of Parent Stock;

(2) merge or consolidate Parent or Merger Sub with any other Person, sell all or substantially all of Parent's assets or adopt a plan of liquidation of Parent;

(3) enter into or acquire any new line of business that (i) is material to Parent and its Subsidiaries taken as a whole and (ii) is not strategically related to the current business or operations of Parent and its Subsidiaries;

(4) except for shares of Parent Stock issued for fair value in arm's length transactions and other than the issuance of shares in the ordinary course of business consistent with past practices pursuant to Parent employee benefit plans, issue, sell, pledge, dispose of, grant, transfer, lease, license, guarantee, encumber, or authorize the issuance, sale, pledge, disposition, grant, transfer, lease, license, guarantee of capital stock of Parent or any of its Subsidiaries (other than the issuance of shares by a wholly owned Subsidiary of Parent to Parent or another wholly owned Subsidiary of Parent), or securities convertible or exchangeable into or exercisable for any shares of such capital stock, or any options, warrants, conversion rights, stock appreciation rights, redemption rights, repurchase rights, agreements, arrangements, calls, commitments or other rights of any kind to acquire any shares of such capital stock or such convertible or exchangeable securities;

(5) declare, set aside or pay any dividend or distribution (whether in cash, stock or property or any combination thereof) on any shares of Parent Stock or on any shares of capital stock of any Subsidiary, other than by wholly owned Subsidiaries;

(6) reclassify, split, combine or subdivide, or repurchase, redeem or otherwise acquire at prices above fair market value, directly or indirectly, any of its capital stock or securities convertible or exchangeable into or exercisable for any shares of its capital stock; or

(7) agree or commit to do any of the foregoing.

5.2. Listing. Parent shall use commercially reasonable efforts to cause the Parent Shares to be issued in connection with the Merger to be approved for listing on Nasdaq, subject to official notice of issuance, prior to the Closing Date.

5.3. Filings; Other Actions; Notification.

(a) Company shall promptly after the date of this Agreement prepare and distribute to the Stockholders a notice of the Company Special Meeting and related materials.

(b) Company and Parent shall cooperate with each other and use (and shall cause their respective Subsidiaries to use) their respective reasonable best efforts to take or cause to be taken all actions, and do or cause to be done all things, necessary, proper or advisable on its part under this Agreement and applicable Laws to consummate and make effective the Merger and the other transactions contemplated by this Agreement as soon as practicable, including preparing and filing as promptly as practicable all documentation to effect all necessary notices, reports and other filings (including by filing no later than 5 business days after the date of this Agreement all applications required to be filed with the FCC; provided, however, that the failure to file within 5 business days will not constitute a breach of this Agreement) and to obtain as promptly as practicable all consents, registrations, approvals, permits and authorizations necessary or advisable to be obtained from any third party and/or any Governmental Entity in order to consummate the Merger or any of the other transactions contemplated by this Agreement; provided, however, that nothing in this Section 5.3 (i) shall require, or be construed to require, Parent to take or to refrain from taking any action, to agree to any restriction with respect to any assets or operations of Parent or its Subsidiaries, or to cause its Subsidiaries to do or agree to do any of the foregoing, in each case that would take effect prior to the Effective Time, or (ii) shall require, or be construed to require. Parent to take or to refrain from taking any action, to agree to any restriction with respect to any assets or operations of Parent or Company or its respective Subsidiaries, or to cause its Subsidiaries to do or agree to do any of the foregoing, if any such action, failure to act, restriction or agreement, individually or in the aggregate, would reasonably be expected to have a Material Adverse Effect on it or a Combined Material Adverse Effect. Subject to applicable Laws relating to the exchange of information, Parent and Company shall have the right to review in advance, and to the extent practicable each will consult the other on, all of the information relating to Parent or Company, as the case may be, and any of their respective Subsidiaries, that appears in any filing made with, or written materials submitted to, any third party and/or any Governmental Entity in connection with the Merger and the other transactions contemplated by this Agreement. To the extent permitted by law, each Party shall provide the other with copies of all correspondence between it (or its advisors) and any Governmental Entity relating to the transactions contemplated by this Agreement and, to the extent reasonably practicable, all telephone calls and meetings with a Governmental Entity regarding the transactions contemplated by this Agreement shall include representatives of Parent and Company. In exercising the foregoing rights, each of Company and Parent shall act reasonably and as promptly as practicable.

(c) To the extent permitted by law, Company and Parent each shall, upon request by the other, furnish the other with all information concerning itself, its Affiliates, directors, officers and stockholders and such other matters as may be reasonably necessary or advisable in connection with any statement, filing, notice or application made by or on behalf of Parent, Company or any of their respective Affiliates to any third party and/or any Governmental Entity in connection with the Merger and the transactions contemplated by this Agreement.

(d) Subject to applicable Laws and the instructions of any Governmental Entity, Company and Parent each shall keep the other apprised of the status of matters relating to completion of the transactions contemplated hereby, including promptly furnishing the other with copies of notices or other communications received by Parent or Company, as the case may be, or any of its Subsidiaries, from any third party and/or any Governmental Entity with respect to the Merger and the other transactions contemplated by this Agreement. Company shall give prompt notice to Parent of any change, fact or condition of which it has Knowledge that is reasonably expected to result in a Material Adverse Effect on Company or of any failure of any condition to Parent's obligations to effect the Merger. Parent shall give prompt notice to Company of any change, fact or condition of which it has Knowledge that is reasonably expected to result in a Material Adverse Effect on Parent or of any failure of any condition to Company's obligations to effect the Merger.

(e) Subject to the proviso set forth in Section 5.3(b), Parent's and Company's obligations under this Section 5.3 shall include, without limitation, the obligation to use their respective reasonable best efforts to defend any lawsuits or other legal proceedings, whether judicial or administrative, challenging the consummation of the Merger or the other transactions contemplated hereby, including using reasonable best efforts to seek to have any stay or other injunctive relief which would prevent or materially delay or impair the consummation of the transactions contemplated by this Agreement entered by any court or other Governmental Entity reversed on appeal or vacated.

5.4. Company Financial Statements. As soon as reasonably practicable, but in any event within 20 days after the end of each calendar month commencing with May 2005, Company will deliver to Parent unaudited consolidated balance sheets of Company and its Subsidiaries as of the end of such calendar month and as at the end of the comparative month in the preceding year, together with unaudited summaries of consolidated earnings of Company and its Subsidiaries for such such calendar month and for the comparative month in the preceding year. As soon as reasonably practicable, but in any event within 45 days after the end of each fiscal quarter of Company, commencing with the quarter ended March 31, 2005, Company will deliver to Parent unaudited consolidated balance sheets of Company and its Subsidiaries as at the end of such fiscal quarter and as at the end of the comparative fiscal quarter of the preceding year, together with the unaudited statements of consolidated income and cash flows for the fiscal quarters then ended.

5.5. Access. Each Party shall permit (and shall cause each of its Subsidiaries to permit) representatives of the other Parties to have reasonable access to all premises, properties, personnel, books, records (including Tax records), contracts and documents of or pertaining to each Party and their respective Subsidiaries, other than information or documentation relating to Acquisition proposals by third parties, except as provided in Section 5.8, and provided that no investigation pursuant to this Section 5.5 shall affect or be deemed to modify any representation or warranty made by any Party herein. Any investigation pursuant to this Section 5.5 shall be conducted in such manner as not to interfere unreasonably with the conduct of the business of the Parties and their Subsidiaries.

5.6. Director and Officer Liability. Parent shall cause the Surviving Corporation, and the Surviving Corporation hereby agrees, to do the following:

(a) From and for six years after the Effective Time, each of Parent and the Surviving Corporation shall, jointly and severally, indemnify and hold harmless each present and former officer and director of the Company and of any Subsidiary of the Company (each an "*Indemnified Person*") in respect of acts or omissions occurring at or prior to the Effective Time to the fullest extent permitted by Michigan Law or provided under the Company's articles of incorporation and bylaws in effect on the date hereof; provided that such indemnification shall be subject to any limitation imposed from time to time under applicable law. The Indemnified Person shall be entitled to control the defense of any action, suit, investigation or proceeding with counsel of his or her own choosing reasonably acceptable to the Surviving Corporation and the Surviving Corporation shall not be liable for the fees of more than one counsel for all Indemnified Persons, other than local counsel, in any one jurisdiction, unless a conflict of interest shall be caused thereby, and *provided further* that the Surviving Corporation shall not be liable for any settlement effected without its written consent (which consent shall not be unreasonably withheld).

(b) Any Indemnified Person wishing to claim indemnification under Section 5.6(a), upon learning of any such claim, action, suit, proceeding or investigation that may give rise to such claim, shall promptly notify Parent thereof, but the failure to so notify shall not relieve Parent or the Surviving Corporation of any liability it may have to such Indemnified Person except to the extent such failure materially and actually prejudices the indemnifying party. In the event of any such claim, action, suit, proceeding or investigation (whether arising before or after the Effective Time), (i) Parent or the Surviving Corporation shall have the right to assume the defense thereof and Parent shall not be liable to such Indemnified Persons for any legal expenses of other counsel or any other expenses subsequently incurred by such Indemnified Persons in connection with the defense thereof, except that if Parent or the Surviving Corporation does not elect to assume such defense or counsel for the Indemnified Persons advises that there are issues that raise conflicts of interest between Parent or the Surviving Corporation and the Indemnified Persons, the Indemnified Persons may retain counsel satisfactory to them, and Parent and the Surviving Corporation shall jointly and severally be obligated to pay all reasonable fees and expenses of such counsel for the Indemnified Persons promptly as statements therefor are received; provided, however, that Parent and the Surviving Corporation shall be obligated pursuant to this Section 5.6(b) to pay for only one firm of counsel for all Indemnified Persons in any jurisdiction unless the use of one counsel for such Indemnified Persons would present such counsel with a conflict of interest; provided, that the fewest number of counsel necessary to avoid conflicts of interest shall be used; (ii) the Indemnified Persons will use their reasonable efforts to cooperate in the defense of any such matter, and (iii) Parent and the Surviving Corporation shall not be liable for any settlement effected without their prior written consent (such consent not to be unreasonably withheld or delayed); and provided, further, that Parent and the Surviving Corporation shall not have any obligation under this Agreement to any Indemnified Person if and when a court of competent jurisdiction shall ultimately determine, and such determination shall have become final, that the indemnification of such Indemnified Person in the manner contemplated hereby is prohibited by applicable law.

(c)' If Parent, the Surviving Corporation or any of its successors or assigns (i) consolidates with or merges into any other Person and shall not be the continuing or surviving corporation or entity of such consolidation or merger, or (ii) transfers or conveys all or substantially all of its properties and assets to any Person, then, and in each such case, to the extent necessary, proper provision shall be made so that the successors and assigns of Parent or the Surviving Corporation, as the case may be, shall assume the obligations set forth in this Section 5.6.

(d) Company may purchase a six-year "*tail*" prepaid policy prior to the Effective Time on terms with respect to coverage and amount substantially comparable to, but no less favorable to Company and the Indemnified Persons than, those under the directors' and officers' liability policy maintained by Company and in effect on the date hereof, *provided* that the amount paid by Company for such "*tail*" policy shall not exceed \$90,000 unless the full amount of the excess is included as a Transaction Expense.

(e) The rights of each Indemnified Person under this Section 5.6 shall be in addition to any rights such Person may have under the articles of incorporation or bylaws of the Company or any of its Subsidiaries, or under Michigan Law or any other applicable laws. These rights shall survive consummation of the Merger and are intended to benefit, and shall be enforceable by, each Indemnified Person and may not be changed as to an Indemnified Person without the consent of the Indemnified Person.

(f) The provisions of this Section 5.6 are intended to be for the benefit of, and shall be enforceable by, each of the Indemnified Persons and their heirs and legal representatives.

5.7. Employee Benefits After the Merger.

(a) Until at least the earlier of the 120th day after the Effective Time and December 31, 2005, Parent shall cause the Surviving Corporation to provide those of its and its Subsidiaries' employees who were employed by Company or its Subsidiaries immediately prior to the Effective Time with compensation and employee benefits at least as favorable, in the aggregate, as the compensation and benefits provided by Company and its Subsidiaries to such employees immediately prior to the date hereof. The preceding sentence shall not preclude Parent or the Surviving Corporation at any time following the Effective Time from terminating the employment of any Company employee.

(b) Parent shall, or shall cause the Surviving Corporation to, give each Company employee full credit in respect of his or her employment with Company or its Subsidiaries prior to the Effective Time for purposes of eligibility, vesting, level of benefits and service under any new employee benefit plans offered by the Surviving Corporation after the Merger ("Surviving Corporation New Plans") or any Parent Employee Benefit Plan in which the Company employee is permitted to participate (to the extent that the corresponding Company Employee Benefit Plan currently provided to Company employees gave such credit).

(c) From and after the Effective Time, Parent will, or will cause the Surviving Corporation to, (i) cause any pre-existing conditions or limitations and eligibility waiting periods (only to the extent such limitations or waiting periods did not apply to the Company employees under the Company Employee Benefit Plans) under any group health plans of Parent, or any group health plans constituting Surviving Corporation New Plans, in which Company employees are permitted to participate to be waived with respect to the Company employees and their eligible dependents and (ii) give each Company employee credit for the plan year in which the Effective Time occurs toward applicable deductibles and annual out-of-pocket limits under group health plans of Parent or group health plans constituting Surviving Corporation New Plans for expenses incurred prior to the Effective Time under the Benefit Plans.

(d) From and after the Effective Time, Parent will, and will cause the Surviving Corporation to, honor, without modification, perform all acts and pay all amounts required or due under or with respect to each Company Benefit Plan and each agreement that relates to any current or former employee of the Company and its Subsidiaries or the terms of any such employee's employment or termination of employment, including, without limitation, all employment, retention, change of control, employment protection, severance, termination, consulting, deferred compensation, executive pension and retirement, welfare and fringe benefit agreements, plans and programs.

(e) Parent acknowledges and agrees that the consummation of the transactions contemplated by this Agreement will constitute a "change of control" of the Company for purposes of each Company Benefit Plan and each program, policy and agreement covering any current or former employee of the Company and its Subsidiaries as has been specifically indicated in Section 5.7(e) of the Company Disclosure Letter to have any such "change of control" provisions, and, accordingly agrees to, and agrees to cause the Surviving Corporation to, honor all provisions relating to a change of control under such Company Benefit Plans, programs, policies and agreements as have been specifically indicated in Section 5.7(e) of the Company Disclosure Letter to have any such "change of control" provisions.

(f) Notwithstanding the foregoing, nothing in this Section 5.7 shall preclude Parent from seeking to (i) modify any employment agreement with the consent of the affected employee or employees or (ii) modify any Company Benefit Plan to the extent such modification is permitted by the terms of such Company Benefit Plan and is consistent with Section 5.7(a).

5.8. Lockup Agreement; Transfer Restrictions. Each Preferred Stockholder, by its receipt of any Parent Shares as Merger Consideration and without any further action by or on behalf of such Preferred Stockholder, and in part consideration of the issuance of such Parent Shares to it, shall be deemed to have acknowledged and agreed, and acknowledges and agrees, with and for the benefit of Parent as follows, and all Parent Shares issued as Merger Consideration are subject, and all holders of such Parent Shares are bound by, to the following:

(a) Except only as provided in the last sentence of this Section 5.8(a), such Preferred Stockholder shall not, for the period from the Closing Date and ending on the first anniversary of the Closing Date, offer, sell, contract to sell, pledge, grant any option to purchase, make any short sale or otherwise dispose of any part of the Parent Shares issued to it in accordance with the terms of this Agreement (the "*Merger Shares*"), any options or warrants convertible or exercisable into such Merger Shares, or any other securities convertible into, exchangeable for or that represent the right to receive such Merger Shares (except pursuant to the Escrow Agreement). The foregoing restriction precludes such Preferred Stockholder from engaging in any hedging or other transaction that is designed to or that reasonably could be expected to lead to or result in a sale or disposition of such Merger Shares even if such Merger Shares would be disposed of by someone other than such Preferred Stockholder, and such prohibited hedging or other transactions would include any short sale or any purchase, sale or grant of any right (including any put or call option) with respect to any of such Merger Shares or with respect to any security that includes, relates to, or derives any part of its value from such Merger Shares. Notwithstanding the foregoing, but subject to any other provisions of this Section 5.8(a), such Preferred Stockholder may sell, pledge or otherwise dispose of up to an amount equal to 25% of such Merger Shares in any three-month period, commencing on the first day of the calendar month beginning after the Closing Date.

(b) Stop transfer instructions will be given to Parent's transfer agents with respect to the Merger Shares and there will be placed on the certificates for the Merger Shares, or any substitutions therefor, a legend stating in substance:

"The shares represented by this certificate are subject to and may only be transferred in accordance with the terms of an agreement dated May 23, 2005, between the registered holder hereof and Talk America Holdings, Inc., a copy of which agreement is on file at the principal offices of Talk America Holdings, Inc."

(c) The Merger Shares were offered and issued in a transaction not involving a public offering, have not been registered under the Securities Act and cannot be resold, pledged, assigned or otherwise disposed of except pursuant to an effective registration statement under the Securities Act or pursuant to an exemption from registration under the Securities Act, and in each case in accordance with any applicable securities laws of any state of the United States.

(d) Stop transfer instructions will be given to Parent's transfer agents with respect to the Merger Shares and there will be placed on the certificates for the Merger Shares, or any substitutions therefor, a legend stating in substance:

"The shares represented by this certificate have not been registered under the Securities Act of 1933 and such shares may not be sold or otherwise transferred except in accordance with an exemption from the registration requirements of the Securities Act of 1933 or pursuant to an effective registration statement under the Securities Act of 1933. The shares represented by this certificate may only be transferred in accordance with the terms of an agreement dated May 23, 2005 between the registered holder hereof and Talk America Holdings, Inc., a copy of which agreement is on file at the principal offices of Talk America Holdings, Inc."

(e) The legend set forth in Section 5.8(d) shall be removed from a certificate representing Merger Shares by delivery of substitute certificates without such legend if such Preferred Stockholder shall have delivered to Parent a copy of a letter from the staff of the Commission, or an opinion of counsel in form and substance reasonably satisfactory to Parent, to the effect that such legend is not required for purposes of the Securities Act or upon sale of the Merger Shares represented by such certificate pursuant to an effective registration statement under the Securities Act.

۰.

5.9. Notices and Filing by Company. As promptly as practicable, but in no event more than three business days, after the date hereof, Company shall give to all Stockholders the notices of the approval and adoption of this Agreement and the Merger and the approval of the Articles Amendment required to be given pursuant to Section 450.1407 of the Michigan Law and the Secretary of Company shall certify in writing to Parent that such notice has been given as so required.

(b) Not later than immediately prior to the Closing, Company shall execute and file with the Michigan Department of Labor & Economic Growth, Bureau of Commercial Services, a certificate of amendment (the "Certificate of Amendment") with respect to the Articles Amendment as provided in Section 450.1631 of the Michigan Law and shall give notice thereof to the Stockholders as and to the extent required by Section 450.1764(2) of the Michigan Law, and the Secretary of Company shall certify in writing to Parent that such notice has been given as so required.

ARTICLE VI CONDITIONS TO OBLIGATIONS TO CLOSE

6.1. Conditions to Each Party's Obligation. The respective obligation of each Party to consummate the Merger is subject to the satisfaction or waiver of each of the following conditions at or prior to the Effective Time:

(a) (i) All approvals and authorizations required to be obtained in respect of the FCC Licenses, the Michigan State Licenses and the Ohio State Licenses for the consummation of the Merger shall have been obtained; and (ii) all other Governmental Consents the failure of which to make or obtain would, individually or in the aggregate, provide a reasonable basis to conclude that Company or its directors or officers would be subject to risk of criminal liability or to risk of civil liability for which they are not entitled to indemnification by Company (or the Surviving Corporation), shall have been made or obtained. For purposes of this Agreement, "Governmental Consents" means all notices, reports, filings, consents, registrations, approvals, permits or authorizations required to be made prior to the Effective Time by Company or Parent or any of their respective Subsidiaries with, or obtained prior to the Effective Time by Company or Parent or any of their respective Subsidiaries from, any Governmental Entity in connection with the execution and delivery of this Agreement and the consummation of the Merger and other transactions contemplated hereby.

(b) Neither any Laws or Governmental Order shall be enacted, promulgated, entered, enforced or deemed applicable to the Merger nor any other action shall have been taken by any Governmental Entity that is in effect and that (i) restrains, enjoins or otherwise prohibits the consummation of the transactions contemplated by this Agreement or (ii) that makes the purchase of, or payment for, some or all of Company Shares illegal.

(c) The Parent Shares to be issued in the Merger shall be approved for listing on Nasdaq, subject to official notice of issuance;

6.2. Conditions to Obligation of Parent and Merger Sub. The obligations of each of Parent and Merger Sub to consummate the Merger are also subject to satisfaction or waiver by Parent or Merger Sub of the following conditions at or prior to the Effective Time:

(a) (i) Each of the representations and warranties of Company set forth in the first three and the last sentences of Section 3.1(a) and in Sections 3.1(b), 3.1(c), 3.1(d), 3.2, 3.3(i) and 3.13 of this Agreement shall be true and correct in all material respects as of the date of this Agreement and as of the Closing Date as though made on and as of the Closing Date (except to the extent any such representation and warranty expressly speaks as of an earlier date, in which case such representation and warranty shall be true and correct as of such earlier date); and (ii) Parent shall have received a certificate signed on behalf of Company by the chief executive officer and the chief financial officer of Company to such effect.

(b) Company shall have performed in all material respects all obligations required to be performed by it under this Agreement at or prior to the Closing Date, and Parent shall have received a certificate signed on behalf of Company by the chief executive officer and the chief financial officer of Company to such effect.

(c) No Governmental Entity of applicable jurisdiction shall have instituted (or, if instituted, shall not have withdrawn) any proceeding seeking any Governmental Order and no Governmental Entity shall have instituted any civil, criminal or administrative action, suit, claim, hearing, investigation or other proceeding the existence of which would, in the reasonable judgment of Parent, individually or in the aggregate, be reasonably likely to result in a failure of the condition set forth in Section 6.1(b).

(d) All Governmental Consents (other than those described in Section 6.1(a)(i)) the failure of which to make or obtain would, individually or in the aggregate, (i) reasonably be expected to result in a Combined Material Adverse Effect or (ii) provide a reasonable basis to conclude that Parent or any of its directors or officers would be subject to the risk of criminal liability, shall have been made or obtained (such consents, together with those consents that are conditions under Section 6.1(a)(i) being the "*Required Governmental Consents*"). All Governmental Consents that have been obtained shall have been obtained without the imposition of any term, condition or consequence the acceptance of which would, individually or in the aggregate, reasonably be expected to have or result in a Combined Material Adverse Effect.

(e) Company shall have obtained the consent or approval of each Person whose consent or approval shall be required under any Contract set forth in Section 6.2(e) of the Company Disclosure Letter to which Company or any of its Subsidiaries is a party in connection with the transactions contemplated by this Agreement except where the failure to obtain such consent or approval, individually or in the aggregate, would not reasonably be expected to result in a Combined Material Adverse Effect.

(f)' Parent and Merger Sub shall have received the resignations, effective as of the Closing, of each director and officer of Company and its Subsidiaries other than those whom Parent shall have specified in writing at least five business days prior to the Closing.

(g) Company and the Representatives shall have delivered to Parent and Merger Sub an executed counterpart of the Escrow Agreement.

(h) There shall not have been any material default by Company or its Subsidiaries under the Switch Lease and shall be in full force and effect.

(i) Parent shall have received a certificate dated the Closing Date and signed on behalf of Company by the chief executive officer and the chief financial officer of Company certifying as to the respective amounts of the Transaction Costs and the Severance Amount (the *"Closing Costs Certificate"*).

Subject to the provisions of applicable law, Parent and Merger Sub may waive, in whole or in part, any condition specified in this Section 6.2 if they execute a writing so stating at or prior to the Closing.

6.3. Conditions to Obligation of Company. The obligation of Company to consummate the Merger are also subject to satisfaction or waiver by Company of the following conditions at or prior to the Effective Time:

(a) (i) Each of the representations and warranties of Parent and Merger Sub set forth in Sections 4.1, 4.2, 4.3 and 4.4(i) of this Agreement shall be true and correct in all material respects as of the date of this Agreement and as of the Closing Date (except to the extent any such representation and warranty expressly speaks as of an earlier date, in which case such representation and warranty shall be true and correct as of such earlier date); and (ii) Company shall have received a certificate signed on behalf of Parent and Merger Sub by the chief executive officer and the chief financial officer of Parent to such effect.

(b) Each of Parent and Merger Sub shall have performed in all material respects all obligations required to be performed by it under this Agreement at or prior to the Closing Date, and Company shall have received a certificate signed on behalf of Parent and Merger Sub by the chief executive officer and the chief financial officer of Parent to such effect.

(c) Parent shall have delivered to Company an executed counterpart of the Escrow Agreement.

(d) There shall not have been any material default by Parent or its Subsidiaries under the Switch Lease and the Switch Lease shall be in full force and effect.

Subject to the provisions of applicable law, Company may waive, in whole or in part, any condition specified in this Section 6.3 if it executes a writing so stating at or prior to the Closing.

ARTICLE VII

SURVIVAL OF REPRESENTATIONS AND WARRANTIES; INDEMNIFICATION

7.1. Survival of Representations and Warranties. The representations and warranties of the Parties contained in this Agreement shall survive the Closing hereunder (even if the damaged Party knew or had reason to know of any misrepresentation or breach of warranty or covenant at the time of Closing) and continue in full force and effect until the Claim Date (subject, in each case, to any applicable statutes of limitations).

7.2. Indemnification.

(a) Indemnification. From and after the Effective Time:

(1) From the Escrow Fund and as provided in, and subject to the terms and conditions of, this Section 7.2, Parent and its officers, directors, affiliates (including the Surviving Corporation), employees, agents and representatives (each a "Parent Indemnified Party" and. collectively. the "Parent Indemnified Parties"), shall be indemnified and held harmless against all claims (including without limitation Third Party Claims. as defined below, and any claims for indemnification against Company, Surviving Corporation or Parent by officers, directors or employees of Company), losses, liabilities, damages, diminutions in value (which term shall be deemed to exclude any decline in the price of the Parent Stock), deficiencies, costs, interest, awards, amounts paid in settlement, judgments, penalties and expenses, including reasonable attorneys' and consultants' fees and expenses and including any such expenses incurred in connection with investigating, defending against or settling any of the foregoing, but excluding liability for any lost profits or incidental, consequential, indirect or special damages (hereinafter individually a "Loss" and collectively "Losses"), incurred or sustained by the Parent Indemnified Parties, or any of them, directly or indirectly, arising out of or by reason of or as a result of (x) any breach or inaccuracy of a representation or warranty of Company contained in this Agreement or in any certificate or other instrument delivered by or on behalf of Company pursuant to this Agreement or (y) any claim by any holder of Common Stock in respect of or arising out of or as a result or by reason of (or alleged to be in respect of or to arise out of or as a result or by reason of) this Agreement or the transactions contemplated thereby, including the Merger, the approval or adoption thereof or any solicitation of such approval or adoption, except to the extent that any such claim in this clause (y) arises out of or is based upon any untrue statement or omission in written materials furnished by Company to the Stockholders in connection with such solicitation in reliance upon and in conformity with written information furnished to Company by or on behalf of Parent specifically for inclusion in such furnished written materials or (iii) the amount, if any, by which the payment required to be paid to any Dissenting Stockholder in respect of its Preferred Stock exceeds the sum of the Withheld Cash and the Dissenter Cash delivered to Parent pursuant to Section 2.4(f)(4) as provided in, and subject to the terms and conditions of, this Section 7.2, or (iv) the amount, if any, by which the sum of the Transaction Costs and the Severance Amount exceed the Deducted Amount (other than the amount of any such excess as shall have been reflected as a reduction of the amount of the Cash Cap as provided in the definition thereof based on the information in the Closing Costs Certificate.

(2) As provided in, and subject to the terms and conditions of, this Section 7.2, Parent shall indemnify and hold harmless each of the Preferred Stockholders and its shareholders, officers, directors, affiliates, employees, agents and representatives (each a "Company Indemnified Party" and, collectively, the "Company Indemnified Parties"), against all Losses incurred or sustained by the Company Indemnified Parties, or any of them, directly or indirectly, arising out of any breach or inaccuracy of a representation or warranty of Parent or Merger Sub contained in this Agreement or in any certificate or other instrument delivered by or on behalf of Parent or Merger Sub pursuant to this Agreement.

(b) Third Party Claims.

(1) Promptly after receipt by any Person entitled to indemnification under this Section 7.2 (the "Indemnified Party") of notice of the commencement of any action by a third party in respect of which, if successful, the Indemnified Party would be entitled to indemnification under this Section 7.2 (a "Third Party Claim"), the Indemnified Party shall notify each person that is obligated to provide such indemnification (an "Indemnifying Party") thereof in writing, but any failure to so notify the Indemnifying Party shall not relieve the Indemnifying Party from any liability that it may have to the Indemnified Party other than to the extent the Indemnifying Party is actually prejudiced thereby.

(2) The Indemnifying Party shall have the right to assume control of the defense of the Indemnified Party against the Third Party Claim with counsel reasonably satisfactory to such Indemnified Party or, if the Indemnifying Party does not assume such defense, to participate in the defense of such Third Party Claim at its own expense.

(3) So long as the Indemnifying Party is diligently conducting the defense of the Third Party Claim:

- (i) the Indemnified Party shall be entitled to participate in the defense of such claim and to employ counsel at its own cost and expense (which expense shall not constitute a Loss unless the Indemnified Party reasonably determines that the Indemnifying Party, because of a conflict of interest, may not adequately represent any interests of the Indemnified Parties and to the extent such costs and expenses are reasonable) to assist in the handling of such claim;
 - (ii) no Indemnifying Party shall consent to the entry of any judgment or enter into any settlement that does not include as an unconditional term thereof the giving by each claimant or plaintiff to each Indemnified Party of a release from all liability in respect of such claim; and
- (iii) the Indemnifying Party shall not be liable to such Indemnified Party hereunder for any legal expenses subsequently incurred by such Indemnified Party in connection with the defense.

(4) Notwithstanding the foregoing, if (x) a Third Party Claim seeks equitable relief or (y) the subject matter of a Third Party Claim relates, in whole or in part, to the ongoing business of any of the Indemnified Parties, which Claim, if decided against any of the Indemnified Parties, would materially adversely affect the ongoing business of any of the Indemnified Parties, or (z) a Third Party Claim, if decided against any of the Indemnified Parties, would, together with any other claims for indemnification under this Section 7.2 by such Indemnified Parties, result in Losses that would reasonably be expected to exceed, by 25% or more, the maximum amount of the Indemnifying Party's remaining indemnification obligations under this Section 7.2 (after giving effect to the limitations herein on the maximum payments under this Section 7.2 by the Indemnifying Party) then, in any such case, the Indemnified Parties alone shall be entitled to contest, defend and settle such Third Party Claim in the first instance and, if the Indemnified Parties do not contest, defend or settle such Third Party Claim, the Indemnifying Party shall have the right to contest and defend (but not settle) such Third Party Claim. If the Indemnified Parties shall have exercised the right to contest, defend and settle any such Third Party Claim instead of the Indemnifying Party by reason of the foregoing provisions of this Section 7.2(b)(4), such Indemnifying Party shall be entitled, at its own cost and expense, to participate in the defense of such claim and to employ counsel.

(5) The Indemnified Party shall obtain the prior written approval of the Indemnifying Party before admitting liability or entering into any settlement of such claim or ceasing to defend against such claim that the Indemnified Party is defending (with such approval not to be unreasonably withheld or delayed), *provided* that the Indemnified Party shall not be required to obtain approval from the Indemnifying Party in respect of any claims to the extent that the Losses resulting from such admission of liability or settlement or cessation of defense exceed, by 25% or more, the maximum amount of the Indemnifying Party's remaining indemnification obligations under this Section 7.2 (after giving effect to the limitations herein on the maximum payments under this Section 7.2 by the Indemnifying Party).

(6) If the Indemnifying Party does not assume defense of the Third Party Claim, the Indemnified Party shall have the right to defend such claim in such manner as it may deem appropriate at the cost and expense of the Indemnifying Party, and the Indemnifying Party will promptly reimburse the Indemnified Party therefor in accordance with this Section 7.2(b).

(7) If the Indemnified Party is a Parent Indemnified Party, the reimbursement of fees, costs and expenses incurred by the defending party as required by this Section 7.2(b) shall be made from the Escrow Fund by periodic payments during the course of the investigations or defense, as and when bills are received or expenses incurred.

(c) 'Other Claims. In the event any Indemnified Party should have a claim under Section 7.2(a) that does not involve a Third Party Claim being asserted against or sought to be collected from such Indemnified Party, the Indemnified Party shall deliver written notice of such claim with reasonable promptness and stating the nature, basis and amount of the claim in reasonable detail to the Indemnifying Party. The failure by any Indemnified Party so to notify the Indemnifying Party shall not relieve any liability under Section 7.2(a), except to the extent that the ability to defend such claim or demand shall have been prejudiced as a result of such failure. If the Indemnifying Party disputes the liability with respect to such claim, the Indemnifying Party and the Indemnified Party shall proceed in good faith to negotiate a resolution of such dispute and, if not resolved through negotiations, such dispute shall be resolved by litigation in an appropriate court of competent jurisdiction.

(d) Termination of Indemnification.

(1) The rights of the Parent Indemnified Parties to be indemnified, defended and held harmless under Section 7.2(a)(1) shall terminate at 11:59 p.m. EST on the Claim Date; *provided, however*, that such obligations to indemnify, defend and hold harmless shall not terminate with respect to any item as to which the Person to be indemnified or the related party thereto shall have, before the expiration of the applicable period, previously made a claim by delivering a notice of such claim (stating in reasonable detail the basis and amount of such claim) to the Company and Preferred Stockholders.

(2) The rights of the Company Indemnified Parties to be indemnified, defended and held harmless under Section 7.2(a)(2) shall terminate at 11:59 p.m. EST on the Claim Date; *provided, however*, that such obligations to indemnify, defend and hold harmless shall not terminate with respect to any item as to which the Person to be indemnified or the related party thereto shall have, before the expiration of the applicable period, previously made a claim by delivering a notice of such claim (stating in reasonable detail the basis and amount of such claim) to Parent.

(e) Escrow Fund; Maximum Payments; Remedy.

(1) By virtue of this Section 7.2 and as security for the indemnity obligations provided for in Section 7.2(a) hereof, at the Effective Time, Parent will deposit with the Escrow Agent the Escrowed Consideration as provided in Section 2.5(a). Such deposit of cash and Parent Shares constituting the Escrowed Consideration (the aggregate value of which, valuing the Parent Shares at the Parent Share Valuation, the "*Escrow Amount*") shall constitute an escrow fund (the "*Escrow Fund*") to be governed by the terms set forth in the Escrow Agreement. Solely for purposes of this Section 7.2 and the allocation of interests in the Escrow Fund among the Preferred Stockholders, and without limitation of the amount of the Escrow Fund available for satisfaction of claims by the Parent Indemnified Parties under this Section 7.2, each Preferred Stockholder shall be deemed to have received and deposited such Preferred Stockholder's *pro rata* share of the Escrow Fund (plus any additional Parent Shares as may be issued upon any stock split, stock dividend or recapitalization effected by Parent after the Effective Time with respect to the Parent Shares constituting part of the Escrow Fund) with the Escrow Agent as provided in Section 2.5(a) of this Agreement. The Escrow Fund shall be deposited with, and shall be held by, the Escrow Agent and shall be available to compensate the Parent Indemnified Parties for any claims by such parties for any Losses suffered or incurred by them and for which they are entitled to recovery under this Agreement. Claims by the Parent Indemnified Parties under this Section 7.2 for Losses shall be limited to, and shall be satisfied solely from, the Escrow Fund.



(2) Any payment to any Parent Indemnified Party under this Section 7.2 from the Escrow Fund shall be deemed made by the Preferred Stockholders on a joint and several basis, provided that each Preferred Stockholder shall have the option to elect, by instruction to the Escrow Agent given as provided in the Escrow Agreement, to have its Allocated Portion (as defined below) of any such payment paid by delivery of (i) cash, (ii) Parent Shares or (iii) a combination of cash and Parent Shares, in each case to the extent of such Preferred Stockholder's Allocated Portion of the Escrow Amount and of the cash and Parent Shares constituting such Preferred Stockholder's Allocated Portion of the Escrow Fund, and with the Parent Shares valued for purposes of such payment at the Parent Share Valuation. At any time, any Preferred Stockholder may substitute cash for such Preferred Stockholder's Parent Shares constituting part of the Escrow Fund, with the Parent Shares being valued for such purpose at the Parent Share Valuation. In connection with any payment as provided in this Section 7.2(e)(2) that shall be made by causing the delivery of Parent Stock out of the Escrow Fund, each of Parent and the Preferred Stockholders shall cooperate in good faith and use its reasonable best efforts to promptly make and promptly process all necessary filings and applications and obtain all consents, approvals, orders, authorizations, registrations and declarations and to comply with all applicable laws necessary for the Preferred Stockholders to make such payment. The "Allocated Portion" of any Preferred Stockholder of (i) any payment to any Parent Indemnified Party or the Escrow Amount is the portion thereof as bears the same relationship to the total payment or the Escrow Amount, as the case may be, as the value of the Merger Consideration to be received by such Preferred Stockholder in accordance with this Agreement (valuing any Parent Shares received at the Parent Share Valuation) bears to the Escrow Amount, and (ii) cash, Parent Shares and the Escrow Fund is the cash and Parent Shares that were to be received as Merger Consideration by such Preferred Stockholder and were deposited in the Escrow Fund, in each case as such Parent Shares may have been replaced by cash by such Preferred Stockholder as herein provided.

(3) Claims by the Company Indemnified Parties under this Section 7.2 for Losses shall be limited to, and shall not exceed, an amount equal to the Escrow Amount, reduced dollar-for-dollar by the amount, if any, of any direct or indirect payment by or on behalf of Parent to any Stockholder or former Stockholder, in each case in its capacity as a Stockholder or former Stockholder, by reason of any claim, statutory or otherwise, in respect of the Merger or this Agreement or the Parent Stock that is made other than under this Section 7.2 (other than (i) the payment of the Merger Consideration as provided in this Agreement, (ii) payments by Parent to the extent that any such payment is by reason of a loss, claim, damage or liability arising out of or based upon any untrue statement or alleged untrue statement or omission or alleged omission in written materials furnished by Company to the Stockholders, in connection with the solicitation of the Stockholders of their approval of this Agreement, in reliance upon and in conformity with written information furnished to Company by or on behalf of Parent specifically for inclusion in such furnished written materials, and (iii) payments made to any Stockholders for breach by Parent of any obligation of Parent under this Agreement that is expressly stated herein to be for the benefit of such Stockholder). No payment by Parent of Losses to any Company Indemnified Party shall be made until the expiration of Parent's indemnification period, as set forth in Section 7.2(e)(2), and any such payment may be withheld or reduced (x) pending the final resolution of any claim that would have reduced the Escrow Amount indemnity obligation of Parent as provided above in this Section 7.2(e)(2) had payment thereof been made prior to such expiration date and as to which Parent shall have, before such expiration date, given notice of such claim (stating in reasonable detail the basis and amount of such claim) to the Company and Preferred Stockholders and (y) by the amount at issue in such claim.

(f) Basket Amount. Notwithstanding any provision of this Section 7.2 to the contrary, an Indemnified Party may not recover any 'Losses under Section 7.2(a) hereof unless and until such Losses exceed \$150,000 (the "Basket Amount"), in which case the Indemnified Party shall be entitled to recover all Losses, including the Basket Amount.

(g) Insurance; Tax Benefit; Mitigation.

(1) All indemnification or reimbursement payments required pursuant to this Section 7.2 shall be (i) made net of all insurance proceeds actually received by the party to be indemnified, and (ii) reduced to take account of any net tax benefit when and as realized by the indemnified party arising from the incurrence or payment of any Loss. Proceeds received from an insurance carrier which are subject to a reservation of rights by the carrier that has not been released in writing by the carrier (*"Reserved Insurance Proceeds"*) shall be taken into account in the calculation of Losses once received by the party to be indemnified; *provided, however*, that Reserved Insurance Proceeds with respect to which the insurance carrier subsequently obtains reimbursement shall be deemed to be Losses on a dollar for dollar basis with the amount by which Losses were previously reduced as a result of the receipt of such proceeds. Purchaser Indemnified Parties and Company Indemnified Parties shall each use commercially reasonable efforts to obtain a release of a reservation of rights from the insurance carrier promptly following the settlement or final adjudication of the claim or proceeding to which the Reserved Insurance Proceeds related.

(2) Each Indemnified Party shall use commercially reasonable efforts to mitigate Losses, including seeking recovery under insurance policies. If a Parent Indemnified Party is the Indemnified Party, it shall reimburse the Escrow Fund (if the Escrow Agreement is still in effect) or the Preferred Stockholders (if the Escrow Agreement has been terminated), or (y) if a Company Indemnified Party is the Indemnified Party, it shall reimburse Parent, for any Loss indemnified by them to the extent all or a portion of such Loss (net of reasonable collection costs) is subsequently recovered by the Indemnified Party under any such insurance, but in each case only to the extent that the amount of such recovery exceeds the amount of the Losses claimed by such Indemnified Party that have not been paid to such Indemnified Party because of the limitations of Section 6 hereof.

(h) Exclusive Remedy.

(1) From and after the Closing and to the fullest extent allowed by law, the indemnification pursuant to this Section 7.2 shall be the exclusive remedy of the Parent Indemnified Parties for any Loss or Losses arising out of or by reason of or as a result of any matter described in clause (i), (ii), (iii) or (iv) of Section 7.2(a)(1).

(2) From and after the Closing and to the fullest extent allowed by law, the indemnification pursuant to this Agreement shall be the exclusive remedy of the Company Indemnified Parties for any Loss or Losses arising out of or by reason of or as a result of any matter described in Section 7.2(a)(2).

(3) Notwithstanding the foregoing in this Section 7.2(h), nothing herein shall prevent any of the Indemnified Parties from bringing an action based upon allegations of fraud or other intentional breach of an obligation of or with respect to any party in connection with this Agreement or the Switch Lease.

ARTICLE VIII TERMINATION

8.1. Termination of Agreement. The Parties may terminate this Agreement and the Merger may be abandoned as provided below at any time prior to the Effective Time before or after the approval by the Stockholders or Merger Sub stockholder:

(a) the Parties may terminate this Agreement by mutual written consent of Parent and Company with the prior authorization of their respective board of directors;

(b) Parent may terminate this Agreement with the prior authorization of the Parent Board by giving written notice to Company at any time prior to the Closing in the event Company has breached any of its representations, warranties, covenants or agreements contained in this Agreement in any material respect, or any such representation or warranty shall have become untrue or incorrect after the execution of this Agreement, such that Section 6.2(a) or Section 6.2(b), as the case may be, would not be satisfied, and (x) such breach is not reasonably capable of being cured prior to the Termination Date or (y) in the case of a breach of a covenant or agreement, if such breach is reasonably capable of being cured prior to the Termination Date, such breach shall not have been cured prior to the Termination Date;

(c) Company may terminate this Agreement with the prior authorization of the Company Board by giving written notice to Parent at any time prior to the Closing in the event Parent or Merger Sub has breached any of its representations, warranties, covenants or agreements contained in this Agreement in any material respect, or any such representation or warranty shall have become untrue or incorrect after the execution of this Agreement, such that Section 6.3(a) or Section 6.3(b), as the case may be, would not be satisfied, and (x) such breach is not reasonably capable of being cured prior to the Termination Date or (y) in the case of a breach of a covenant or agreement, if such breach is reasonably capable of being cured prior to the Termination Date, such breach shall not have been cured prior to the Termination Date; and

(d) either Company or Parent may terminate this Agreement with the prior authorization of its respective board of directors by giving written notice to the other Party if (i) the Closing shall not have occurred on or before December 31, 2005 (the "*Termination Date*"), by reason of the failure of any condition precedent under ARTICLE VI hereof; or (ii) there shall be any law or regulation that makes consummation of the Merger illegal or otherwise prohibited or any judgment, injunction, order or decree of any court or governmental body having competent jurisdiction enjoining Company or Parent from consummating the Merger and such judgment, injunction, order or decree shall have become final and nonappealable; *provided* that the right to terminate this Agreement pursuant to this Section 8.1(d) shall not be available to any Party that has breached its obligations under this Agreement or under any other agreement contemplated hereunder in any manner that shall have proximately contributed to the occurrence of the failure of a condition to the consummation of the Merger.

8.2. Effect of Termination. If this Agreement is terminated pursuant to Section 8.1, all rights and obligations of the Parties hereunder shall terminate without any liability of any Party to any other Party, except that the provisions of the Confidentiality Agreement and the provisions of Sections 8.2 and 10.11 shall survive any such termination, and *provided, however*, except as otherwise provided herein, no such termination shall relieve any Party of any liability or damages resulting from any willful or intentional breach of any covenant in this Agreement.

ARTICLE IX REGISTRATION RIGHTS

9.1. Definitions. For purposes of this ARTICLE IX, the following capitalized terms shall have the following meanings:

"Holder" means the holders, from time to time, of the Securities.

۰,

"*Majority Holders*" means, at any time after the effectiveness of a Shelf Registration Statement under this ARTICLE IX, the Holders of a majority of the then outstanding number of Securities registered under a Shelf Registration Statement and means, at any time after the date of this Agreement and prior to the effectiveness of a Shelf Registration Statement under this ARTICLE IX, the Holders of a majority of the then outstanding number of Securities.

"Prospectus" means the prospectus included in any Shelf Registration Statement (including, without limitation, a prospectus that discloses information previously omitted from a prospectus filed as part of an effective registration statement in reliance upon Rule 430A under the Securities Act), as amended or supplemented by any prospectus supplement, with respect to the terms of the offering of any portion of the Securities covered by such Shelf Registration Statement, and all amendments and supplements to such prospectus, including post–effective amendments.

"Securities" means the Parent Shares issued as Merger Consideration pursuant to Section 2.4(f)(2) of this Agreement, except such Parent Shares as shall have been sold pursuant to the Shelf Registration Statement or otherwise transferred by the Holder thereof pursuant to an exemption from registration under the Securities Act as a result of which exempt transfer the transferee's disposition of the Parent Shares would not be restricted under the Securities Act.

"Shelf Registration" means a registration effected pursuant to Section 9.2.

"Shelf Registration Period" has the meaning set forth in Section 9.2(b).

"Shelf Registration Statement" means a "shelf" registration statement of Parent pursuant to the provisions of Section 9.2 hereof that covers some or all of the Securities, as applicable, on an appropriate form under Rule 415 under the Securities Act, or any similar rule that may be adopted by the SEC, and all amendments and supplements to such registration statement, including post-effective amendments, in each case including the Prospectus contained therein, all exhibits thereto and all material incorporated by reference therein.

9.2. Shelf Registration; Suspension of Use of Prospectus.

(a) After the Closing Date, Parent shall prepare and, as promptly as practicable after the Closing Date, but not later than 45 days (or 71 days if the delay in filing within such 45-day period is due to the failure of Company and Company's independent auditors to provide all Company's financial information (and pro forma financial information based thereon) required to be included in the Shelf Registration Statement beginning promptly after the date hereof or to otherwise cooperate in preparing such financial information (and pro forma financial information Statement relating to the offer and sale of the Securities by the Holders from time to time in accordance with the methods of distribution elected by such Holders and set forth in such Shelf Registration Statement, and thereafter shall use its best efforts to cause such Shelf Registration to be declared effective under the Securities Act.

(b) Parent shall use its best efforts to keep the Shelf Registration Statement continuously effective in order to permit the Prospectus forming part thereof to be usable by Holders until the earliest of (i) the second anniversary of the Closing Date, (ii) the date on which the Securities may be sold pursuant to paragraph (k) of Rule 144 (or any successor provision) promulgated by the SEC under the Securities Act by a person that is not, and has not been during the preceding three months, an "affiliate," as such term is used in such paragraph (k), of Parent, and (iii) such date as of which all the Securities have been sold pursuant to the Shelf Registration Statement or otherwise transferred by the Holder thereof pursuant to an exemption from registration under the Securities Act as a result of which exempt transfer the transferee's disposition of the Parent Shares would not be restricted under the Securities Act (in any such case, such period being called the "Shelf Registration Period"). Parent shall be deemed not to have used its best efforts to keep the Shelf Registration Statement effective during the requisite period if it voluntarily takes any action that would result in Holders of Securities covered thereby not being able to offer and sell such Securities during that period, unless such action is required (x) by applicable law or (y) pursuant to Section 9.2(c) hereof, and, in either case, so long as Parent promptly thereafter complies with the requirements of Securition 9.3(i) hereof, if applicable.

(c) Parent may suspend the use of the Prospectus for a period not to exceed 45 days in any three-month period or for three periods not to exceed an aggregate of 60 days in any twelve-month period for valid business reasons, to be determined by Parent Board in its sole reasonable judgment (not including avoidance of Parent's obligations hereunder), including, without limitation, the acquisition or divestiture of assets, public filings with the SEC, pending corporate developments and similar events; *provided* that Parent promptly thereafter complies with the requirements of Section 9.3(i) hereof, if applicable.

9.3. Registration Procedures. In connection with any Shelf Registration Statement, the following provisions shall apply:

(a) Parent shall furnish to the Holders, prior to the filing thereof with the SEC, a copy of any Shelf Registration Statement, and each amendment thereof and each amendment or supplement, if any, to the Prospectus included therein.

(b) Parent shall ensure that (i) any Shelf Registration Statement and any amendment thereto and any Prospectus forming part thereof and any amendment or supplement thereto comply in all material respects with the Securities Act and the rules and regulations thereunder, (ii) any Shelf Registration Statement and any amendment thereto does not, when it becomes effective, contain an untrue statement of a material fact or omit to state a material fact required to be stated therein or necessary to make the statements therein not misleading and (iii) any Prospectus forming part of any Shelf Registration Statement, and any amendment or supplement to such Prospectus, does not include an untrue statement of a material fact or omit to state a material fact necessary in order to make the statements therein, in light of the circumstances under which they were made, not misleading; *provided* that no representation or agreement is made hereby with respect to information provided in writing by any Holder with respect to such Holder required to be included in any Shelf Registration or Prospectus pursuant to the Securities Act or the rules and regulations thereunder.

(c) (1) Parent shall advise the Holders and, if requested by any such Holder, confirm such advice in writing:

- (i) when a Shelf Registration Statement and any amendment thereto has been filed with the SEC and when the Shelf Registration Statement or any post-effective amendment thereto has become effective; and
 - (ii) of any request by the SEC for amendments or supplements to the Shelf Registration Statement or the Prospectus included therein or for additional information and of receipt of any comment letters from the SEC in respect of the Shelf Registration Statement or the Prospectus.
- (2) Parent shall advise the Holders and, if requested by any such Holder, confirm such advice in writing:
 - (i) of the issuance by the SEC of any stop order suspending the effectiveness of the Shelf Registration Statement or the initiation of any proceedings for that purpose;
 - (ii) of the receipt by Parent of any notification with respect to the suspension of the qualification of the Securities included in any Shelf Registration Statement for sale in any jurisdiction or the initiation or threat of any proceeding for such purpose; and
 - (iii) of the suspension of the use of the Prospectus pursuant to Section 2(c) hereof or of the happening of any event that requires the making of any changes in the Shelf Registration Statement or the Prospectus so that, as of such date, the statements therein are not misleading and do not omit to state a material fact required to be stated therein or necessary to make the statements therein (in the case of the Prospectus, in light of the circumstances under which they were made) not misleading (which advice shall be accompanied by an instruction to suspend the use of the Prospectus until the requisite changes have been made).



(d) Parent'shall use its best efforts to obtain the withdrawal of any order suspending the effectiveness of any Shelf Registration Statement at the earliest possible time.

(e) Parent shall furnish to each Holder of Securities included within the coverage of any Shelf Registration Statement, without charge, at least one copy of such Shelf Registration Statement and any post-effective amendment thereto, including financial statements and schedules, and, if the Holder so requests in writing, all exhibits (including those incorporated by reference).

(f) Parent shall, during the Shelf Registration Period, deliver to each Holder of Securities included within the coverage of any Shelf Registration Statement, without charge, as many copies of the Prospectus (including each preliminary Prospectus) included in such Shelf Registration Statement and any amendment or supplement thereto as such Holder may reasonably request; and Parent consents to the use of the Prospectus or any amendment or supplement thereto by each of the selling Holders in connection with the offering and sale of the Securities covered by the Prospectus or any amendment or supplement thereto.

(g) Prior to any offering of Securities pursuant to any Shelf Registration Statement, Parent shall register or qualify or cooperate with the Holders of Securities included therein and their respective counsel in connection with the registration or qualification of such Securities for offer and sale under the securities or blue sky laws of such jurisdictions as any such Holders reasonably request in writing and do any and all other acts or things necessary or advisable to enable the offer and sale in such jurisdictions of the Securities covered by such Shelf Registration Statement; *provided, however*, that Parent will not be required to qualify generally to do business in any jurisdiction where it is not then so qualified or to take any action that would subject it to general service of process or to taxation in any such jurisdiction where it is not then so subject.

(h) Parent shall cooperate with the Holders to facilitate the timely preparation and delivery of certificates representing Securities to be sold pursuant to any Shelf Registration Statement free of any restrictive legends (except such as may be required by Section 5.8) and in such denominations and registered in such names as Holders may request prior to sales of Securities pursuant to such Shelf Registration Statement.

(i)' Upon the occurrence of any event contemplated by Section 9.3(c)(2)(iii) above, Parent shall promptly prepare a post-effective amendment to any Shelf Registration Statement or an amendment or supplement to the related Prospectus or file any other required document so that, as thereafter delivered to purchasers of the Securities included therein, the Prospectus will not include an untrue statement of a material fact or omit to state any material fact necessary to make the statements therein, in the light of the circumstances under which they were made, not misleading.

(j) Parent may require each Holder of Securities to be sold pursuant to any Shelf Registration Statement to furnish to Parent such information regarding the Holder and the distribution of such Securities as may, from time to time, be required by the Securities Act and the rules and regulations promulgated thereunder, and the obligations of Parent to any Holder hereunder shall be expressly conditioned on the compliance of such Holder with such request.

9.4. Registration Expenses. Parent shall bear all expenses incurred in connection with the performance of its obligations under Sections 9.2 and 9.3 hereof. Notwithstanding the provisions of this Section 9.4, each Holder shall bear the expense of any broker's commission, agency fee or any other discount or commission.

9.5. Indemnification and Contribution.

(a) In connection with any Shelf Registration Statement, Parent agrees to indemnify and hold harmless each Holder of Securities covered thereby, the directors, officers, employees and agents of each such Holder and each person who controls any such Holder within the meaning of either the Securities Act or the Securities Exchange Act against any and all losses, claims, damages or liabilities, joint or several, to which they or any of them may become subject under the Securities Act, the Securities Exchange Act or other Federal or state statutory law or regulation, at common law or otherwise, insofar as such losses, claims, damages or liabilities (or actions in respect thereof) arise out of or are based upon any untrue statement or alleged untrue statement of a material fact contained in the Shelf Registration Statement as originally filed or in any amendment thereof, or in any preliminary Prospectus or Prospectus, or in any amendment thereof or supplement thereto, or arise out of or are based upon the omission or alleged omission to state therein a material fact required to be stated therein or necessary to make the statements therein not misleading, and agrees to reimburse each such indemnified party, as incurred, for any legal or other expenses reasonably incurred by them in connection with investigating or defending any such loss, claim, damage, liability or action; provided, however, that Parent will not be liable in any case to the extent that any such loss, claim, damage or liability arises out of or is based upon (A) any such untrue statement or alleged untrue statement or omission or alleged omission made therein in reliance upon and in conformity with written information furnished to Parent by or on behalf of any such Holder specifically for inclusion therein, (B) use of a Shelf Registration Statement or the related Prospectus during a period when a stop order has been issued in respect of such Shelf Registration or any proceedings for that purpose have been initiated or use of a Prospectus when use of such Prospectus has been suspended pursuant to Section 9.2(c); provided, further, in each case, that Holders received prior notice of such stop order, initiation of proceedings or suspension or (C) if the Holder fails to deliver a Prospectus or the then current Prospectus to the extent required. This indemnity agreement will be in addition to any liability that Parent may otherwise have.



(b) Each Holder of Securities covered by a Shelf Registration Statement (including the Initial Stockholders) severally agrees to indemnify and hold harmless (i) Parent, (ii) each of its directors, (iii) each of its officers who signs such Shelf Registration Statement and (iv) each person who controls Parent within the meaning of either the Securities Act or the Securities Exchange Act to the same extent as the foregoing indemnity from Parent to each such Holder, but only with reference to written information relating to such Holder furnished to Parent by or on behalf of such Holder specifically for inclusion in the documents referred to in the foregoing indemnity. This indemnity agreement will be in addition to any liability that any such Holder may otherwise have.

(c) Promptly after receipt by an indemnified party under this Section 9.5 of notice of the commencement of any action, such indemnified party will, if a claim in respect thereof is to be made against the indemnifying party under this Section 9.5, notify the indemnifying party in writing of the commencement thereof; but the failure so to notify the indemnifying party (i) will not relieve it from liability under Section 9.5(a) or 9.5(b) above unless and to the extent it did not otherwise learn of such action and such failure results in the forfeiture by the indemnifying party of substantial rights and defenses and (ii) will not, in any event, relieve the indemnifying party from any obligations to any indemnified party other than the indemnification obligation provided in Section 9.5(a) or 9.5(b) above. The indemnifying party shall be entitled to appoint counsel of the indemnifying party's choice at the indemnifying party's expense to represent the indemnified party in any action for which indemnification is sought (in which case the indemnifying party shall not thereafter be responsible for the fees and expenses of any separate counsel retained by the indemnified party or parties except as set forth below); provided, however, that such counsel shall be reasonably satisfactory to the indemnified party. Notwithstanding the indemnifying party's election to appoint counsel to represent the indemnified party in an action, the indemnified party shall have the right to employ separate counsel (including local counsel), and the indemnifying party shall bear the reasonable fees, costs and expenses of such separate counsel (and local counsel) if (i) the use of counsel chosen by the indemnifying party to represent the indemnified party would present such counsel with a conflict of interest, (ii) the actual or potential defendants in, or targets of, any such action include both the indemnified party and the indemnifying party and the indemnified party shall have reasonably concluded that there may be legal defenses available to it and/or other indemnified parties that are different from or additional to, and may conflict with, those available to the indemnifying party, (iii) the indemnifying party shall not have employed counsel reasonably satisfactory to the indemnified party to represent the indemnified party within a reasonable time after notice of the institution of such action or (iv) the indemnifying party shall authorize the indemnified party to employ separate counsel at the expense of the indemnifying party; provided further, that the indemnifying party shall not be responsible for the fees and expenses of more than one separate counsel (together with appropriate local counsel) representing all the indemnified parties under Section 9.5(a) or 9.5(b) above. An indemnifying party will not, without the prior written consent of the indemnified parties, settle or compromise or consent to the entry of any judgment with respect to any pending or threatened claim, action, suit or proceeding in respect of which indemnification or contribution may be sought hereunder (whether or not the indemnified parties are actual or potential parties to such claim or action) unless such settlement, compromise or consent includes an unconditional release of each indemnified party from all liability arising out of such claim, action, suit or proceeding.

(d) In the event that the indemnity provided in Section 9.5(a) or 9.5(b) is unavailable to or insufficient to hold harmless an indemnified party for any reason, then each applicable indemnifying party, in lieu of indemnifying such indemnified party, shall have a joint and several obligation to contribute to the aggregate losses, claims, damages and liabilities (including legal or other expenses reasonably incurred in connection with investigating or defending same) (collectively "Losses") to which such indemnified party may be subject in such proportion in such proportion as is appropriate to reflect the relative fault of the indemnifying party, on the one hand, and such indemnified party, on the other hand, in connection with the statements or omissions that resulted in such Losses. The relative fault shall be determined by reference to, among other things, whether the untrue or alleged untrue statement of a material fact or the omission or alleged omission to state a material fact relates to information supplied by Parent or the Holders of Securities covered by the Registration Statement in question and the parties' relative intent, knowledge, access to information and opportunity to correct or prevent such statement or omission. The parties agree that it would not be just and equitable if contribution were determined by pro rata allocation or any other method of allocation that does not take account of the equitable considerations referred to above. Notwithstanding the provisions of this Section 9.5(d), (i) no person guilty of fraudulent misrepresentation (within the meaning of Section 11(f) of the Securities Act) shall be entitled to contribution from any person who was not guilty of such fraudulent misrepresentation, and (ii) no party shall be liable for contribution under this Section 9.5(d) except to the extent and under such circumstances as such party would have been liable to indemnify under this Section 9.5 if such indemnification were enforceable under applicable law. For purposes of this Section 9.5, each person who controls a Holder within the meaning of either the Securities Act or the Securities Exchange Act and each director, officer, employee and agent of such Holder shall have the same rights to contribution as such Holder, and each person who controls Parent within the meaning of either the Securities Act or the Securities Exchange Act, each officer of Parent who shall have signed the Shelf Registration Statement and each director of Parent shall have the same rights to contribution as Parent, subject in each case to the applicable terms and conditions of this Section 9.5(d).

(e) The provisions of this Section 9.5 will remain in full force and effect, regardless of any investigation made by or on behalf of any Holder or Parent or any of the officers, directors or controlling persons referred to in Section 9.5 hereof, and will survive the sale by a Holder of Securities covered by a Shelf Registration Statement.

9.6. Successors and Assigns; Survival. The provisions of this ARTICLE IX shall inure to the benefit of and be binding upon the successors and assigns of each of Parent and the Holders, including, without the need for an express assignment or any consent by Parent thereto, subsequent Holders. Parent hereby agrees to extend the benefits of the provisions of this ARTICLE IX to any Holder and any such Holder may specifically enforce the provisions of this ARTICLE IX as if an original Holder. The provisions of this ARTICLE IX shall survive the consummation of the Merger and shall terminate on the last day of the Shelf Registration Period, except that the provisions of Section 9.5 shall survive until the sixth anniversary of the last day of the Shelf Registration Period and the provisions of Section 9.4 shall survive such termination of this ARTICLE IX.

ARTICLE X MISCELLANEOUS

10.1. Press Releases and Public Announcements. The initial press release disclosing this Agreement shall be a press release issued jointly by Parent and Company. Company shall issue no press releases or otherwise make any public announcements with respect to the Merger or the other transactions contemplated by this Agreement without the express written consent of Parent. Parent shall consult with Company prior to issuing any press releases or otherwise making public announcements with respect to the Merger and the other transactions contemplated by this Agreement and, except as may be required by applicable Laws or by obligations pursuant to any listing agreement with or rules of Nasdaq or by the request of any Governmental Entity, will not issue any such press releases or make any such public statement prior to such consultation. Company and Parent each shall consult with the other prior to making any filings with any third party or any Governmental Entity (including any national securities exchange) with respect thereto.

10.2. No Third-Party Beneficiaries. This Agreement is not intended to, and does not, confer any rights or remedies hereunder upon any Person other than the Parties and their respective successors and permitted assigns; *provided, however*, that (a) the Stockholders shall be deemed third party beneficiaries solely with respect to the right to receive the Per Share Merger Consideration pursuant to ARTICLE II hereof, (b) the Indemnified Parties shall be deemed third party beneficiaries solely with respect to Section 7.2, (c) the Indemnified Persons shall be deemed third party beneficiaries solely with respect to Section 5.6, and (d) the Holders shall be deemed third party beneficiaries solely with respect to ARTICLE IX.

10.3. Entire Agreement. This Agreement and the other documents referred to herein, including the Confidentiality Agreement, the Escrow Agreement and the Switch Lease constitute the entire agreement among the Parties and supersede all other prior understandings, agreements or representations by or among the Parties, written or oral, to the extent they relate in any way to the subject matter hereof.

10.4. Binding Effect; Assignment. This Agreement shall be binding upon and inure to the benefit of the Parties and their respective successors and permitted assigns. This Agreement shall not be assignable by operation of law or otherwise; provided, however, that Parent may designate, by written notice to Company, another wholly owned direct or indirect subsidiary to be a Constituent Corporation in lieu of Merger Sub, in which event all references herein to Merger Sub shall be deemed references to such other subsidiary, except that all representations and warranties made herein with respect to Merger Sub as of the date of this Agreement shall be deemed representations and warranties made with respect to such other subsidiary as of the date of such designation; provided that any such designation shall not relieve Parent or Merger Sub of any of its obligations hereunder or materially impede or delay the consummation of the transactions contemplated by this Agreement or otherwise materially impede the rights of the Stockholders under this Agreement. Any purported assignment in violation of this Agreement will be void ab initio.

10.5. Counterparts. This Agreement may be executed (including by facsimile) in one or more counterparts, each of which shall be deemed an original instrument but all of which together will constitute the same instrument.

10.6. Headings. The table of contents and headings contained in this Agreement are for convenience of reference only, do not constitute part of this Agreement and shall not be deemed to limit or otherwise affect in any way the meaning or interpretation of this Agreement.

10.7. Notices. Any notice, request, demand, claim or other communication or document to be given hereunder shall be in writing and delivered personally or sent by registered or certified mail, return receipt requested, postage prepaid, by facsimile or by overnight courier, addressed to the intended recipient as set forth below:

If to Company:

LDMI Telecommunications, Inc. 277777 Franklin Road, Suite 500 Southfield, Michigan 48034 Attention: Patrick O'Leary, President and Chief Executive Officer Facsimile: (888) 877-5364

with a Copy to:

Dykema Gossett PLLC 400 Renaissance Center Detroit, Michigan 48243 Attention: Thomas S. Vaughn, Esq. Facsimile: (313) 568-6915

If to Parent:

Talk America Holdings, Inc. 6805 Route 202 New Hope, PA 18938 Attention: Aloysius T. Lawn, IV, Executive Vice President – General Counsel Facsimile: 215.862.1960

with a Copy to:

Arnold & Porter LLP 399 Park Avenue New York, New York 10022 Attention: Jonathan C. Stapleton, Esq. Facsimile: (212) 715-1111

If to Merger Sub:

Lion Acquisition Corp. c/o Talk America Holdings, Inc. 6805 Route 202 New Hope, PA 18938 Attention: Aloysius T. Lawn, IV, Executive Vice President – General Counsel Facsimile: 215.862.1960

with a Copy to:

Arnold & Porter LLP 399 Park Avenue New York, New York 10022 Attention: Jonathan C. Stapleton, Esq. Facsimile: (212) 715-1111,

if to any Holder under ARTICLE IX:

At the record address of such Holder as maintained by Parent's transfer agent, or to such other persons or addresses as may be designated in writing by the party to receive such notice as provided above. Any notice, request, instruction or other communication or document given as provided above shall be deemed given to the receiving party upon actual receipt, if delivered personally; three business days after deposit in the mail, if sent by registered or certified mail; upon confirmation of successful transmission if sent by facsimile (provided that if given by facsimile such notice, request, instruction or other document shall be followed up within one business day by dispatch pursuant to one of the other methods described herein); or on the next business day after deposit with an overnight courier.

10.8. Governing Law. THIS AGREEMENT SHALL BE GOVERNED BY AND CONSTRUED IN ACCORDANCE WITH THE LAWS OF THE STATE OF DELAWARE WITHOUT GIVING EFFECT TO ANY CHOICE OR CONFLICT OF LAW PROVISION OR RULE (WHETHER OF THE STATE OF DELAWARE OR ANY OTHER JURISDICTION) THAT WOULD CAUSE THE APPLICATION OF THE LAWS OF ANY JURISDICTION OTHER THAN THE STATE OF DELAWARE, EXCEPT THAT THE MICHIGAN LAW SHALL APPLY TO THE EFFECTUATION OF THE MERGER.

10^t9. Amendments and Waivers. The Parties may mutually amend any provision of this Agreement at any time prior to the Effective Time with the prior authorization of their respective boards of directors; except that (a) any amendment will be subject to the restrictions contained in the Michigan Law, to the extent applicable, and (b) any amendment to the provisions of ARTICLE IX and this clause (b) and any waiver or consent to departure from the provisions thereof shall require the consent of the Majority Holders. No amendment of any provision of this Agreement shall be valid unless the same shall be in writing and signed by all of the Parties. The conditions to each of the Parties' obligations to consummate the Merger are for the sole benefit of such Party and may be waived by such Party in whole or in part to the extent permitted by applicable Laws. No waiver by any Party of any default, misrepresentation or breach of warranty or covenant hereunder, whether intentional or not, shall be deemed to extend to any prior or subsequent default, misrepresentation or breach of warranty or covenant hereunder or affect in any way any rights arising by virtue of any prior or subsequent such occurrence.

10.10. Severability. The provisions of this Agreement shall be deemed severable and the invalidity or unenforceability of any provision shall not affect the validity or enforceability of the other provisions hereof. If any provision of this Agreement, or the application thereof to any Person or any circumstance, is invalid or unenforceable, (a) a suitable and equitable provision shall be substituted therefor in order to carry out, so far as may be valid and enforceable, the intent and purpose of such invalid or unenforceable provision and (b) the remainder of this Agreement and the application of such provision to other Persons or circumstances shall not be affected by such invalidity or unenforceability, nor shall such invalidity or unenforceability affect the validity or enforceability of such provision, or the application thereof, in any other jurisdiction.

10.11. Expenses. Except as expressly set forth elsewhere in this Agreement, each of Company and Parent shall bear its own costs and expenses (including legal fees and expenses) incurred in connection with this Agreement and the transactions contemplated hereby.

10.12. Survival. This ARTICLE X and the agreements of Company, Parent and Merger Sub contained in Sections 5.6, 5.7, 5.8 and 7.2 and in ARTICLE IX and the agreements of the Preferred Stockholders in Section 5.8 shall survive the consummation of the Merger. This ARTICLE X and the agreements of Company, Parent and Merger Sub contained in Section 10.11 (Expenses) and Section 8.2 (Effect of Termination) and the Confidentiality Agreement shall survive the termination of this Agreement. If the Merger shall be consummated, the representations and warranties of Company set forth in ARTICLE III and the representations and warranties of Parent set forth in ARTICLE IV shall survive the consummation of the Merger for the period set forth in Section 7.1; except as otherwise set forth in this Section 10.12, all other representations, warranties, covenants and agreements in this Agreement shall not survive the consummation of this Agreement.

70

10.13. Construction.

1.0

(a) The Parties have participated jointly in the negotiation and drafting of this Agreement. In the event an ambiguity or question of intent or interpretation arises, this Agreement shall be construed as if drafted jointly by the Parties and no presumption or burden of proof shall arise favoring or disfavoring any Party by virtue of the authorship of any of the provisions of this Agreement.

(b) The Parties intend that each representation, warranty and covenant contained herein shall have independent significance. If any Party has breached any representation, warranty or covenant contained herein in any respect, the fact that there exists another representation, warranty or covenant relating to the same subject matter (regardless of the relative levels of specificity) that the Party has not breached shall not detract from or mitigate the fact that the Party is in breach of the first representation, warranty or covenant.

(c) Each of Company and Parent has or may have set forth information in its respective disclosure letter in a section thereof that corresponds to the section of this Agreement to which it relates. A matter set forth in one section of a disclosure letter need not be set forth in any other section of the disclosure letter so long as its relevance to the latter section of the disclosure letter or section of this Agreement is readily apparent on the face of the information disclosed in the disclosure letter to the Person to which such disclosure is being made. The fact that any item of information is disclosed in such a disclosure letter shall not be construed to mean that such information is required to be disclosed pursuant to this Agreement. Such information and the dollar thresholds set forth herein shall not be used as a basis for interpreting the terms "material," "Material Adverse Effect" or other similar terms in this Agreement.

10.14. Incorporation of Exhibits and Schedules. The Exhibits and Schedules identified in this Agreement are incorporated herein by reference and made a part hereof.

10.15. Specific Performance. Each of the Parties acknowledges and agrees that the other Parties would be damaged irreparably in the event any of the provisions of this Agreement are not performed in accordance with their specific terms or otherwise are breached. Accordingly, each of the Parties agrees that the other Parties shall be entitled to an injunction or other equitable relief to prevent breaches of the provisions of this Agreement and to enforce specifically this Agreement and the terms and provisions hereof in any action instituted in any court of the United States or any state thereof having jurisdiction over the Parties and the matter (subject to the provisions set forth in Section 10.16 below), in addition to any other remedy to which they may be entitled, at law or in equity.

10:16. Submission to Jurisdiction. Each Party irrevocably submits to the jurisdiction of the courts of the State of Delaware and the federal courts of the United States of America located in the State of Delaware for the purposes of any suit, action or other proceeding arising out of this Agreement and of the documents referred to in this Agreement and any transaction contemplated hereby. Each Party irrevocably and unconditionally waives any objection to the laying of venue of any action, suit or proceeding arising out of this Agreement or any transaction contemplated hereby in said courts and hereby further irrevocably and unconditionally waives and agrees not to plead or claim in any such court that any such action, suit or proceeding brought in any such court has been brought in an inconvenient forum. Each Party further irrevocably consents to the service of process out of any of the aforementioned courts in any such suit, action or other proceeding by the mailing of copies thereof by mail to such party at its address set forth in this Agreement, such service of process to be effective upon acknowledgment of receipt of such registered mail; provided that nothing in this Section 10.16 shall affect the right of any party to serve legal process in any other manner permitted by law. The consent to jurisdiction set forth in this Section 10.16 shall not constitute a general consent to service of process in the State of Delaware and shall have no effect for any purpose except as provided in this Section 10.16. The parties agree that a final judgment in any such suit, action or proceeding shall be conclusive and may be enforced in other jurisdictions by suit on the judgment or in any other manner provided by law.

10.17. Waiver of Jury Trial. EACH PARTY ACKNOWLEDGES AND AGREES THAT ANY CONTROVERSY THAT MAY ARISE UNDER THIS AGREEMENT IS LIKELY TO INVOLVE COMPLICATED AND DIFFICULT ISSUES, AND THEREFORE EACH SUCH PARTY HEREBY IRREVOCABLY AND UNCONDITIONALLY WAIVES ANY RIGHT SUCH PARTY MAY HAVE TO A TRIAL BY JURY IN RESPECT OF ANY LITIGATION DIRECTLY OR INDIRECTLY ARISING OUT OF OR RELATING TO THIS AGREEMENT, OR THE TRANSACTIONS CONTEMPLATED BY THIS AGREEMENT. EACH PARTY CERTIFIES AND ACKNOWLEDGES THAT (i) NO REPRESENTATIVE, AGENT OR ATTORNEY OF ANY OTHER PARTY HAS REPRESENTED, EXPRESSLY OR OTHERWISE, THAT SUCH OTHER PARTY WOULD NOT, IN THE EVENT OF LITIGATION, SEEK TO ENFORCE THE FOREGOING WAIVER, (ii) EACH PARTY UNDERSTANDS AND HAS CONSIDERED THE IMPLICATIONS OF THIS WAIVER, (iii) EACH PARTY MAKES THIS WAIVER VOLUNTARILY, AND (iv) EACH PARTY HAS BEEN INDUCED TO ENTER INTO THIS AGREEMENT BY, AMONG OTHER THINGS, THE MUTUAL WAIVERS AND CERTIFICATIONS IN THIS SECTION 10.17.

IN WITNESS WHEREOF, the Parties hereto have executed this Agreement effective the date first above written.

.

,

LDMI TELECOMMUNICATIONS, INC.

By: /s/ Pat O'Leary

• 🖉 🔹 •

Name: Pat O'Leary Title: CEO

LION ACQUISITION CORP.

By: <u>/s/ Aloysius T. Lawn IV</u>

Name: Aloysius T. Lawn IV Title: EVP-General Counsel

TALK AMERICA HOLDINGS, INC.

By: /s/ Aloysius T. Lawn IV

Name: Aloysius T. Lawn IV Title: EVP-General Counsel

Exhibit 99.1

Talk America to Acquire LDMI Telecommunications

HIGHLIGHTS:

- Creates leading midwest integrated communications provider (ICP)
- Accelerates Talk America's entry into the business market
- Broad product suite for the residential and business markets
- · Combination expected to generate significant operating efficiencies
 - VOIP business product rollout in fourth quarter 2005
 - Increased Talk America guidance for 2005

Southfield, MI and New Hope, PA, – May 23, 2005 – Talk America (NASDAQ:TALK) announced today that it has entered into an agreement to acquire LDMI Telecommunications, a leading facilities-based communications provider serving business and residential customers primarily in Michigan and Ohio. The acquisition of LDMI, when completed, will provide Talk America with a leadership position in the small and medium sized commercial marketplace and will expand its networking footprint.

Ed Meyercord, Chief Executive Officer and President of Talk America, commented "LDMI is an outstanding company with an excellent reputation in our core market. The LDMI team has developed a highly competitive T-1 product set and built strong distribution and operational capabilities. LDMI is resource rich with a talented and experienced team of telecom professionals who will be instrumental in executing our market expansion strategy. Across the board, this will be a great fit with our company."

Pat O'Leary, Chief Executive Officer and President of LDMI, said "we are delighted to be joining with Talk America, they are executing their networking plan seamlessly and have tremendous back office resources. The combination creates a formidable competitor in the Midwest committed to customer satisfaction."

ESTABLISHED PLAYER IN THE BUSINESS MARKET

LDMI, a privately held company established in 1992, currently provides service on over 131,000 voice equivalent lines and 27,000 data equivalent lines, primarily in Michigan and Ohio, supported by networking infrastructure in 83 end offices. LDMI purchased networking assets in 2003 and, in anticipation of regulatory changes, transformed itself into a facilities-based ICP. LDMI has over 2,300 T-1 customers and delivers a full suite of voice and data offerings, including specialized services such as virtual private networks, hosted applications and security services. The company has built a loyal customer base which includes "Smart-T" customers who receive integrated voice and data services that are dynamically allocated for efficient bandwidth utilization. The company also serves larger customers, such as AC Delco, a division of General Motors, with more sophisticated service offerings.

STRONG FINANCIAL POSITION AND OPERATING EFFICIENCIES

The combination of LDMI and Talk America will create a financially strong company with no debt, significant cash flow and a platform for continued market expansion. Talk America's financial flexibility gives it the opportunity to selectively invest in new markets and to grow through future acquisitions.

In 2004, LDMI's total revenues were \$120.7 million and its EBITDA was \$5.9 million. Talk America expects LDMI to generate EBITDA of \$8 million in 2005 and \$14 million in 2006, including the impact of merger related synergies. The transaction will also bring savings on capital expenditures of at least \$4 million through 2006.

LDMI currently has approximately 158,000 voice and data equivalent lines of which 118,000 are in Michigan and 40,000 are in Ohio. Mr. Meyercord continued, "LDMI's traffic is driven primarily by business customers, and should yield significant networking efficiencies when combined with our base of primarily residential customers."

VOIP BUSINESS PRODUCT ROLLOUT IN THE FOURTH QUARTER 2005

It is expected that, when the acquisition is completed, the combined companies will leverage Talk America's soft switching platform to roll out VOIP business services in the fourth quarter 2005. Talk America will utilize its Metaswitch to broaden LDMI's T-1 product suite and enable the company to expand into new geographic territories and increase its addressable market.

TRANSACTION EXPECTED TO CLOSE IN THE THIRD QUARTER 2005

At the closing of the transaction, in exchange for all of the stock of LDMI, Talk America will pay \$24 million in cash and issue 1.8 million shares of its common stock, with an estimated value of \$16 million based on the closing price as of May 20, 2005. As of April 30, 2005, LDMI had total debt, net of cash, of \$8.1 million. The purchase price for LDMI's equity together with LDMI's current debt, net of cash, totals \$48 million and represents a 3.4x multiple of LDMI's expected 2006 EBITDA, including the impact of merger related synergies. The cash portion of the purchase price, as well as the repayment of LDMI's debt at closing, will be funded out of cash on hand.

The LDMI acquisition transaction has been approved by the board of directors and the stockholders of LDMI and is subject to the receipt of certain regulatory approvals, which both companies expect to receive. The shares of common stock will be issued only to the holders of LDMI's preferred stock and will not be registered under the Securities Act of 1933. Talk America has agreed to file a registration statement with the Securities and Exchange Commission as promptly as reasonably practicable after the closing to permit resales of common stock by such holders. The companies expect that the transaction will close in the third quarter 2005.

FINANCIAL GUIDANCE

٠, پر ۱

Talk America has increased its financial and operating guidance for the full year 2005 to reflect the strength of its core business operations. Talk America has also provided financial and operating guidance that includes projected results for LDMI, assuming that its acquisition of LDMI is completed as of July 31, 2005.

		2005	
Metrics	Talk Previous	Talk Revised	Talk and LDMI (3)
Voice Lines on Network	175k	190k	255k
Data Lines on Network		40k	70k
Total Revenue	\$380-\$390 mm	\$385-\$395 mm	\$430-\$440 mm
EBITDA (1)	\$73–\$77 mm	\$78 - \$82 mm	\$83-\$87 mm
Capital Expenditures (2)	\$43 - \$47 mm	\$43 - \$47 mm	\$48 - \$52 mm

(1) Earnings Before Interest, Taxes, Depreciation and Amortization (EBITDA) is defined as operating income (loss) plus depreciation and amortization.

(2) Includes capitalized software.

(3) Based on July 31, 2005 closing date.

CONFERENCE CALL

Talk America and LDMI management will host a conference call to discuss the acquisition of LDMI at 8:30 a.m. ET on May 24, 2005. The call can be accessed by dialing the following: US (800) 720–9457, International, (303) 957–1279. A replay of the call will be available through 7:00 p.m. ET on May 31, 2005 by dialing the following: US (800) 633–8284, International (402) 977–9140. The reservation number for the replay is 21248125.

ABOUT TALK AMERICA

Talk America is a leading competitive communications provider that offers phone services and high speed internet access to both residential and business customers. Talk America delivers value in the form of savings, simplicity and quality service to its customers through its leading edge network and award-winning back office. For further information, visit us online at: http://www.talkamerica.com.

Please Note: The statements contained herein regarding the consummation of the acquisition of LDMI, the future results of LDMI and the effects and results of the acquisition on the business and results of operations of Talk America should be, and certain other of the statements contained herein may be considered forward-looking statements within the meaning of Section 27A of the Securities Act of 1933 and Section 21E of the Securities Exchange Act of 1934. Such statements are identified by the use of forward-looking words or phrases, including, but not limited to, "estimates," "expects," "expected," "anticipates," "anticipated," "forecast," "guidance," and "targets". These forward-looking statements are based on our current expectations. Although we believe that the expectations reflected in such forward-looking statements are reasonable, there can be no assurance that such expectations will prove to have been correct. Forward-looking statements involve risks and uncertainties and our actual results could differ materially from our expectations. In addition to those factors discussed in the foregoing, important factors that could cause such actual results to differ materially include, among others, the non-completion of the LDMI acquisition or our inability to integrate effectively and as anticipated the business of LDMI upon the completion of the acquisition, dependence on the availability and functionality of local exchange carriers' networks as they relate to the unbundled network element platform, failure to establish and deploy our own local network as we plan to do or to operate it in a profitable manner, increased price competition for long distance and local services, failure of the marketing of the bundle of local and long distance services and long distance services under our direct marketing channels to a smaller marketing footprint, attrition in the number of end users, failure to manage our collection management systems and credit controls for customers, interruption in our network and information systems, failure to provide adequate customer service, and changes in government policy, regulation and enforcement and/or adverse judicial or administrative interpretations and rulings relating to regulations and enforcement, including, but not limited to, the continued availability of the unbundled network element platform of the local exchange carriers network and unbundled network element pricing methodology.

For a discussion of such risks and uncertainties, which could cause actual results to differ from those contained in the forward-looking statements, see the discussions contained in our Annual Report on Form 10-K for the year-ended December 31, 2004, filed March 16, 2005, as amended by our Form 10-K/A filed March 30, 2005, and any subsequent filings. We undertake no obligation to update our forward-looking statements.

Contact Info:

Talk America Jeff Schwartz 215–862–1097 jschwartz@talk.com Created by 10KWizard Technology <u>www.10KWizard.com</u>

•7

Exhibit 3

· 2 · 4

OFFICERS AND DIRECTORS

Officers and Directors

LDMI Telecommunications, Inc. d/b/a LDMI Telecommunications also d/b/a FoneTel

.

Officers:

Patrick O'Leary Kim McDonald Linda Hanson Mark Wayne Mike Mahoney Richard Riordan Thomas Vaughn CEO and President Executive VP of IT & Operations Executive VP of Customer Care & Small Business Executive VP of Sales & Marketing Executive VP & Chief Financial Officer Director of Business Planning & Asst Secretary Secretary

Directors:

James Blanchard Thomas E. Darden, Jr. Dan Dennis David M. Hillman Greg Jones Michael A. M. Keehner William C. Mulligan Patrick O'Leary Kevin E. Sheehan Michael Skinner

Officers and Directors (Cont'd.)

<u>Talk America Inc.</u>

Officers:

`£

• 1

Edward Meyercord, III	CEO and President
Aloysius T. Lawn, IV	Executive Vice President, General Counsel and Secretary
David G. Zahka	Chief Financial Officer
Thomas Walsh	Senior Vice President – Finance and Treasurer
Jeffrey Earhart	Executive Vice President – Customer Operations
Warren Brasselle	Executive Vice President – Network Operations
Timothy Leonard	Chief Information Officer
Craig Pizer	Associate General Counsel and Assistant Secretary

Directors:

Edward Meyercord, III Aloysius T. Lawn, IV David G. Zahka